SECURITIES AND EXCHANGE COMMISSION

FORM 10-12G/A

Initial general form for registration of a class of securities pursuant to Section 12(g) [amend]

Filing Date: **2008-08-29 SEC Accession No.** 0001397807-08-000014

(HTML Version on secdatabase.com)

FILER

AmREIT Monthly Income & Growth Fund IV LP

CIK:1382787| IRS No.: 000000000

Type: 10-12G/A | Act: 34 | File No.: 000-53203 | Film No.: 081049054

SIC: 6798 Real estate investment trusts

Mailing Address 8 GREENWAY PLAZA STE 1000 HOUSTON TX 77046 Business Address 8 GREENWAY PLAZA STE 1000 HOUSTON TX 77046

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

Amendment No. 2 to Form 10

GENERAL FORM FOR REGISTRATION OF SECURITIES PURSUANT TO SECTION 12(b) OR 12(g) OF THE SECURITIES ACT OF 1934

AMREIT MONTHLY INCOME & GROWTH FUND IV, L.P.

(Exact name of registrant as specified in its charter)

	(=	egas mas apresigned as a sum of			
Delawar	·e	20-5685431			
(State or other jur	isdiction of	(I.R.S. Employ	er		
incorporation or or	rganization)	Identification N	lo.)		
9 C DI	C'4- 1000				
8 Greenway Plaza Houston,		77046			
(address of principle ex		(Zip Code)			
(uuaress of principle es	tecutive offices)	(Zip Code)			
		(713) 850-1400			
	(Registrant's tel	lephone number, including area code)			
		Copy to:			
		Jason W. Goode Alston & Bird LLP			
	120	11 West Peachtree Street			
		inta, Georgia 30309-3424			
	Aua	(404) 881-7000			
		(404) 001-7000			
	Securities to be reg	gistered under Section 12(b) of the Act:			
Title of each	class	Name of each exchang	e on which		
to be so regis		Each class is to be registered			
None		None			
	Securities to be reg	gistered under Section 12(g) of the Act:			
	Limi	ited Partnership Interests			
		(Title of Class)			
		accelerated filer, an accelerated filer, non-accelera			
	n of "large accelerat	ted filer," "accelerated filer" and "smaller reporti	ng company" in Rule 126-2		
of the Exchange Act. (Check one):					
Large accelerated filer		Accelerated filer			
Non-accelerated filer		Smaller Reporting Company	X		
	(Do not o	check if a smaller reporting company)			

TABLE OF CONTENTS

ITEM	DESCRIPTION				
	CAUTIONARY NOTE REGARDING FORWARD LOOKING STATEMENTS	1			
1	BUSINESS	1			
2	FINANCIAL INFORMATION	16			
3	<u>PROPERTIES</u>	19			
4	SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT	21			
5	DIRECTORS AND EXECUTIVE OFFICERS	21			
6	EXECUTIVE COMPENSATION	21			
7	CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS	22			
8	LEGAL PROCEEDINGS	23			
9	MARKET PRICE OF AND DIVIDENDS ON THE REGISTRANT'S COMMON EQUITY AND OTHER STOCKHOLDER MATTERS	23			
10	RECENT SALES OF UNREGISTERED SECURITIES	24			
11	DESCRIPTION OF REGISTRANT'S SECURITIES TO BE REGISTERED	24			
12	INDEMNIFICATION OF DIRECTORS AND OFFICERS	27			
13	FINANCIAL STATEMENTS AND SUPPLEMENTARY DATA	27			
14	CHANGES IN AND DISAGREEMENTS WITH ACCOUNTANTS ON ACCOUNTING AND FINANCIAL DISCLOSURE	27			
15	FINANCIAL STATEMENTS AND EXHIBITS	27			
1.3	THE METAL STATEMENTS AND LAMBITO	41			

CAUTIONARY NOTE REGARDING FORWARD LOOKING STATEMENTS

Certain statements contained in this Registration Statement on Form 10 of AmREIT Monthly Income & Growth Fund IV, L.P. other than historical facts may be considered forward-looking statements. We refer to AmREIT Monthly Income & Growth Fund IV, L.P. and, where required by the context in which the term is used, as the "Partnership," "we," "us" and "our." Such statements include, in particular, statements about our plans, strategies and prospects and are subject to certain risks and uncertainties, as well as known and unknown risks, which could cause actual results to differ materially from those projected or anticipated. Therefore, such statements are not intended to be a guarantee of our performance in future periods. Such forward-looking statements can generally be identified by our use of forward-looking terminology such as "may," "will," "would," "could," "should," "expect," "intend," "plan," "anticipate," "estimate," "predict," "believe," "potential," "continue" or other similar words. Readers are cautioned not to place undue reliance on these forward-looking statements, which speak only as of the date this Registration Statement is filed with the Securities and Exchange Commission, which we refer to as the SEC. We make no representation or warranty, express or implied, about the accuracy of any such forward-looking statements contained in this Registration Statement, and we do not intend to publicly update or revise any forward-looking statements, whether as a result of new information, future events or otherwise. See the section captioned "Item 1. Description of Business - Risk Factors" of this Registration Statement on Form 10.

Any forward-looking statements are subject to unknown risks, uncertainties and other factors and are based on a number of assumptions involving judgments with respect to, among other things, future economic, competitive and market conditions, all of which are difficult or impossible to predict accurately. To the extent that our assumptions differ from actual results, our ability to meet such forward-looking statements, including our ability to generate positive cash flow from operations, provide distributions to our partners and maintain the value of our real estate properties, may be significantly hindered.

ITEM 1. BUSINESS

Background

We are filing this Form 10 to register our units of limited partnership interest, or the Units, pursuant to Section 12(g) of the Securities Exchange Act of 1934, as amended, or the Exchange Act. Pursuant to the terms of a private placement memorandum dated November 15, 2006, which we refer to as the Offering Memorandum, we began raising proceeds from a private placement, which we refer to as the Offering. The Offering began on November 15, 2006, and we closed it on April 1, 2008, at which point we had sold approximately 2,000 Units and raised aggregate gross proceeds of approximately \$50.0 million, approximately \$5.3 million of which was used to cover the costs of the Offering. We refer to the holders of our Units as Limited Partners. We are subject to the registration requirements of Section 12(g) of the Exchange Act because the aggregate value of our assets exceeds applicable thresholds and the Units are held by 500 or more persons. As a result of our obligation to register our securities with the SEC under the Exchange Act, we will be subject to the requirements of the Exchange Act rules. In particular, we will be required to file quarterly reports on Form 10-Q, annual reports on Form 10-K and current reports on Form 8-K and otherwise comply with the disclosure obligations of the Exchange Act applicable to issuers filing registration statements pursuant to Section 12(g) of the Exchange Act. This Registration Statement on Form 10 and other reports and other information we subsequently file can be inspected and copied at the SEC Public Reference Room at 100 F Street, N.E., Washington, D.C. 20549. Such reports and other information may also be obtained from the SEC's website at http://www.sec.gov. Further information about the operation of the Public Reference Room may be obtained by calling the SEC at 1-800-SEC-0330.

We are a Delaware limited partnership formed on October 10, 2006 to acquire, develop and operate, directly or indirectly through joint venture arrangements, a portfolio of commercial real estate consisting primarily of multi-tenant shopping centers and mixed-use developments throughout the Southern and Southwestern United States. Our investment strategy is to (1) sell approximately 40% of our properties when appropriate and re-invest the net sales proceeds into additional properties and (2) retain the remaining properties as income-producing assets during our entire operating period. At the end of our seven year operating period, which expires on November 15, 2013, we will sell our properties and distribute the net proceeds to our Partners, as defined below. Our principal office is located at 8 Greenway Plaza, Suite 1000, Houston, Texas 77046. Our telephone number is (713) 850-1400.

Our general partner is AmREIT Monthly Income & Growth IV Corporation, a Delaware corporation that we refer to as the General Partner. Our General Partner is a wholly-owned subsidiary of AmREIT, a Texas real estate investment trust listed on the American Stock Exchange (AMEX: AMY). AmREIT and its predecessors have sponsored and advised 18 partnerships formed for the purpose of investing in properties during its 23 year history. Our General Partner has the exclusive right to manage our business and affairs on a day-to-day basis pursuant to our limited partnership agreement, which we refer to as the Partnership Agreement. Our General Partner is responsible for all of our investment decisions, including decisions relating to the properties to be developed, the method and timing of financing or refinancing the properties, the selection of tenants, the terms of the leases, the method and timing of the sale of our properties and the reinvestment of net sales proceeds. Our General Partner utilizes the services of AmREIT and its affiliates in performing its duties under the Partnership Agreement. Our General Partner has contributed \$1,000 to us for its general partner interest, and AmREIT has contributed \$800,000 to us for its limited partner interest. We refer to our General Partner and our Limited Partners collectively as the Partners.

We issued Units in the Offering in reliance upon exemptions from the registration requirements of the Securities Act of 1933, as amended, or the Securities Act, and state securities laws. As a result, our Limited Partners may not transfer the Units except pursuant to an effective registration statement or pursuant to an exemption from registration. The Units are not currently listed on a securities exchange and there currently is no market for the Units. We do not intend to list the Units at this time and have no plans to list the Units on an exchange in the future.

We commenced our principal operations on December 8, 2006, when we acquired our first interest in a property. On January 12, 2007, we raised the minimum offering of \$1,000,000 pursuant to the terms of our Offering Memorandum and issued the initial 40 Units.

We own investments in four properties, all of which are located in Texas, through joint ventures with affiliates and non-affiliates. Three of the properties are multi-tenant retail properties and the fourth property consists of vacant land for development into a multi-tenant retail property. The three existing retail properties comprise approximately 612,000 rentable square feet. As of December 31, 2007 the three existing retail properties were 75% leased. We do not own any properties directly.

The following is a summary of the four properties in which we owned an interest as of December 31, 2007. Each property is owned through a joint venture and is not consolidated in our financial statements.

Property	Location	Percent Owned	_	Square Footage	Percentag Leased	e	Annualized Gross Base Rent(1)
Casa Linda Plaza (2)	Dallas	50	%	324,638	82	%	\$3,207,000
Woodlake Square (3)	Houston	40	%	205,522	73	%	1,985,000
Westheimer Gessner							
(3)	Houston	40	%	82,120	50	%	627,000
Cambridge Holcombe							
(4)	Houston	50	%	TBD		?%	TBD -
Total				612,280			\$5,819,000

- Annualized gross base rent represents base rents in place on leases with rent having commenced as of December 31, 2007 and does not reflect straight-line rent or other adjustments under generally accepted accounting principles.
- (2) Property is under redevelopment.
- (3) Property is planned for redevelopment.
- (4) Property is vacant land and is planned for redevelopment.

See "Item 3. Properties" for a more detailed description of our investments in properties.

Investment Objectives

Our investment objectives are:

- to preserve and protect our Limited Partners' capital contributions;
- to provide cash distributions to our Partners through the operation of our properties; and
- to add value to our properties in which we have invested during our operating period and to realize appreciation upon the ultimate sale of such properties.

Investment Strategy

We were formed to acquire, develop and operate a portfolio of commercial real estate consisting primarily of multi-tenant shopping centers and mixed-use developments. We will seek to maximize our income and capital growth during our opGeneral Partnererating period by (1) selling approximately 40% of our properties when appropriate and re-investing the net sales proceeds into additional properties (we refer to this process as Active Management), and (2) owning the remaining properties as income-producing assets during our entire operating period. Our operating period will continue until November 15, 2013. The operating period may be extended to November 15, 2015 only with the consent of the majority of Units held by our Limited Partners. At the end of our operating period, the General Partner will begin an orderly liquidation of our properties and distribute the net proceeds to our Partners. If our General Partner does not diligently pursue the liquidation of our properties at the end of our operating period, AmREIT will forfeit its \$800,000 investment in Units.

Equity Allocation

As of March 31, 2008, we had received \$43.8 million in net proceeds from the Offering (net of \$5.9 million in offering costs). We have invested approximately \$27.6 million of such proceeds in real properties through various joint ventures. We have approximately \$13.9 million in cash and cash equivalents remaining as of March 31, 2008 for future investments in real property. The remaining net proceeds from the Offering were distributed to our Limited Partners as a return of capital. Once we have invested all of the net proceeds from the Offering, we estimate that approximately 60% to 70% of our capital will be invested in existing commercial shopping centers, primarily multi-tenant properties, mixed-use properties with a strong retail component and, to a lesser degree, single-tenant properties. Three of our four current investments are in existing retail properties. We expect to have invested the remaining approximately 30% to 40% of our capital for equity investments in the development of commercial shopping centers, consisting primarily of multi-tenant and mixed-use developments and, to a lesser degree, single-tenant developments. However, if our General Partner determines that the risk/return dynamic of the development investment activities or the acquisition of existing retail centers dramatically improves or declines, our capital will be reallocated accordingly.

As we make additional investments, we will seek to avoid tenant concentration in one industry by pursuing properties with tenants representing a variety of industries. We will also seek to avoid tenant concentration by limiting the size and number of our properties held by a single tenant or brand. We intend to diversify our tenant base so that no single tenant or brand represents more than 15% of our overall gross revenues from our portfolio following the investment of all of the proceeds from the Offering.

Investments in Properties with Operating Histories

Once we have invested all of the net proceeds from the Offering, we expect approximately 60% to 70% of our portfolio to be comprised of existing shopping centers leased to high quality tenants, consisting primarily of multi-tenant centers, mixed-use properties with a dominant retail component and, on a selective basis, free standing single-tenant properties. These investments will primarily be shopping centers that are grocery-anchored, strip center, mixed-use or lifestyle properties whose tenants consist of national, regional and local retailers. We anticipate that our grocery-anchored shopping centers will be anchored by an established grocery store operator in the region, such as Albertson's and Randall's. Our lifestyle centers will typically be anchored by a combination of national and regional tenants that provide customer traffic and tenant draw for specialty and restaurant tenants that support the local consumer. Our other shopping centers typically will be leased to national and regional tenants such as Walgreen's, Border's Bookstore, 24 Hour Fitness, Petco and Jos A. Bank, as well as a mix of local and value retailers. We may also invest in shopping centers that are leased to national drug stores, national restaurant chains, national value-oriented retail stores and other regional and local retailers. We seek existing properties where the majority of the leases are either leased or guaranteed by the lessee's parent company, not just the operator of the individual location, and in areas of substantial retail shopping traffic. Our strategy is to acquire properties that attract tenants that provide basic staples and convenience items to local customers. We believe that sales of these items are less sensitive to business cycle fluctuations than higher priced retail items.

Development and Redevelopment Properties

Once we have invested all of the net proceeds from the Offering, we expect approximately 30% to 40% of our portfolio to be comprised of development properties, either directly or indirectly through joint ventures. Of the four properties in which we currently own an interest, Casa Linda is currently under redevelopment. Additionally, we plan to redevelop Woodlake Square and Westheimer Gessner. Cambridge Holcombe consists of vacant land located in the Texas Medical Center in Houston, Texas and we are in the initial stages of

planning this property's development. The amount of equity committed to development and redevelopment projects is generally 25% to 100% of the total cost of the project, with the remaining costs being funded through lines of credit, construction financing or other property level mortgage financing. We work closely with local development partners in these transactions throughout the development process.

We plan to redevelop Woodlake Square and Westheimer Gessner, and it is likely that these properties will become mixed-use developments which may include office, residential, entertainment and hospitality components. It is possible that our future investments will have mixed-use components as well. Our General Partner will analyze the market surrounding each mixed-use development to determine the optimal mix of retail to non-retail components. We will develop in locations that provide limited competition, quality location and strong market fundamentals. We intend to commence the leasing process before construction.

Our General Partner may hire a general contractor to provide construction and construction management services for each of its development and redevelopment projects. The general contractor will be entitled to fees for providing these services, and these fees may be paid on a fixed price basis or a cost plus basis. We anticipate that AmREIT Construction Company, an affiliate of our General Partner, will provide construction and construction management services for many of our development and redevelopment projects. We may or may not competitively bid the construction management services for projects for which AmREIT Construction Company is selected. Where AmREIT Construction Company is selected to provide construction and construction management services, such services will be provided on terms and conditions no less favorable to us than can be obtained from independent third parties for comparable services in the same location.

Each of our development and redevelopment projects has and will have a project manager assigned to ensure all necessary functions are performed. The project manager is responsible for coordinating all phases of the project, including the feasibility study of each project prior to the commencement of development and much of the pre-development work. Each development will also have a construction manager who is responsible for coordinating all the outsourced trades including architectural, engineering, environmental and construction contractors. The construction manager will be an employee of AmREIT Construction Company in the event that it serves as the general contractor of a development project. The project and construction managers will be jointly responsible for the preparation and adherence to the development budgets. Capital inflows and outflows are carefully tracked and compared against budgets. Actual costs versus budget reports will be prepared on a monthly basis for review by various parties including the development team, management team and lenders. The project and construction managers will work in unison to ensure each project is built within budget and on a timely basis.

We may employ our capital in at-risk situations to tie up developable land sites using, among other things, purchase agreements and options. Such commitments may not necessarily result in the eventual acquisition of a land site, as we may elect to forfeit funds after completing our due diligence.

Location of Properties

We seek investments in properties located throughout the United States, with a primary focus on markets with increasing population growth and urban density. As a result of our General Partner's experience in developing, acquiring and managing retail real estate in metropolitan Texas markets, each of the four properties in which we currently own an interest is located in Texas. The economies of the Texas metropolitan markets where we own investments will have a significant impact on our cash flow and the value of our properties. Although a downturn in the economies of these metropolitan areas could adversely affect our business, general retail and grocery anchored shopping centers that provide necessity-type items tend to be less sensitive to macroeconomic downturns.

Although we intend to invest only in properties in the United States, we are not prohibited from making investments in foreign countries that meet our investment criteria.

Investment Decisions

Our General Partners uses commercially reasonable efforts to present to us suitable investments consistent with our investment objectives and policies. In pursuing our investment objectives and making investment decisions for us, our General Partner considers relevant real estate property and financial factors, including the location of the property, its suitability for any development contemplated or in progress, its income-producing capacity, the prospects for long-range appreciation, liquidity and tax considerations. Moreover, to the extent feasible, our General Partner strives to select a diversified portfolio of the properties in terms of geography, type of property and industry of the tenants, although the number and mix of properties acquired will largely depend upon real estate and market conditions and other circumstances existing at the time properties are acquired.

Prior to acquiring a property, our General Partner undertakes an extensive site review. Our General Partner also typically undertakes a long-term viability and market value analysis, including an inspection of the property and surrounding area by an acquisition specialist and an assessment of market area demographics, consumer demand, traffic patterns, surrounding land use, accessibility, visibility, competition and parking. Our General Partner may also take the following steps, depending on the property and terms agreed to:

- obtain an independent appraisal of the property;
- obtain an independent engineering report of the property's mechanical, electrical and structural integrity (sale/leaseback properties only);
- evaluate both the current and potential alternative uses of the property; and
- obtain an independent Phase I environmental site assessment.

Our General Partner is not required to obtain an appraisal in connection with an acquisition, although it is anticipated that if third-party financing is being provided by a commercial lender, such lender will obtain an independent appraisal.

Real Estate Fundamentals

Our General Partner believes that sound real estate fundamentals will allow us to attract the best tenants and produce the best results for our real estate portfolio. Our General Partner believes that factors such as corner locations, high automobile traffic counts, high populations, high household incomes and limited opportunities for competition produce favorable conditions for the success of the tenant and the retail property. Each of our current investment properties possesses these characteristics. Corner locations traditionally offer favorable access because these locations can access traffic in all directions. High traffic passing a retail property provides maximum exposure for retail tenants. A dense population base surrounding a retail property provides a large consumer base for a tenant's business. Areas that have high household income have more disposable income that is affected less by economic cycles. Locations that have few opportunities for new retail properties offer a limited supply of space and thus have the best likelihood of growing rental rates. Although a shopping center seldom offers all of these factors, our General Partner will use these criteria to measure the quality and relative value of opportunities relative to others in evaluating each proposed real estate investment.

Our General Partner also believes that its ability to obtain locations near national commercial tenants such as Wal-Mart, Home Depot and Target, which are major traffic generators for other commercial tenants, should enable us to attract brand name, high quality tenants.

Tenant Quality and Monitoring

We seek to attract high quality tenants for our properties. A tenant will be considered "high quality" if at the time of acquisition, the tenant has a regional or national presence, operating history of 10 or more years and a net worth in excess of \$50 million. When available, our

General Partner will rely on national credit rating agencies such as Standard & Poor's to assist in such determination. If public data is not available, our General Partner will rely on its experience, its own credit analysis and resources provided by its lenders to qualify a prospective tenant.

If a tenant has a public debt rating, we will seek tenants that have: a debt rating by Moody's of Baa3 or better or a credit rating by Standard & Poor's of BBB- or better; or a guaranty for its payments under the lease by a guarantor with a debt rating by Moody's of Baa3 or a credit rating by Standard & Poor's of BBB or better.

Moody's ratings are opinions of future relative creditworthiness incorporating an evaluation of franchise value, financial statement analysis and management quality. The rating given to a debt obligation describes the level of risk associated with receiving full and timely payment of principal and interest on that specific debt obligation and how that risk compares with that of all other debt obligations. The rating therefore measures the ability of a company to generate cash in the future. Standard & Poor's assigns a credit rating both to companies as a whole and to each issuance or class of a company's debt. A Standard & Poor's credit rating of BBB-, which is the lowest investment grade rating given by Standard & Poor's, is assigned to companies or issuances that exhibit adequate protection parameters. However, adverse economic conditions or changing circumstances are more likely to lead to a weakened capacity of the company to meet its financial commitments.

Net Leases

Our properties typically enter into net leases with their tenants. "Net leases" are leases that typically require that tenants pay all or a majority of the operating expenses, including real estate taxes, special assessments and sales and use taxes, utilities, insurance and building repairs related to the property, in addition to the lease payments. There are various forms of net leases, typically classified as triple net, double net and bondable. Triple net and bondable leases typically require the tenant to pay all costs associated with a property in addition to the base rent and percentage rent, if any. Double net leases typically hold the landlord responsible for the roof and structure of the building while the tenant is responsible for all remaining expenses associated with the real estate. Since each lease is an individually negotiated contract between two or more parties, each contract will have different obligations of both the landlord and tenant. Many large national tenants have standard lease forms that generally do not vary from property to property, and we will have limited ability to revise the terms of leases to those tenants.

Our properties have tenants with varying lease terms. We have acquired interests in properties under which the lease terms have partially run. We evaluate the lease term risk based on criteria such as whether the property is in an attractive location, difficult to replace or has other significant favorable real estate attributes. The leases for our properties generally require our tenants to pay a predetermined annual base rent. Some of these leases contain provisions that increase the amount of base rent payable at points during the lease term and/or percentage rent that can be calculated by a number of factors. In addition, these leases generally require that each tenant pay the cost of the liability insurance covering the property or provide such coverage. The third-party liability coverage will insure, among others, us, our General Partner, and any entity formed by us to hold the property. The leases for our properties generally require that each tenant obtain, at its own expense, property insurance naming the above parties as an insured party for fire and other casualty losses in an amount that generally equals the full replacement value of such property. Tenants are generally required to obtain our General Partner's approval of all such insurance.

Ownership Structure

As of December 31, 2007, we owned interests in four properties through joint ventures with affiliated and non-affiliated entities. The joint ventures are structured as limited partnerships and directly own fee simple interests in the underlying properties. We own a portion of each of the limited partnerships and have the ability to significantly influence the strategic and operating decisions of the joint ventures.

For our future investments, we may continue to acquire properties through joint ventures, or we may acquire, directly or indirectly, (1) fee simple interests in owned real property and (2) leasehold interests in real property subject to long-term ground leases. We may acquire future properties individually or on a portfolio basis. Our General Partner and its affiliates may purchase properties in their own name or in entities that they control, assume loans in connection with the purchase of properties, and temporarily hold title to properties for the purpose of facilitating the acquisition of properties by us. Our General Partner and its affiliates may also borrow money or obtain financing and complete construction of properties.

For future investments we may make in development properties, we may enter into arrangements with the seller or developer, provided that the property is pre-leased to a high quality tenant. In these cases, we will be obligated to purchase the property at the completion of construction, provided that the construction conforms to definitive plans, specifications and costs approved in advance by our General Partner. We will receive a certificate of an architect, engineer or other appropriate party, stating that the property complies with all plans and specifications. If renovation or remodeling is required prior to the purchase of a property, our General Partner expects to pay a negotiated maximum amount upon completion.

We may enter into sale and leaseback transactions, under which we will purchase a property and lease the property back to the seller.

Joint Ventures

As of December 31, 2007, we owned interests in four properties through joint ventures with affiliates of our General Partner and non-affiliates. The joint ventures are structured as limited partnerships, in which we own interests. Each of these limited partnerships owns one of the four underlying properties. Each general partner is responsible for establishing policies and operating procedures with respect to the business and affairs of the underlying property. However, we and each of our joint venture partners must approve all significant decisions involving the properties, including decisions involving acquisitions and dispositions of properties, refinancing, operational budgets and significant capital improvements.

We may invest in additional properties through joint ventures with third-party developers and real estate investors, including our General Partner, its affiliates and entities owned or managed by its affiliates. Such joint ventures may include investments in limited liability companies or other co-ownership arrangements whose purpose is the acquisition or improvement of the properties. Our General Partner and its affiliates may provide services to the joint venture, including, but not limited to, acquisition, development, management, leasing and/or real estate disposition services. Our current joint venture investments contain, and we will not enter in future joint venture investments unless they contain, the following features:

- our right either to approve significant decisions of the joint venture or to control operations of the joint venture, subject to the right of the joint venture partner to approve sales or refinancing;
- the total compensation paid by us and the joint venture to our General Partner and its affiliates in connection with a joint venture will not exceed the compensation which would be permissible under the Partnership Agreement if we owned 100% of the joint venture;
- no duplication of joint venture costs and expenses and our costs and expenses relating to the joint venture business, including organization and syndication expenses, acquisition and development costs; and
- any purchase, sale or financing transactions between the joint venture partner and our General Partner or its affiliates must be on terms, which are commercially reasonable and comparable to those relating to transactions between unrelated parties.

Our investments in Casa Linda Plaza, Woodlake Square and Westheimer Gessner were made through joint ventures with affiliates of our General Partner. For any future investment with our General Partner, its affiliate, or an entity owned or managed by an affiliate, our General Partner or the managing member, largest shareholder, general partner or other controlling or majority owner of the affiliate or such other entity may contribute capital to the joint venture on the same terms and conditions as us. Allocable profits in a joint venture will be calculated based on the sum of net sale proceeds from the sale of a property (after repayment of debt) plus reserves less capital contributions of each joint venture partner plus actual origination and carrying costs of the additional financing incurred in connection with such property. Distributions will be pro rata to the joint venture partners based on their aggregate capital contributions.

Our General Partner or its affiliate may form another partnership or other investment vehicle with essentially the same investment objectives as us and such entity may acquire properties through a joint venture. Such other entities may have as investors controlling persons or other former and current investors in programs sponsored by affiliates. The terms and conditions upon which persons become investors in such other entities may differ from our terms and conditions.

Sale of the Properties

Operating Period Dispositions

During our operating period, our General Partner intends to hold our properties until such time as a sale or other disposition appears to be advantageous to achieve our investment objectives or until it appears that such objectives will not be met. We have not yet sold any of our investments in properties. Our General Partner anticipates an average holding period for Actively Managed properties (approximately 40% of the portfolio) of 12 to 48 months, recognizing that certain projects may have a shorter or longer holding period. When deciding whether to sell properties during our operating period, our General Partner will consider factors such as potential capital appreciation, cash flow, the availability of other attractive investment opportunities and federal income tax considerations.

Reinvestment of Net Sales Proceeds

During our operating period, our General Partner intends to reinvest the net sales proceeds generated from the sale of our Actively Managed properties. In making the determination of whether to reinvest the net proceeds from a particular sale, our General Partner will first determine whether we have adequate cash flow to support the anticipated monthly distribution to our Limited Partners. Second, our General Partner will determine if there are sufficient reserves to pay the special tax distribution to our Limited Partners at the end of the year.

Liquidation Period Dispositions

Our General Partner will in good faith actively market for sale all of our properties other than those in the development or redevelopment stage and commence an orderly Partnership liquidation on or before November 15, 2013. Properties in the development or redevelopment stage at the end of the operating period will be marketed for sale upon completion. The operating period may be extended to November 15, 2015 only with the consent of Limited Partners owning a majority of the outstanding Units. If our General Partner does not take all commercially reasonable efforts to diligently pursue the portfolio sale and liquidation on our behalf as described above, AmREIT shall forfeit its \$800,000 investment in our Units.

Once our General Partner has marketed for sale all of our properties, it may take months or years for our General Partner to sell all of our properties and wind up our operations. In connection with the sale of a property, we may take purchase money obligations secured by a mortgage on the property as partial payment, thereby delaying any distribution of sale proceeds to our Limited Partners over the term to maturity of such obligations. The terms of payment to be accorded by us will be affected by custom in the area in which the property is located and the then-prevailing economic conditions.

AmREIT's Purchase Rights

We have granted AmREIT a limited right of first offer to purchase our wholly-owned properties that our General Partner determines are in our best interests to sell. For properties we own through a joint venture, we will grant AmREIT this right of first offer, subject to the approval of our joint venture partners. If our General Partner determines that it is in our best interests to sell one of our properties, our General Partner will notify AmREIT of our desire to sell such property. AmREIT will then have 30 days to determine whether to pay the market value to acquire the property for itself. To determine the market value of a property, both AmREIT and we, at our own cost and expense, shall appoint a real estate appraiser with at least five years full-time commercial appraisal experience and who is a member of the Appraisal Institute (MAI designation). If either of us fails to appoint an appraiser, the single appraiser appointed shall be the sole appraiser and shall determine market value. Each appraiser shall conduct an independent appraisal of the property within 30 days after the two appraisers are appointed. If the appraised values are within five percent of each other, the market value of the property shall be the median of the two appraisals. If the two appraisals are more than five percent apart, a third appraiser meeting the qualifications stated above and independent from each party shall be appointed by the existing appraisers. Each of the parties shall bear one-half of the cost of the third appraiser. Within 30 days after its selection, the third appraiser shall complete its appraisal of the property. Upon completion of the third appraisal, if the low appraisal and/or the high appraisal are/is more than five percent lower and/or higher than the middle appraisal, the low appraisal and/or the high appraisal shall be disregarded. If only one appraisal is disregarded, the remaining two appraisals shall be added together and their total divided by two, with the resulting quotient being the market value. If both the low appraisal and the high appraisal are disregarded as stated above, the middle appraisal shall be the market value. AmREIT will have 10 business days after the final determination of market value to elect to purchase the property.

If AmREIT agrees to pay the market value for a property, as determined above, we will sell the property to AmREIT. If AmREIT declines to acquire a property or fails to notify us of its intent to acquire a property within the requisite time periods, then we will market the property to third parties.

Marketing

Our General Partner believes that relationships and networking are the two primary components of marketing properties for sale. Our General Partner will use its in-house staff along with its relationships with key commercial brokers across the country to sell our properties.

Leverage

Our investments in Casa Linda Plaza, Woodlake Square and Westheimer Gessner were made by our contributing equity to joint ventures. The Casa Linda Plaza and Woodlake Square joint ventures used leverage to acquire its properties, while the Westheimer Gessner joint venture used only the cash proceeds from our equity contribution and the equity contribution of our affiliates to acquire its property. For our future investments in properties with operating histories, we intend to continue to leverage our investments in these properties using traditional, commercial real estate lending sources, as underwritten by our General Partner and the lender.

Our investment in Cambridge Holcombe was made by our contributing equity to a joint venture, which in turn, acquired Cambridge Holcombe using cash proceeds from our equity contribution and the equity contribution of our joint venture partner. We expect that for this and future development projects, local developers will employ construction financing consistent with that of traditional real estate projects and underwritten by the development partner, our General Partner and the lender. Construction financing employed by development partners will not directly be part of our capitalization as it will be off-balance sheet to the Partnership.

We may finance the acquisition of future investments in properties pursuant to new financing or assumption of existing indebtedness. We may incur secured or unsecured indebtedness at any time during our term. We may refinance a property after it has increased in value or when more favorable terms are available may permit us to retain such property and at the same time generate distributions to the Partners, enable us to engage in renovation or remodeling activities, or to make further acquisitions. We may incur debt for expenditures related to the properties, including to facilitate the sale or to pay for capital expenditures. There is no fixed limit on the term of any particular borrowing or the amount thereof. Generally, we may not incur indebtedness (or any refinancing thereof) to acquire or improve properties in an amount greater than 75% of our cash and cash equivalents plus the market value of our portfolio based on a cap rate approach applied to the net operating income of the property, with a target of 60% of the value of our assets. However, pursuant to our Partnership Agreement, we have exceeded this leverage limitation to fund our initial equity contribution to the Casa Linda Plaza joint venture. No assurance can be given as to the future availability of credit, the amount or terms thereof, or the restrictions that may be imposed on our future borrowings by lenders. The terms of our loans may require the properties to be held by special purpose entities whose controlling interest must be held by a corporation.

We may borrow money from AmREIT or its affiliates if our General Partner, in the exercise of its fiduciary duties, determines that the transaction is on terms that are fair and reasonable and no less favorable to us than comparable loans between unaffiliated parties.

Other Investments

We currently own interests in four properties through joint ventures. We may invest in other entities that own real estate, including in connection with joint ventures. We make these investments when our General Partner considers it more efficient to acquire an entity owning such real property rather than to acquire the properties directly. We have and may continue to acquire less than all of the ownership interests of such entities if our General Partner determines that such interests are undervalued and that a liquidation event in respect of such interests are expected within our operating period.

To the extent our General Partner determines it is in our best interest, due to the state of the real estate market, in order to diversify our investment portfolio or otherwise, our General Partner may make or invest in mortgage loans. We currently do not own any interests in mortgage loans. The criteria for investing in mortgage loans will be substantially the same as those involved in its investment in properties. Our Partnership Agreement limits our investment in mortgage loans to no more than 20% of the net proceeds from the Offering, unless the maximum allocation is increased by our General Partner with the consent of the Limited Partners owning a majority of the outstanding Units.

Our Partnership Agreement prohibits us from investing in securities of other issuers, except for permitted temporary investments pending utilization of our funds to acquire properties and except for investments in joint ventures and similar co-ownership arrangements, unless otherwise agreed to by Limited Partners holding a majority of the outstanding Units.

Our General Partner invests our reserves and other available funds not committed to investments in properties in United States government securities, securities issued and fully guaranteed by United States government agencies, securities issued and fully guaranteed by states or municipalities, certificates of deposit and time or demand deposits in commercial banks, bankers' acceptances, savings and loan association deposits or deposits in members of the Federal Home Loan Bank System, or money market instrument funds.

Conflicts of Interest

Our General Partner is subject to various conflicts of interest arising out of its relationship with us, the Limited Partners and AmREIT. Our General Partner and its affiliates have and will continue to try to balance our interests with their duties to other AmREIT-sponsored programs. However, to the extent that our General Partner or its affiliates take actions that are more favorable to other entities than to us, these actions could have a negative impact on our financial performance and, consequently, on distributions to our Limited Partners. Some of these conflicts are described below.

Limited Financial Resources

Our General Partner has no assets other than its general partner interest in us and its investment in our Units. In the event we have substantial capital needs, our General Partner will not have sufficient financial resources to satisfy these needs. In addition, AmREIT, the parent of our General Partner, has substantial financial obligations related to its properties and its interest in other programs and may not be able to provide us financial assistance in the event we have capital needs.

Interests in Other Real Estate Programs

AmREIT sponsors and manages real estate programs and ventures and intends to sponsor and manage more programs and ventures in the future. The only existing entities that our General Partner's management also manages which have similar investment objectives and may compete with us are AmREIT Income & Growth Fund, Ltd., AmREIT Monthly Income & Growth Fund, Ltd., AmREIT Monthly Income and Growth Fund II, Ltd., REITPlus, Inc. and AmREIT which we collectively refer to as the Similar AmREIT Programs. In addition to identifying, originating and effecting property acquisitions and development projects for us, our General Partner's management and its affiliates will continue to acquire and develop real estate for the account of other affiliated and unaffiliated investors. Conflicts of interest of our General Partner's management may arise in its allocating opportunities between us and other programs, particularly where our General Partner's profit or loss interest in such other investment is different than our General Partner's interest in the Partnership.

Conflicts of interest may also arise in connection with our General Partner's responsibilities to us and the responsibilities of its affiliates to other entities. For example, conflicts of interest could arise in management's allocation of its time and access to resources, such as financing, goods, material or labor, or in connection with its access to the leasing or resale markets, particularly during times these resources are scarce or in short supply. Conflicts could also result in the selection and marketing of projects if shortages of properties, materials or labor are insufficient causing market demands to require our General Partner's management to allocate project opportunities between us and its affiliates.

Because the management of our General Partner manages other investment funds and entities with similar investment strategies, including AmREIT, competition for properties will create a conflict of interest. Management expects to manage this conflict by providing a pipeline of real estate projects and opportunities to support all of its activities. All potential development and acquisition opportunities will initially be presented to AmREIT. If AmREIT elects not to make the investment, our General Partner's management will determine which of the entities that it advises, including us, would be most appropriate to make the proposed investment. To determine which entity should make the investment, management will first evaluate the investment objectives of each investment fund and determine if the opportunity is suitable for each fund. If the proposed investment is appropriate for more than one fund, management will then evaluate the portfolio of each fund, both in terms of geographic diversity and tenant concentration, to determine if the investment is most suitable to one fund. If the geographic diversity and tenant concentration analysis is not determinative, management will allocate the property to the fund with uncommitted funds available for the longest period of time. Our General Partner's management may also allow multiple investment funds to enter into joint ventures for the purchase or acquisition of a property.

Competition for Management Services

Our General Partner's management is engaged in substantial activities apart from our business, including their duties to AmREIT and other real estate programs sponsored by AmREIT. As such, they will devote only so much of their entire time to our affairs as is reasonably required in their judgment and they could have conflicts of interest in allocating their time between us and other entities. Our General Partner believes that it has sufficient staff to fully discharge its responsibilities to us.

Leasing Agents

Our General Partner retains the services of affiliated leasing agents to lease the properties in which we have invested. Because these leasing agents provide similar services to affiliates of our General Partner, including AmREIT, they face conflicts of interest if they are seeking to lease our properties and similar properties of its affiliates at the same time. In such an event, the affiliated leasing agent will seek to mitigate any potential conflict by presenting a potential tenant with all of the available properties, so that the potential tenant can select the property with the size, rent, location and other characteristics most suitable to its needs.

Properties

We rely on our General Partner and its affiliates in the selection, management and sale of the properties and do not have independent representation in this regard. Conflicts of interest could arise in connection with any interests affiliates of our General Partner may have in a particular property, including interests it may have as an affiliate of AmREIT in connection with any sale of a property to AmREIT, and those of the Limited Partners in operating the Partnership. See "Item 1. Business – Risk Factors."

Sale of Properties

Our General Partner actively manages our Actively Managed properties during the operating period. During the operating period, our General Partner will sell our properties from time to time to third-party real estate investors. Our General Partner may receive compensation in the form of a brokerage commission for the work performed in the sale of the properties. Because there is potential for our General Partner to earn a brokerage commission on each property sale, there is a conflict of interest in that our General Partner may sell a property simply to earn a brokerage commission, even if it is not in our best interest or the best interest of the Limited Partners.

No Arm's-Length Agreements

The compensation payable to our General Partner and its affiliates has not been determined by arm's- length negotiations. Also, a significant portion of this compensation is payable irrespective of the quality of the services provided or our success or profitability. There is no assurance that the amounts or terms of such compensation will not exceed that which would be paid to unrelated persons under similar circumstances in arm's-length transactions.

AmREIT's Interests in Other Programs

AmREIT, our General Partner's parent, engages for the account of others in other business ventures involving real estate development and investment. Moreover, AmREIT or its affiliates, including our General Partner, may in the future serve as management for the general partner of other companies or ventures, and acquire, develop and operate real estate related activities in the same areas as ours for their own account. Neither we nor any Limited Partner will be entitled to any interest in such other ventures.

Transactions with Affiliates

We may, from time to time, sell some of our properties to AmREIT. As a result of the inherent conflict in such a sale, we will only sell a property to AmREIT if they agree to pay the market value or the amount of a bona fide final third-party offer for that property. See "Item 1. Business – Sale of the Properties – AmREIT's Repurchase Rights" for a more detailed description of AmREIT's rights. In addition, we may borrow money from AmREIT or its affiliates. Although our General Partner will only approve an affiliated borrowing transaction if, in the exercise of its fiduciary duties, it determines that the terms are fair and reasonable and no less favorable to us than comparable loans between unaffiliated third parties, our General Partner could face conflicts of interest in connection with such a transaction that may not be resolved in the best interests of our Limited Partners.

In the event our General Partner or its affiliates receive compensation from us for any additional services performed on our behalf, such services will only be provided on terms and conditions no less favorable to us than can be obtained from independent third parties for comparable services in the same location. Such services may include, but are not limited to, leasing coordination fees, construction and construction management fees, including in connection with renovation and remodeling, and tax appeal fees. The fees for such services, if provided by affiliates, will be separately itemized and retained in our records.

Affiliated Property Manager and Construction Manager

AmREIT Realty Investment Corporation, or ARIC, an affiliate of our General Partner, provides property management services for the Woodlake Square, Casa Linda and Westheimer Gessner properties and may provide property management services for properties in which we acquire an interest in the future. ARIC is a wholly owned subsidiary of AmREIT and the officers of our General Partner are also officers of ARIC. As a result, we might not always have the benefit of independent property management to the same extent as if our General

Partner and the property manager were unaffiliated. In addition, given that ARIC is an affiliate of our General Partner, our agreements with ARIC are not negotiated at arm's-length, as they would between unrelated parties.

AmREIT Construction Company, an affiliate of our General Partner, will in the future provide construction and construction management services for certain of our development and redevelopment projects. AmREIT

Construction Company is currently engaged to provide construction management services for the Casa Linda Plaza property. As a result, we will not have the benefit of independent construction management to the same extent as if AmREIT Construction Company was unaffiliated. Since AmREIT Construction Company is an affiliate of our General Partner, we do not have the benefit of arm's-length negotiation of any contracts we enter into with AmREIT Construction Company that would apply between unrelated parties.

Lack of Separate Representation

Our legal counsel acts and may in the future act, as counsel to us, our General Partner, AmREIT and certain of our respective affiliates. There is a possibility that in the future, the interests of the various parties may become adverse, and under the Code of Professional Responsibility of the legal profession, our counsel may be precluded from representing any one or all of such parties. In the event that a dispute were to arise between us, our General Partner, AmREIT or any of our respective affiliates, separate counsel for such matters will be retained as and when appropriate.

Tax Matters Partner

Our General Partner serves as our tax matters partner and in the event of an audit of our federal income tax returns by the Internal Revenue Service, or IRS, it is possible that the interests of our General Partner in such an audit could become inconsistent with or adverse to the interests of the Limited Partners. Our expenses in contesting any such audit may reduce the amount of cash available for distribution. Further, our General Partner, who is primarily responsible for contesting federal income tax adjustments proposed by the IRS, may be subject to various conflicts of interest in connection with the negotiation and settlement of issues raised by the IRS in a federal income tax audit.

Employees

We have no employees. Our affairs are managed by our General Partner and our General Partner and its affiliates provide services to us related to acquisitions, property management, accounting, investor relations and other administrative services. We are dependent upon our General Partner and its affiliates for these services.

See "Item 7. Certain Relationships and Related Transactions" for discussion of our compensation arrangements.

Insurance

We believe that the properties in which we own an interest are adequately insured.

Competition

As we purchase properties for our portfolio, we are in competition with other potential buyers, including the Similar AmREIT Programs, for the same properties. As a result, we may either have to pay more to purchase the property than we would if there were no other potential acquirers or locate another property that meets our investment criteria. Although our existing retail properties are currently 75% leased and we have acquired, and intend to continue to acquire, properties subject to existing leases, the leasing of real estate is highly competitive, and we may experience competition for tenants from owners and managers of competing projects. As a result, we may have to provide free rent, incur charges for tenant improvements, or offer other inducements, or we might not be able to timely lease the space, all of which may have an adverse impact on our results of operations. At the time we elect to dispose of our properties, we will also be in competition with sellers of similar properties to locate suitable purchasers for their properties.

Concentration of Credit Risk

As of December 31, 2007, we had approximately \$18.3 million invested in a government security money market account. Such amounts are readily convertible into cash for use in our operations.

We have geographic concentration in our property holdings. In particular, as of December 31, 2007, all of our properties were located in Texas. We have tenant concentration in our properties. Rental income generated from Borders represented 11% of our 2007 rental income.

RISK FACTORS

Risks Related to an Investment in AmREIT Monthly Income & Growth Fund IV, L.P.

There is no public market for our Units; therefore, it is difficult for a Limited Partner to sell his or her Units.

There currently is no public market for our Units and we have no obligation or current plans to apply for listing on any public securities market. A Limited Partner may not sell, assign or otherwise transfer his or her Units without the prior written consent of our General Partner, which may be withheld for any reason or no reason. In addition, our Partnership Agreement restricts our Partners' ability to participate in a public securities market or anything substantially equivalent to one by providing that any transfer which may cause us to be classified as a "publicly traded partnership" as defined in Section 7704 of the Internal Revenue Code, or the Code, shall be deemed void and shall not be recognized.

We presently intend to effect a liquidity event on our before November 15, 2013; however, there can be no assurance that we will effect a liquidity event within such time or at all.

Under our Partnership Agreement, our General Partner is required to use commercially reasonable efforts to cause all property owned by us to be listed for sale on our before November 15, 2013, provided however, that such date may be extended to November 15, 2015 by a majority vote of the Limited Partners. If our General Partner has not taken such action or is not diligently pursuing such action by the applicable date, AmREIT shall forfeit its \$800,000 investment in our Units. In the event the General Partner determines that it is in the best interests of the Limited Partners not to pursue a liquidity event by the applicable date, it may request that the Limited Partners further extend the date by which the General Partner must cause our property investments to be listed for sale by amending our partnership agreement. If the Limited Partners do not approve such an amendment, our General Partner will diligently pursue a liquidity event. Until we effect a liquidity event, it will be very difficult for our Limited Partners to have liquidity for their investment in our Units other than limited liquidity through the sale or transfer of their Units with the prior written consent of our General Partner.

If our investments do not perform as we expect, then we may not be able to achieve our investment objectives or pay distributions to our Partners.

Our ability to achieve our investment objectives and to pay distributions is dependent upon the performance of our investments. Except for certain matters regarding the basic structure of the Partnership, all management decisions are made solely by our General Partner. Our Limited Partners must rely entirely on the management ability of our General Partner, which ability may be affected by the past experience of its management and our General Partner's present and future financial condition, among other factors. Our General Partner may not have been or may not be successful in obtaining suitable investments on financially attractive terms so that our objectives will be achieved by these investments.

Our properties face significant competition.

We face significant competition from other owners, operators and developers of retail properties. All or substantially all of our properties face competition from similar properties in the same markets. Such competition may affect our ability to attract and retain tenants and may reduce the rents we are able to charge. These competing properties may have vacancy rates higher than our properties, which may cause their owners to rent space at lower rental rates than those charged by us or to provide greater tenant improvement allowances or other leasing concessions. This combination of circumstances could adversely affect our results of operations, liquidity and financial condition, which could reduce distributions to our Partners.

There may be significant fluctuations in our quarterly results.

Our quarterly operating results fluctuate based on a number of factors, including, among others, interest rate changes; the volume and timing of our property acquisitions; the amount and timing of income generated by our real estate investments; the recognitions of gains or losses on property sales; the level of competition in our market; and general economic conditions, especially those affecting the retail industries. As a result of these factors, results for any quarter should not be relied upon as being indicative of performance in future quarters.

The economic performance and value of our properties depend on many factors, each of which could have an adverse impact on our cash flows and operating results.

The economic performance and the value of our properties depend on many factors, including, among others, the following:

- changes in the national, regional and local economic climate;
- local conditions such as an oversupply of space or a reduction in demand for retail real estate in the area;
- the attractiveness of the properties to tenants;
- competition from other available space;
- our ability to provide adequate management services and to maintain our properties;
- increased operating costs, if these costs cannot be passed through to tenants; and
- the expense of periodically renovating, repairing and releasing spaces.

Our properties consist primarily of multi-tenant retail centers and our performance is linked to general economic conditions in the market for retail space. The market for retail space has been and may continue to be adversely affected by weakness in the national, regional and local economies where our properties are located, the adverse financial condition of some large retailing companies, the ongoing consolidation in the retail sector, the excess amount of retail space in a number of markets and increasing consumer purchases through catalogues and the Internet. To the extent that any of these conditions occur, they are likely to affect market rents for retail space. In addition, we may face challenges in the management and maintenance of the properties or encounter increased operating costs, such as real estate taxes, insurance and utilities, which may make our properties unattractive to tenants.

Our dependence on rental income may adversely affect our ability to meet our debt obligations and make distributions to our Partners.

A significant percentage of our income is derived from rental income from our portfolio of properties. As a result, our performance depends on our ability to collect rent from tenants. Our income and our ability to make distributions would be negatively affected if a significant number of our tenants, or any of our major tenants:

- delay lease commencements;
- decline to extend or renew lease expiration;
- fail to make rental payments when due; or
- close stores or declare bankruptcy.

Any of these actions could result in the termination of the tenants' leases and the loss of rental income attributable to the terminated leases. Lease terminations by an anchor tenant or a failure by that anchor tenant to occupy the premises could also result in lease terminations or reductions in rent by other tenants in the same shopping center under the terms of some leases. In addition, we cannot be sure that any tenant whose lease expires will renew that lease or that we will be able to re-lease space on economically advantageous terms. The loss of rental revenues from a number of our tenants and our inability to replace such tenants may adversely affect our profitability and our ability to meet debt and other financial obligations and make distributions to our Partners.

Our ability to make distributions and achieve our investment objectives is dependent upon the performance of our General Partner in the acquisition and sale of real properties, the management of our portfolio, the selection of tenants for our properties and the determination of any financing arrangements. If our General Partner's management is distracted by intense activity of affiliated programs unrelated to us, our General Partner may be unable to allocate time and resources to our operations. If our General Partner is unable to allocate sufficient resources to oversee and perform our operations for any reason, we may be unable to achieve our investment objectives or to pay distributions to our Partners. In addition, because our General Partner does not have any employees or financial resources of its own, our success is dependent upon the continued contributions of certain of the key personnel of AmREIT, our General Partner's parent company, each of whom would be difficult to replace.

If AmREIT loses or is unable to obtain key personnel, our ability to implement our investment strategies could be delayed or hindered, which could adversely affect our ability to make distributions to our Partners and the value of our Limited Partners' investment.

Our success depends to a significant degree upon the contributions of certain of our General Partner's executive officers and other key personnel of AmREIT, our General Partner's parent, including, among others, H. Kerr Taylor, Chad C. Braun and Brett Treadwell, each of whom would be difficult to replace. We cannot guarantee that all, or any particular one of these individuals, will remain affiliated with us. We believe that our future success depends, in large part, upon AmREIT's ability to hire and retain highly skilled managerial, operational and marketing personnel. Competition for such personnel is intense, and our General Partner may not be successful in attracting and retaining such skilled personnel. If AmREIT loses or is unable to obtain the services of key personnel or does not establish or maintain appropriate strategic relationships, our ability to implement our investment strategies could be delayed or hindered, and the value of our Limited Partners' investment may decline.

We compete with other AmREIT affiliated entities for opportunities to acquire or sell certain real properties, which could adversely affect our investment opportunities.

We compete with other AmREIT affiliated entities for opportunities to acquire or sell certain types of real properties. AmREIT, REITPlus, Inc. and another AmREIT sponsored program, AmREIT Monthly Income & Growth Fund III, Ltd., or MIG III, are actively seeking investments substantially similar to those we seek to acquire. All potential development and acquisition opportunities are first presented to AmREIT. Only if AmREIT elects not to make a development project investment does its management determine which of the other AmREIT sponsored programs, including us, would be most appropriate to make the proposed investment. There is a risk that our General Partner will select for us a property investment that provides lower returns to us than a property purchased by another AmREIT sponsored program, and we may not have the opportunity to invest in certain investment opportunities that fit our investment strategy. In addition, our General Partner or its affiliates may form other investment vehicles that may directly compete with us.

In addition, we may sell or seek tenants for our properties at the same time as AmREIT and other AmREIT sponsored programs are selling and seeking tenants for properties. The Similar AmREIT Programs own and manage properties in the same geographical areas in which we own properties. Therefore, our properties may compete for tenants with other properties owned and managed by the Similar AmREIT Programs. Our General Partner faces conflicts of interest when evaluating tenant leasing opportunities for our properties and real properties owned and managed by AmREIT affiliated entities and these conflicts of interest may have a negative impact on our ability to attract and retain tenants.

When we determine to sell one of our properties, AmREIT has a right of first offer to acquire such property at market value, which may limit the amount of proceeds we would otherwise receive upon the sale of such property.

When we determine to sell one of our properties, our General Partner is required to provide AmREIT with notice and 15 days to provide us with a binding offer to purchase the property. If AmREIT elects to purchase one of our properties, the purchase price will be at fair market value, which will be determined by one or more independent appraisers. Because a property purchased by AmREIT pursuant to its right of first offer will never be marketed for sale to third parties, it is possible that the appraised market value of the property paid by AmREIT would be less than the price of the property we could receive from an unaffiliated third party, thereby reducing the return on our investments.

Our General Partner faces conflicts of interest in the incentive fee structure under the Partnership Agreement, which could result in actions that are not in the long-term best interests of our Limited Partners.

Under our Partnership Agreement, our General Partner is entitled to fees that are structured in a manner intended to provide incentives to our General Partner to perform in our best interests and in the best interests of our Limited Partners. However, because our General Partner is entitled to receive substantial compensation in the form of the asset management fee (described in greater detail in "Item 7. Certain Relationships and Related Transactions") regardless of our performance, our General Partner's interests are not wholly aligned with those of our Limited Partners. In that regard, our General Partner could be motivated to recommend additional investments not in our best interests in order for us to generate the specified levels of assets that would entitle our General Partner to fees. In addition, our General Partner's entitlement to fees upon the sale of our assets and to participate in sale proceeds could result in our General Partner recommending sales of our investments at the earliest possible time at which sales of investments would produce the level of return that would entitle our General Partner to compensation relating to such sales, even if continued ownership of those investments might be in our best long-term interest.

Risks Related to Our Corporate Structure

Our Limited Partners have limited control over changes in our policies and operations.

Our General Partner determines our major policies, including our policies regarding our investment objectives, acquisitions, dispositions, financing, growth, debt capitalization and distributions. Our General Partner may amend or revise these and other policies without a vote of the Limited Partners. Our General Partner's broad discretion in setting policies and our Limited Partners' inability to exert control over those policies increases the uncertainty and risks a Limited Partner has with an investment in us.

We may not have sufficient cash available from operations to pay distributions, and, therefore, distributions may include a return of capital.

Distributions payable to our Partners may include a return of capital, rather than a return on capital. The actual amount and timing of distributions are determined by our General Partner in its discretion and typically will depend on the amount of funds available for distribution, which will depend on items such as current and projected cash requirements and tax considerations. As a result, our distribution rate and payment frequency may vary from time to time. During the early stages of our operations, we may not have sufficient cash available from operations to pay distributions. Therefore, we may need to use proceeds from the Offering to make cash distributions, which may reduce the amount of proceeds available for investment and operations. All of the distributions made to our Limited Partners have constituted a return of capital and such distributions have exceeded our cash flow from operations by approximately \$329,000 as of December 31, 2007.

Our Limited Partners' investment return may be reduced if we are required to register as an investment company under the Investment Company Act.

We are not registered as an investment company under the Investment Company Act of 1940, as amended. If for any reason we were required to register as an investment company, we would have to comply with a variety of substantive requirements under the Investment Company Act imposing, among other things:

- limitations on capital structure;
- restrictions on specified investments;
- prohibitions on transactions with affiliates; and
- compliance with reporting, record keeping, voting, proxy disclosure and other rules and regulations that would significantly change our operations.

In order to maintain our exemption from regulation as an investment company under the Investment Company Act, we engage primarily in the business of investing in interests in real estate and have invested a significant portion of the proceeds of our Offering in properties within one year after the Offering ended. We have invested the uninvested proceeds of our Offering in government securities with low returns to avoid being required to register as an investment company under the Investment Company Act.

To maintain compliance with the Investment Company Act exemption, our General Partner may be required to impose limitations on our investment activities. In particular, our General Partner may limit the percentage of our assets that fall into certain categories specified in the Investment Company Act, which could result in us holding assets we otherwise might desire to sell and selling assets we otherwise might wish to retain. In addition, we may have to acquire additional assets that we might not otherwise have acquired or be forced to forgo investment opportunities that we would otherwise want to acquire and that could be important to our investment strategy. If we were required to register as an investment company but failed to do so, we would be prohibited from engaging in our business, and criminal and civil actions could be brought against us. In addition, our contracts would be unenforceable unless a court were to require enforcement, and a court could appoint a receiver to take control of us and liquidate our business.

Risks Related to Investments in Properties

Our future operating results will be affected by economic and regulatory changes that impact the real estate market in general.

We are subject to risks generally attributable to the ownership of property, including:

- changes in national, regional or local economic, demographic, or real estate market conditions;
- changes in supply of or demand for similar properties in an area;
- increased competition for property investments targeted by our investment strategy;
- bankruptcies, financial difficulties or lease defaults by our tenants;
- changes in interest rates and availability of financing; and
- changes in government rules, regulations and fiscal policies, including changes in tax, real estate, environmental and zoning laws.
- All of these factors are beyond our control. Any negative changes in these factors could affect our ability to meet our obligations and make distributions.

A property that incurs a vacancy could be difficult to sell or re-lease.

A property may incur a vacancy either by the continued default of a tenant under its lease or the expiration of one of our leases. In addition, certain of the properties we acquire may have some level of vacancy at the time of closing. Certain other properties may be specifically suited to the particular needs of a tenant and may become vacant. Therefore, we may have difficulty obtaining a new tenant for any vacant space we have in our properties. If the vacancy continues for a long period of time, we may suffer reduced revenues resulting in lower cash distributions. In addition, the resale value of the property could be diminished because the market value may depend principally upon the value of the leases of such property.

We may not have funding for future tenant improvements.

When a tenant at one of our properties does not renew its lease or otherwise vacates its space in one of our properties, it is likely that, in order to attract one or more new tenants, we will be required to expend substantial funds to construct new tenant improvements in the vacated space. Substantially all of the net proceeds from the Offering will be invested in properties and we do not anticipate that we will maintain permanent working capital reserves. We do not currently have an identified funding source to provide funds which may be required in the future for tenant improvements and tenant refurbishments in order to attract new tenants. If we do not establish sufficient reserves for working capital or obtain adequate secured financing to supply necessary funds for capital improvements or similar expenses, we may be required to defer necessary or desirable improvements to our properties. If we defer such improvements, the applicable properties may decline in value, and it may be more difficult for us to attract or retain tenants to such properties or the amount of rent we can charge at such properties may decrease. We cannot provide assurance that we will have any sources of funding available to us for repair or reconstruction of damaged property in the future.

Our investments in development and redevelopment properties are subject to certain risks which could harm our operating results and reduce the amount of funds available for distribution.

As a component of our strategy for realizing growth in the value of our investments, we seek to develop new properties and redevelop existing properties. As we develop and redevelop properties, such properties will be subject to a number of risks, including, but not limited to:

- the inability to obtain or delays in obtaining required zoning, occupancy and other governmental approvals;
- a lack of operating and rental history;
- inability to reach projected occupancy and rental rates;
- higher than estimated construction costs and cost overruns; and

the inability to obtain financing or to obtain financing on terms favorable to us.

Geographic concentration of our properties in the state of Texas could make us vulnerable to negative economic and other trends in Texas.

As a result of our General Partner's investment experience in the State of Texas, our current properties are all located in the State of Texas. We similarly anticipate that a majority of our future investments will also be in properties located in Texas. Therefore, an economic downturn in Texas could have an adverse effect on our financial condition and our ability to make distributions to our Limited Partners.

Our properties are subject to property taxes that may increase in the future, which could adversely affect our cash flow.

Our properties are subject to real and personal property taxes that may increase as tax rates change and as the properties are assessed or reassessed by taxing authorities. Certain of our leases provide that the property taxes, or increases therein, are charged to the lessees as an expense related to the properties that they occupy while other leases will generally provide that we are responsible for such taxes. In any case, as the owner of the properties, we are ultimately responsible for payment of the taxes to the applicable government authorities. If real property taxes increase, our tenants may be unable to make the required tax payments, ultimately requiring us to pay the taxes even if otherwise stated under the terms of the lease. If we fail to pay any such taxes, the applicable taxing authority may place a lien on the property and the property may be subject to a tax sale. In addition, we are generally responsible for real property taxes related to any vacant space.

Uninsured losses or premiums for insurance coverage relating to property may adversely affect Limited Partners' returns.

We attempt to adequately insure all of our properties against casualty losses. There are types of losses, generally catastrophic in nature, such as losses due to wars, acts of terrorism, earthquakes, floods, hurricanes, pollution or environmental matters that are uninsurable or not economically insurable, or may be insured subject to limitations, such as large deductibles or co-payments. Risks associated with potential terrorism acts could sharply increase the premiums we pay for coverage against property and casualty claims. Additionally, mortgage lenders sometimes require commercial property owners to purchase specific coverage against terrorism as a condition for providing mortgage loans. These policies may not be available at a reasonable cost, if at all, which could inhibit our ability to finance or refinance our properties. In such instances, we may be required to provide other financial support, either through financial assurances or self-insurance, to cover potential losses. Changes in the cost or availability of insurance could expose us to uninsured casualty losses. In the event that any of our properties incurs a casualty loss which is not fully covered by insurance, the value of our assets will be reduced by any such uninsured loss. In addition, we cannot provide assurance that funding will be available to us for repair or reconstruction of damaged property in the future.

We compete with numerous other parties or entities for property investments and tenants.

We compete with numerous other entities, including our affiliates, to buy properties and to attract tenants. Some of these entities have greater experience and financial strength. There is no assurance that we will be able to acquire additional properties or attract additional tenants on favorable terms, if at all. For example, our competitors may be willing to offer space at rental rates below our rates, causing us to lose existing or potential tenants and pressuring us to reduce our rental rates to retain existing tenants or convince new tenants to lease space at our properties. All of these factors could adversely affect our results of operations, financial condition, value of our investments and ability to pay distributions.

Delays in the acquisition, development and redevelopment of properties may have adverse effects on our Limited Partners' investment.

Delays we encounter in the selection, acquisition, development and redevelopment of properties could adversely affect the returns of our Limited Partners. If we acquire properties during the early stages of construction, it will typically take several months to complete construction and rent available space. Therefore, our Limited Partners could suffer delays in the distribution of income attributable to those particular properties. Delays in completion of construction could give tenants the right to terminate preconstruction leases for space at a newly developed project. We may incur additional risks when we make periodic progress payments or other advances to builders prior to completion of construction. Each of those factors could result in increased costs of a project or loss of our investment. In addition, we will be subject to normal lease-up risks relating to newly constructed projects. Furthermore, the price we agree to pay to acquire property will be based on our projections of rental income and expenses and estimates of the fair market value of property upon completion of construction. If our projections are inaccurate, we may pay too much for a property.

The terms of joint venture agreements or other joint ownership arrangements into which we have entered or may enter could impair our operating flexibility and our results of operations.

In connection with the purchase of real estate, we may enter into joint ventures with third parties, including affiliates of AmREIT. We may also purchase or develop properties in co-ownership arrangements with the sellers of the properties, developers or other persons. Through December 31, 2007, we have entered into eight joint venture arrangements. These structures involve participation in the investment by other parties whose interests and rights may not be the same as ours. Our joint venture partners may have rights to take some actions over which we have no control and may take actions contrary to our interests. Joint ownership of an investment in real estate involves risks not associated with direct ownership of real estate, including the following: a venture partner may at any time have economic or other business interests or goals which become inconsistent with our business interests or goals, including inconsistent goals relating to the sale of properties held in a joint venture or the timing of the termination and liquidation of the venture;

- a venture partner might become bankrupt and such proceedings could have an adverse impact on the operation of the partnership or joint venture; and
- actions taken by a venture partner might have the result of subjecting the property to liabilities in excess of those contemplated.

Under certain joint venture arrangements, neither venture partner may have the power to control the venture, and an impasse could be reached, which might adversely affect the joint venture and decrease potential returns to our Partners. If we have a right of first refusal or buy/sell right to buy out a venture partner, we may be unable to finance such a buy-out or we may be forced to exercise those rights at a time when it would not otherwise be in our best interest to do so. If our interest is subject to a buy/sell right, we may not have sufficient cash, available borrowing capacity or other capital resources to allow us to purchase an interest of a venture partner subject to the buy/sell right, in which case we may be forced to sell our interest when we would otherwise prefer to retain our interest. In addition, we may not be able to sell

our interest in a joint venture on a timely basis or on acceptable terms if we desire to exit the venture for any reason, particularly if our interest is subject to a right of first refusal of our venture partner.

Costs of complying with governmental laws and regulations may adversely affect our income and the cash available for distribution.

All property investments and the operations conducted in connection with such investments are subject to federal, state and local laws and regulations relating to environmental protection and human health and safety. Some of these laws and regulations may impose joint and several liability on customers, owners or operators for the costs to investigate or remediate contaminated properties, regardless of fault or whether the acts causing the contamination were legal.

Under various federal, state and local environmental laws, a current or previous owner or operator of real property may be liable for the cost of removing or remediating hazardous or toxic substances on such property. Such laws often impose liability whether or not the owner or operator knew of, or was responsible for, the presence of such hazardous or toxic substances. In addition, the presence of hazardous substances, or the failure to properly remediate these substances, may adversely affect our ability to sell, rent or pledge such property as collateral for future borrowings. Environmental laws also may impose restrictions on the manner in which property may be used or businesses may be operated. Some of these laws and regulations have been amended so as to require compliance with new or more stringent standards as of future dates. Compliance with new or more stringent laws or regulations or stricter interpretation of existing laws may require us to incur material expenditures. Future laws, ordinances or regulations may impose material environmental liability. Additionally, our tenants' operations, the existing condition of land when we buy it, operations in the vicinity of our properties, such as the presence of underground storage tanks, or activities of unrelated third parties may affect our properties. In addition, there are various local, state and federal fire, health, life-safety and similar regulations with which we may be required to comply, and which may subject us to liability in the form of fines or damages for noncompliance. In connection with the acquisition and ownership of our properties, we may be exposed to such costs in connection with these regulations. The cost of defending against environmental claims, of any damages or fines we must pay, of compliance with environmental regulatory requirements or of remediating any contaminated property could materially adversely affect our business, assets or results of operations and, consequently, amounts available for distribution to our Partners.

If a tenant declares bankruptcy, we may be unable to collect balances due under relevant leases.

Any of our tenants, or any guarantor of a tenant's lease obligations, could be subject to a bankruptcy proceeding pursuant to Title 11 of the bankruptcy laws of the United States. Such a bankruptcy filing would bar all efforts by us to collect pre-bankruptcy debts from these entities or their properties, unless we receive an enabling order from the bankruptcy court. Post-bankruptcy debts would be paid currently. If a lease is assumed, all pre-bankruptcy balances owing under it must be paid in full. If a lease is rejected by a tenant in bankruptcy, we would have a general unsecured claim for damages. If a lease is rejected, it is unlikely we would receive any payments from the tenant because our claim is capped at the rent reserved under the lease, without acceleration, for the greater of one year or 15.0% of the remaining term of the lease, but not greater than three years, plus rent already due but unpaid. This claim could be paid only in the event funds were available, and then only in the same percentage as that realized on other unsecured claims.

A tenant or lease guarantor bankruptcy could delay efforts to collect past due balances under the relevant leases, and could ultimately preclude full collection of these sums. Such an event could cause a decrease or cessation of rental payments that would mean a reduction in our cash flow and the amount available for distributions to our Partners. In the event of a bankruptcy, we cannot provide assurance that the tenant or its trustee will assume our lease. If a given lease, or guaranty of a lease, is not assumed, our cash flow and the amounts available for distributions to our Partners may be adversely affected.

Risks Associated with Retail Properties

Our properties primarily consist of retail properties. Our performance, therefore, is linked to the market for retail space generally.

The market for retail space has been and could be adversely affected by weaknesses in the national regional and local economies, the adverse financial condition of some large retailing companies, the ongoing consolidation in the retail sector, the excess amount of retail space in a number of markets and competition for tenants with other shopping centers in our markets. Because our properties primarily consist of retail tenants, a downturn in the market for retail space generally may hurt our performance.

In addition, some of our leases provide for base rent plus contractual base rent increases. Other leases include a percentage rent clause for additional rent above the base amount based upon a specified percentage of the sales our tenants generate. Under those leases which contain percentage rent clauses, our revenue from tenants may increase as the sales of our tenants increase. Generally, retailers face declining revenues during downturns in the economy. As a result, the portion of our revenue which we may derive from percentage rent leases could decline upon a general economic downturn.

Our retail tenants will face competition from numerous retail channels, which may reduce our profitability and ability to pay distributions.

Retailers at our properties face continued competition from, discount or value retailers, factory outlet centers, wholesale clubs, mail order catalogues and operators, television shopping networks, and shopping via the Internet. Such competition could adversely affect our tenants and, consequently, our revenues and funds available for distribution.

Our future revenue will be impacted by the success and economic viability of our anchor retail tenants. Our reliance on single or significant tenants in certain buildings may decrease our ability to lease vacated space and adversely affect the returns on our Limited Partners' investment.

In the retail sector, a tenant occupying all or a large portion of the gross leasable area of a retail center, commonly referred to as an anchor tenant, may become insolvent, may suffer a downturn in business, or may decide not to renew its lease. Any of these events would result in a reduction or cessation in rental payments to us and would adversely affect our financial condition. A lease termination by an anchor tenant could result in lease terminations or reductions in rent by other tenants whose leases permit cancellation or rent reduction if another tenant's lease is terminated. We may own properties where the tenants may have rights to terminate their leases if certain other tenants are no longer open for business. These "co-tenancy" provisions may also exist in some leases where we own a portion of a retail property and one or more of the anchor tenants leases space in that portion of the center not owned or controlled by us. If such tenants were to vacate their space, tenants with co-tenancy provisions would have the right to terminate their leases with us or seek a rent reduction from us. In such event, we may be unable to re-lease the vacated space. Similarly, the leases of some anchor tenants may permit the anchor tenant to transfer its lease to another retailer. The transfer to a new anchor tenant could cause customer traffic in the retail center to decrease and thereby reduce the income generated by that retail center. A lease transfer to a new anchor tenant could also allow other tenants to make reduced rental payments or to terminate their leases at the retail center. In the event that we are unable to re-lease the vacated space to a new anchor tenant, we may incur additional expenses in order to re-model the space to be able to re-lease the space to more than one tenant.

Risks Associated with Debt Financing

We incur mortgage indebtedness and other borrowings, which increases our business risks, potentially hinders our ability to make distributions and potentially decreases the value of our Limited Partners' investment.

We generally finance a portion of the purchase price of properties by borrowing funds. High debt levels cause us to incur higher interest charges, which results in higher debt service payments and could be accompanied by restrictive covenants. If there is a shortfall between the cash flow from a property and the cash flow needed to service mortgage debt on that property, then the amount available for distributions to Limited Partners may be reduced. In addition, incurring mortgage debt increases the risk of loss since defaults on indebtedness secured by a property may result in lenders initiating foreclosure actions. In that case, we could lose the property securing the loan that is in default, thus reducing the value of our Limited Partners' investment. For tax purposes, a foreclosure on any of our properties will be treated as a sale of the property for a purchase price equal to the outstanding balance of the debt secured by the mortgage. If the outstanding balance of the debt secured by the mortgage exceeds our tax basis in the property, we will recognize taxable income on foreclosure, but we would not receive any cash proceeds. We may give full or partial guarantees to lenders of mortgage debt to the entities that own our properties. When we give a guaranty on behalf of an entity that owns one of our properties, we will be responsible to the lender for satisfaction of the debt if it is not paid by such entity. If any mortgage contains cross-collateralization or cross-default provisions, a default on a single property could affect multiple properties. If any of our properties are foreclosed upon due to a default, our ability to pay cash distributions to our Partners will be adversely affected.

Higher mortgage rates may make it difficult for us to finance or refinance properties, which could reduce the number of properties we can acquire and the amount of cash distributions we can make to our Partners.

If mortgage debt is unavailable at reasonable rates in the future, we may not be able to finance the initial purchase of properties on terms we deem reasonable. In addition, when we place mortgage debt on properties, we run the risk of being unable to refinance such debt when the loans come due, or of being unable to refinance on favorable terms. If interest rates are higher when we refinance debt, our income could be reduced. We may be unable to refinance debt at appropriate times, which may require us to sell properties on terms that are not advantageous to us, or could result in the foreclosure of such properties. If any of these events occur, our cash flow would be reduced. This, in turn, would reduce cash available for distribution to our Partners and may hinder our ability to raise additional capital by issuing securities or by borrowing additional money.

Increases in interest rates could increase the amount of our debt payments and adversely affect our ability to make distributions to our Partners.

Interest we pay on our debt obligations reduces cash available for distributions. Variable rate debt causes increases in interest rates to increase our interest costs, which would reduce our cash flows and our ability to make distributions to our Partners. In addition, if we need to repay existing debt during periods of rising interest rates, we could be required to liquidate one or more of our investments in properties at times which may not permit realization of the maximum return on such investments.

Lenders may require us to enter into restrictive covenants relating to our operations, which could limit our ability to make distributions to our Partners.

When providing financing, a lender may impose restrictions on us that affect our distribution and operating policies and our ability to incur additional debt. Loan documents we enter into may contain covenants that limit our ability to further mortgage the property or discontinue insurance coverage. In addition, loan documents may limit our ability to terminate certain operating or lease agreements related to the property. These or other limitations may adversely affect our flexibility and our ability to achieve our investment objectives.

Federal Income Tax Risks

Failure to qualify as a partnership would eliminate the anticipated tax benefits associated with an investment in us.

We are organized and operated in a manner to maintain our status as a partnership for federal income tax purposes. We have not sought, will not seek and probably could not obtain, an advance ruling from the Internal Revenue Service, or IRS, as to our status as a partnership for federal income tax purposes. If the IRS were to successfully challenge our status as a partnership, we would be taxable as a corporation. In such event, this would substantially reduce our cash available to make distributions to the Partners.

The IRS could challenge our and our Limited Partners' ability to realize certain favorable tax treatment.

We have not requested, and do not expect to request, a ruling from the IRS with respect to any tax aspect of an investment in us. The IRS could challenge the Limited Partners' ability to realize certain favorable tax treatment upon an audit of us. Any adjustment resulting from an audit by the IRS could result in adjustments to the Limited Partners' tax returns and could lead to an examination of other items in such returns unrelated to us or an examination of prior tax returns. Moreover, the Limited Partners could incur substantial legal and accounting costs should the IRS challenge a position taken by us on our tax returns regardless of the outcome of such a challenge.

Additionally, the IRS may contend that certain fees and payments which we deduct should in fact be deductible over a longer period of time, or that such payments are excessive and may not be deducted. Moreover, the IRS may contend that such payments should be treated as non-deductible partnership distributions or syndication fees. If the IRS was successful in any of these contentions, our forecasted taxable income and our federal income tax liability could be increased.

ITEM 2. FINANCIAL INFORMATION

Management's Discussion and Analysis or Plan of Operation

The following discussion and analysis should be read in conjunction with the accompanying consolidated financial statements and notes thereto. See also "Cautionary Note Regarding Forward Looking Statements" preceding Item 1.

Overview

We are a Delaware limited partnership formed on October 10, 2006 to acquire, develop and operate, directly or indirectly through joint venture arrangements, a portfolio of commercial real estate consisting primarily of multi-tenant shopping centers and mixed-use developments throughout the Southern and Southwestern United States. We focus on properties characterized by high automobile traffic counts, high populations, high household incomes and limited opportunities for competition.

We have no employees and are managed by AmREIT Monthly Income & Growth IV Corporation, our General Partner, pursuant to our Partnership Agreement. Our General Partner is a subsidiary of AmREIT, a Texas real estate investment trust. AmREIT's common stock is publicly traded on the American Stock Exchange under the symbol "AMY." In addition to owning its general partner interest in us for which it contributed \$1,000, AmREIT contributed \$800,000 to us in exchange for Units. The remaining Units are held by other Limited Partners. We qualify as a partnership for federal income tax purposes.

As of December 31, 2007, we have ownership interests in four real estate properties. Although we exercise significant influence over the activities of these properties, we do not have a controlling financial interest in them. Accordingly, all of our real estate investments are reported under the equity method of accounting pursuant to U.S. generally accepted accounting principles. We allocate income and loss from these properties based on our pro rata ownership percentage in each property. As a result, our operating results and cash flows are primarily influenced by rental income from the properties in which we have invested and interest expense on the related property acquisition indebtedness. During the year ended December 31, 2007 and the period from October 10, 2006 (inception) through December 31, 2006, we did not own any consolidated properties. Therefore we earned rental income only indirectly during those periods which was generated by the properties in which we own an interest. Interest income was our only source of revenue for the year ended December 31, 2007. As of December 31, 2007, our three existing retail properties were 75% leased, and the debt leverage ratio of our portfolio was approximately 59%, with 55% debt carrying a fixed rate of interest and 45% carrying a variable rate debt interest.

We commenced our principal operations on December 8, 2006 when we acquired our first interest in a property. On January 12, 2007, we raised the minimum offering of \$1.0 million pursuant to the terms of our Offering Memorandum and issued 40 Units to the initial investors. Our Offering was closed as of April 1, 2008, and, as of that date, we had raised aggregate gross proceeds of approximately \$50.0 million from the sale of 2,000 Units.

The four properties in which we own an interest comprise approximately 612,000 rentable square feet. We used \$18.3 million in proceeds from the Offering to acquire interest in these properties. The properties are located in highly populated suburban communities in Texas.

Our Units were sold pursuant to exemptions from registration under the Securities Act and are not currently listed on any securities exchange. The Units will be transferable only if we register them under applicable securities laws, and such registration is not expected. We do not anticipate that any public market for the Units will develop.

Summary of Critical Accounting Policies

Our results of operations and financial condition, as reflected in the accompanying consolidated financial statements and related footnotes, are subject to management's evaluation and interpretation of business conditions, retailer performance, changing capital market conditions and other factors, which could affect the ongoing viability of our tenants. Management believes the most critical accounting policies in this regard are revenue recognition, regular evaluation of whether the value of a real estate asset has been impaired, allowance for uncollectible accounts and accounting for real estate acquisitions. We evaluate our assumptions and estimates on an on-going basis. We base our estimates on historical experience as well as various other assumptions that we believe to be reasonable under the circumstances. Actual results could vary from those estimates, perhaps in material adverse ways, and those estimates could be different under different assumptions or conditions.

Investment in Non-consolidated Entities

As of December 31, 2007, we have ownership interests in four real estate properties. As discussed above, we do not have a controlling financial interest in these investments. Accordingly, all of our real estate investments are reported under the equity method of

accounting pursuant to U.S. generally accepted accounting principles. Certain of the significant accounting policies below are applicable specifically to property-level reporting and represent policies that are therefore primarily relevant at the investee entity level as of December 31, 2007.

We own interests in the four properties through joint ventures. The joint ventures lease space to tenants under agreements with varying terms. The majority of the leases are accounted for as operating leases with revenue being recognized on a straight-line basis over the terms of the individual leases. When our joint ventures acquire a property, the term of existing leases is considered to commence as of the acquisition date for the purposes of this calculation.

Accrued rents are included in tenant receivables. Revenue from tenant reimbursements of taxes, maintenance expenses and insurance is recognized in the period the related expense is recorded. Additionally, certain of the lease agreements contain provisions that grant additional rents based on tenants' sales volumes (contingent or percentage rent). We defer the recognition of contingent or percentage rental income until the specific targets as defined in lease agreements that trigger the contingent or percentage rental income are achieved. Cost recoveries from tenants are included in rental income in the period the related costs are incurred.

Joint Ventures Accounting Policies

The joint ventures in which we invest account for real estate acquisitions pursuant to SFAS No. 141, *Business Combinations* (SFAS No. 141). Accordingly, they allocate the purchase price of the acquired properties to tangible and intangible assets as well as liabilities acquired. The joint ventures compute depreciation using the straight-line method over an estimated life of up to 50 years for buildings, up to 20 years for site improvements and over the term of the lease for tenant improvements. The intangible assets are being amortized over the remaining term of their respective leases.

The joint ventures in which we invest assess their properties for impairment whenever events or changes in circumstances indicate that the carrying amount of the assets, including accrued rental income, may not be recoverable through operations. We determine whether an impairment in value occurred by comparing the estimated future cash flows (undiscounted and without interest charges), including the residual value of the property, with the carrying value of the individual property. If impairment is indicated, a loss will be recorded for the amount by which the carrying value of the asset exceeds its fair value. We do not believe that any of our joint venture investments have carrying values that are in excess of their respective fair values as of December 31, 2007.

Recently Issued Accounting Pronouncements

In September 2006, the Financial Accounting Standards Board ("FASB") issued SFAS No. 157, *Fair Value Measurement* (SFAS No. 157). SFAS No. 157 defines fair value, establishes a framework for measuring fair value and requires enhanced disclosures about fair value measurements. SFAS No. 157 requires companies to disclose the fair value of its financial instruments according to a fair value hierarchy. Additionally, companies are required to provide certain disclosures regarding instruments within the hierarchy, including a reconciliation of the beginning and ending balances for each major category of assets and liabilities. SFAS No. 157 is effective for our fiscal year beginning January 1, 2008. The adoption of SFAS No. 157 is not expected to have a material effect on our results of operations or financial position.

In February 2007, the FASB issued SFAS No. 159, *The Fair Value Option for Financial Assets and Financial Liabilities* (SFAS No. 159). SFAS No. 159 expands opportunities to use fair value measurement in financial reporting and permits entities to choose to measure many financial instruments and certain other items at fair value. This statement is effective for fiscal years beginning after November 15, 2007. We currently do not plan to measure any eligible financial assets and liabilities at fair value under the provisions of SFAS No. 159.

In December 2007, the FASB issued SFAS No. 141 (Revised 2007), *Business Combinations* (SFAS No. 141R). SFAS No. 141R will change the accounting for business combinations. Under SFAS No. 141R, an acquiring entity will be required to recognize all the assets acquired and liabilities assumed in a transaction at the acquisition-date fair value with limited exceptions. SFAS No. 141R applies prospectively to business combinations for which the acquisition date is on or after the beginning of the first annual reporting period beginning on or after December 15, 2008. We are currently evaluating the potential impact of SFAS No. 141R on our financial position and results of operations beginning for fiscal year 2009.

In December 2007, the FASB issued SFAS No. 160, *Noncontrolling Interests in Consolidated Financial Statements—an amendment of ARB No. 51* (SFAS No. 160). SFAS No. 160 establishes accounting and reporting standards for the noncontrolling interest in a subsidiary and for the deconsolidation of a subsidiary. It clarifies that a noncontrolling interest in a subsidiary is an ownership interest in the consolidated entity that should be reported as equity in the consolidated financial statements. SFAS No. 160 is effective for fiscal years, and interim periods within those fiscal years, beginning on or after December 15, 2008. SFAS No. 160 requires retroactive adoption of the presentation and disclosure requirements for existing minority interests. All other requirements of SFAS No. 160 shall be applied prospectively. We are currently evaluating the potential impact of the adoption of SFAS No. 160 on our consolidated financial statements.

Factors Which May Influence Results of Operations

Income (Loss) from Non-Consolidated Entities

We have ownership interests in four real estate properties, all of which are reported under the equity method of accounting. Accordingly, substantially all of our real estate operations are reflected as income or loss from non-consolidated entities which represents our portion of that property's results of operations. The amount of rental income generated by the properties in which we have invested depends principally on the ability of the property managers to maintain the occupancy rates of currently leased space and to lease currently available space and space available from unscheduled lease terminations at levels not less than the existing rental rates. Negative trends in one or more of these factors could adversely affect our rental income in future periods.

Expenses

Our expenses could increase due to the costs incurred in order to comply with the requirements of being a public reporting company, among other things.

Results of Operations

We commenced our principal operations on December 8, 2006, when we acquired our first interest in a property. During 2007, we acquired interests in three additional properties through separate joint ventures, on August 31, 2007, November 20, 2007 and December 28, 2007. We report our investments in joint ventures under the equity method of accounting given our ability to exercise significant influence over them. We will account for any direct property acquisitions as purchases and the results of their operations will be included in our consolidated financial statements from the respective dates of acquisition.

General and Administrative

General and administrative expenses increased to approximately \$156,000 during the year ended December 31, 2007. This increase was primarily due to us not commencing our principal operations until 2007.

Asset Management Fees - Related Party

Asset management fees – related party increased to approximately \$204,000 during the year ended December 31, 2007 compared to zero dollars for the period from October 10, 2006 (inception) through December 31, 2006. Asset management fees have increased commensurate with the increase in assets under management.

Legal and Professional Fees

Legal and professional fees increased approximately \$102,000 to approximately \$119,000 during the year ended December 31, 2007 compared to approximately \$17,000 for the period from October 10, 2006 (inception) through December 31, 2006. This increase was due to fees associated with audit and tax services related to the full 2007 year versus the shorter 2006 period as well as to legal fees associated with the formation of certain of our non-consolidated entities.

Interest and Other Income

Interest and other income increased to approximately \$523,000 during the year ended December 31, 2007 compared to zero dollars for the period from October 10, 2006 (inception) through December 31, 2006. This increase was due to earning interest on a higher balance of investable funds during 2007 as we continued our capital-raising efforts. We invest our excess cash in short-term investments or overnight funds until properties suitable for acquisition can be identified and acquired.

Interest Expense

Interest expense increased approximately \$75,000 to approximately \$101,000 during the year ended December 31, 2007 compared to approximately \$26,000 for the period from October 10, 2006 (inception) through December 31, 2006. This increase was primarily due to a note payable to a related party that incurred three months of interest during 2007 as compared to a partial month of interest during 2006. In early 2007, this note payable was paid in full.

Loss From Non-Consolidated Entities

Loss from non-consolidated entities was a loss of \$1.1 million for the year ended December 31, 2007 compared to a \$28,000 loss for the period from October 10, 2006 (inception) through December 31, 2006. This income or loss represents our ownership portion of our joint ventures' net income or loss for the period.

Liquidity and Capital Resources

We expect to meet our short-term liquidity requirements through cash on-hand and net cash provided by distributions from joint ventures. We expect to meet our long-term liquidity requirements through proceeds from secured or unsecured financings from banks and other lenders, the selective and strategic sale of properties and net cash flows from operations. We expect that our primary uses of capital will be for investment in real estate properties and related improvements, the payment of operating expenses, including interest expense on any outstanding indebtedness, and the payment of distributions to our Partners. As we sell properties during our operating period, we plan to strategically reinvest the proceeds from such sales rather than distributing the proceeds to our Partners.

At December 31, 2007 and December 31, 2006, our cash and cash equivalents totaled approximately \$19.3 million and approximately \$796,000, respectively. Cash flows provided by (used in) operating activities, investing activities and financing activities for the year ended December 31, 2007 and for the period from October 10, 2006 (inception) through December 31, 2006, are as follows (in thousands):

	2007	2006		
Operating activities	\$ 360	\$	714	
Investing activities	\$ (14,547)	\$	(5,244)	
Financing activities	\$ 32,740	\$	5,326	

Net cash provided by operating activities decreased approximately \$354,000 to approximately \$360,000 for the year ended December 31, 2007 compared to a net cash provided by operating activities of approximately \$714,000 for the period from October 10, 2006 (inception) through December 31, 2006. Loss before the effect of loss from non-consolidated entities stayed relatively constant at \$61,000 for the year ended December 31, 2007 versus \$43,000 for the period from October 10, 2006 (inception) through December 31, 2006. The primary driver of the reduction in operating cash flows was a reduction in working capital cash flows of \$336,000. This reduction was driven primarily by increased accounts receivable and cash outflows on accounts payable -related party of \$113,000 and \$220,000, respectively.

Net cash used in investing activities increased approximately \$9.3 million to approximately \$14.5 million for the year ended December 31, 2007 compared to approximately \$5.2 million for the period from October 10, 2006 (inception) through December 31, 2006. This increase was attributable to our investments in Woodlake Square, Westheimer Gessner and Cambridge Holcombe in 2007 as well as an earnest money deposit made during 2007 related to Shadow Creek Ranch, an investment that was consummated in 2008.

Net cash provided by financing activities increased approximately \$27.4 million to approximately \$32.8 million for the year ended December 31, 2007 compared to approximately \$5.3 million for the period from October 10, 2006 (inception) through December 31, 2006. This increase was primarily due to an increase of approximately \$38.7 million in capital raised, net of issuance costs, during 2007 as compared to 2006. This increase was partially offset by a reduction in cash flows from notes payable of \$9.9 million as well as an increase in distributions to our Partners of \$1.4 million which represent a return of capital.

Until we acquire properties, we invest all excess cash in short-term investments or overnight funds. This investment strategy allows us to offset a portion of the interest costs from our fixed-rate mortgage loans and provides us with the liquidity to acquire properties at such time as those suitable for acquisition are located.

We believe that inflation has a minimal effect on our income from operations. We expect that increases in store sales volumes due to inflation, as well as increases in the consumer price index, may contribute to capital appreciation of our properties. These factors, however, also may have an adverse impact on the operating margins of the tenants of the properties.

Indemnification of General Partner

Our Limited Partners, within the limits of their respective capital contributions and to the extent of their allocable portion of Partnership assets, agree to indemnify and hold our General Partner harmless from and against actual and reasonable third-party claims or lawsuits arising out of our activities and operations, except those instances in which our General Partner failed to adhere to its fiduciary obligations to us, or acted with gross negligence or willful or wanton misconduct. Amounts paid to indemnify our General Partner may be recouped to the extent such payments relate to a third-party claim or lawsuit arising out of a breach by our General Partner of its fiduciary obligations to us.

Conflicts of Interest

Our General Partner is a subsidiary of AmREIT, a Texas real estate investment trust. Affiliates of AmREIT act as sponsor, general partner or advisor to various private real estate programs. As such, there are conflicts of interest where AmREIT or its affiliates, while serving in the capacity as sponsor, general partner or advisor for another AmREIT-sponsored program, may be in competition with us in connection with property acquisitions, property dispositions and property management. The compensation arrangements between affiliates of AmREIT

and these other AmREIT real estate programs could influence our General Partner's management of us. See "Item 1. Business – Conflicts of Interest."
Off Balance Sheet Arrangements
We had no off balance sheet arrangements as of December 31, 2007.

~18~

ITEM 3. PROPERTIES

Overview

During the period from October 10, 2006 (inception) through December 31, 2007, we acquired interests in four properties. We have investments in each property through joint venture arrangements. We have included a description of the types of real estate in which we have invested and will invest, and a summary of our investment policies and limitations on investment and the competitive conditions in which we operate under "Item 1. Business" above. We believe our properties are suitable for their intended use and are adequately insured.

Dorgantago

As of December 31, 2007, we owned interests in the following properties, all of which are located in Texas:

Property	Location	Square Footage	Percentage Leased		Annualized Gross Base Rent (1)	of Total Annualiz Gross Ba	zed ase
Casa Linda Plaza(2)	Dallas	324,638	82	%	\$3,207,000	55	%
Woodlake Square (2)	Houston	205,522	73	%	1,985,000	34	%
Westheimer Gessner (2)	Houston	82,120	50	%	627,000	11	%
Cambridge Holcombe (2) (3)	Houston						
Total		612,280			\$5,819,000	100	%

- Annualized gross base rent represents base rents in place on leases with rent having commenced as of December 31, 2007 and does not reflect straight-line rent or other adjustments under generally accepted accounting principles.
- (2) Property is owned through a joint venture that is not consolidated in our financial statements.
- (3) Property is vacant land and under development.

Description of Our Real Estate Investments

Casa Linda Plaza

On December 8, 2006, we acquired a 50% interest in the Casa Linda Plaza property, a 324,638 square foot retail shopping center located in Dallas, Texas, through a joint venture arrangement with an affiliate of our General Partner, MIG III. Our joint venture partner owns the remaining 50% interest in the joint venture. The property was purchased from an unaffiliated third party. We used proceeds from the Offering for our contributed equity and the joint venture obtained a 7-year mortgage loan from Morgan Stanley Mortgage Capital, Inc. to fund the acquisition of the Casa Linda Plaza property. The loan was in the amount of \$38.0 million, bears an annual interest rate of 5.48% and is interest-only until maturity.

The Casa Linda Plaza property is in average condition for a property of its type and age. However, substantial renovation is required to update the property. The Casa Linda joint venture has commenced the first phase of the property renovation and estimates that the renovation will last between 24 to 36 months from start to finish and will cost approximately \$6.8 million. The renovation will be financed through cash flows from property operations as well as through cash contributions from MIG III and from us in proportion to our respective ownership interests. The planned renovation will allow the property to maintain its historical character and prominence in the community while updating the property's features.

Albertson's is the largest tenant occupying 59,561 square feet with a lease scheduled to expire in July 2016. Additional tenants are Petco, 24 Hour Fitness, Starbucks, Wachovia, Chili's, Blockbuster, Washington Mutual, El Fenix and Guaranty Federal Bank. The property was originally built between 1946 and 1949.

Woodlake Square

On August 31, 2007, we acquired a 40% interest in the Woodlake Square property, a 205,522 square foot retail shopping center located in Houston, Texas, through a joint venture arrangement with affiliates of our General Partner, MIG III and ARIC. Our joint venture partners own the other 60% interest in the joint venture. The property was purchased from an unaffiliated third party. We used proceeds from the Offering for our contributed equity and the joint venture obtained a 3-year mortgage loan from Frost Bank to fund the acquisition of the Woodlake Square property. The loan was in the amount of \$23.8 million, all of which represents a variable rate obligation.

The Woodlake Square property is in average condition for a property of its type and age. We intend to redevelop the property in order to maximize the return on our investment. We intend to redevelop the property in order to maximize the return on our investment. The joint venture is in the planning stages of a significant redevelopment which will likely include multi-family residential, office and retail components.

Randall's is the largest tenant occupying 56,430 square feet with a lease scheduled to expire in July 2013. Additional tenants are Walgreen's, Hooters, Jos. A. Bank Clothiers, Ragin Cajun, Woodlake Children's Center and Amazon Grill.

Westheimer Gessner

On November 21, 2007, we acquired a 40% interest in the Westheimer Gessner property, a 82,120 square foot retail shopping center located in Houston, Texas, through a joint venture arrangement with affiliates of our General Partner, MIG III and ARIC. Our joint venture partners own the other 60% interest in the joint venture. The property was purchased from an unaffiliated third party. We used proceeds from the Offering for our contributed equity and the joint venture paid cash for the acquisition of the Westheimer Gessner property.

The Westheimer Gessner property is in average condition for a property of its type and age. We intend to redevelop the property in order to maximize our return on investment. We intend to redevelop the property in order to maximize our return on investment. The joint venture is in the planning stages of a significant redevelopment which will likely include a multi-family residential component and a retail component.

Border's is the single tenant occupying 41,060 square feet with a lease scheduled to expire in February 2010.

Cambridge Holcombe

On December 31, 2007, we acquired a 50% interest in the Cambridge Holcombe property through a joint venture arrangement with an unaffiliated third party. This vacant land is located in the Texas Medical Center in Houston, Texas. Our joint venture partner owns the other 50% interest in the joint venture. The property was purchased from an unaffiliated third party. We used proceeds from the Offering for our contributed equity and the joint venture paid cash for the acquisition of the Cambridge Holcombe property. We are in the initial stages of planning this property's development.

Portfolio Information

The following table shows our five highest tenant industry concentrations for our property portfolio as of December 31, 2007:

		Percentage	
Total Number of	Annualized Gross Base	Total Annualized Gross Base	
22	\$1,594,596	27.4	%
19	\$1,525,177	26.2	%
4	\$928,410	16.0	%
15	\$449,129	7.7	%
5	\$305,923	5.3	%
	Number of Tenants 22 19 4 15	Number of Tenants Gross Base Rent 22 \$1,594,596 19 \$1,525,177 4 \$928,410 15 \$449,129	Total Number of Tenants Annualized Gross Base Rent Gross Base Rent 22 \$1,594,596 27.4 19 \$1,525,177 26.2 4 \$928,410 16.0 15 \$449,129 7.7

The following table shows lease expirations for the properties that we own an investment in as of December 31, 2007, during each of the next ten years and thereafter, assuming no exercise of renewal options or termination rights:

Year of Lease Expiration	Total Number of Leases	Square Feet	Annualized Gross Base Rent	Percentage of Total Annualized Gross Base Rent	Percentage of Total Square Feet
2008	30	101,087	\$ 1,345,286	23.12%	22.05%
2009	20	58,318	\$ 888,106	15.26%	12.72%
2010	17	88,989	\$ 1,292,422	22.21%	19.41%
2011	11	35,433	\$ 710,017	12.20%	7.73%
2012	10	21,293	\$ 448,827	7.71%	4.64%
2013	5	67,144	\$ 387,756	6.66%	14.65%
2014 - 2015	-	-	\$ -	-	-
2016	1	59,561	\$ 287,381	4.94%	12.99%
2017	-	-	\$ -	-	-
thereafter	4	26,646	\$ 459,600	7.90%	5.81%

As of December 31, 2007, we had invested \$18.3 million of the proceeds from the Offering in the real property described above through joint ventures. The joint ventures have obtained an aggregate of \$69.6 million in mortgage loans for their acquisition of the properties

~20~

in which we have invested. We anticipate acquiring additional properties until we have fully invested the proceeds from the Offering. As of

December 31, 2007, we have not disposed of any property.

ITEM 4. SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT

We know of no person (including a "group" as that term is used in Section 13(d)(3) of the Exchange Act) who is the beneficial owner of more than five percent of our Units.

We have no officers or directors. Our General Partner owns our sole general partner interest and also owns Units. No person who is an officer of our General Partner as of the date of this filing owns any direct interest in the Partnership. AmREIT, the sole shareholder of our General Partner, has assigned the economic interest in 28.5% of our General Partner to certain of its management team. The address of the officers of our General Partner is 8 Greenway Plaza, Suite 1000, Houston, Texas 77046.

As of December 31, 2007, there were 1,796 Units issued and outstanding owned by 748 investors.

ITEM 5. DIRECTORS AND EXECUTIVE OFFICERS

We have no directors or executive officers. We are managed by our General Partner. Our General Partner does not have any employees and relies upon the personnel of AmREIT and its affiliates to perform services for us. Our Partnership Agreement provides that we will continue until December 31, 2026, unless sooner terminated.

The following table sets forth certain information regarding the officers and director of our General Partner, all of whom are officers of AmREIT and expected to make a significant contribution to us.

Name	Age	Position
H. Kerr Taylor	57	Chairman, President and Chief Executive Officer
Chad C. Braun	36	Vice President, Chief Financial Officer, Treasurer
		and Secretary
Brett P. Treadwell	38	Vice President – Finance

H. Kerr Taylor serves as the Chairman, President and CEO of our General Partner and is the sole director of our General Partner. He is the founder of AmREIT and serves as its Chairman of the Board, Chief Executive Officer and President. For over 22 years he has guided the growth of AmREIT and its predecessors. His primary responsibilities include overseeing strategic initiatives as well as building, coaching and leading AmREIT's team of professionals. Mr. Taylor has over 30 years of experience within the real estate industry, and has participated in over 300 transactions involving brokerage, development and management of premier real estate projects. He attended Trinity University, graduating with a Bachelor of Arts degree and then attended Southern Methodist University where he received his Masters Degree in Business Administration. Mr. Taylor also attended law school at South Texas College of Law where he received his Doctor of Jurisprudence. Mr. Taylor is chairman of the board of Pathways for Little Feet and serves as a board member of Life House, Inc., Uptown District and as an Elder of First Presbyterian Church. Mr. Taylor is a lifetime member of the International Council of Shopping Centers and Urban Land Institute and is a member of the Texas Bar Association.

Chad C. Braun serves as the Vice President, CFO, Secretary and Treasurer of the General Partner and as AmREIT's Executive Vice President, Chief Financial Officer, Treasurer and Secretary. Mr. Braun is responsible for corporate finance, equity capital markets, debt structuring and placement, investor relations, accounting, SEC reporting, and he oversees investment sponsorship and product creation. Mr. Braun has over twelve years of accounting, financial and real estate experience, and prior to joining AmREIT in 1999, he served as a manager in the real estate advisory services group at Ernst & Young, LLP. He has provided extensive consulting and audit services, including financial statement audits, portfolio acquisition and disposition, portfolio management, merger integration and process improvement, financial analysis and capital markets and restructuring transactions, to a number of real estate investment trusts and private real estate companies. Mr. Braun graduated from Hardin Simmons University with a Bachelor of Business Administration degree in accounting and finance and subsequently earned the CPA designation and his Series 63, 7, 24 and 27 securities licenses. He is a member of the National Association of Real Estate Investment Trusts and the Texas Society of Certified Public Accountants.

Brett P. Treadwell serves as our General Partner's Vice President — Finance and he also serves in this position for AmREIT. Within AmREIT he is responsible for its financial reporting function as well as for assisting in the establishment and execution of AmREIT's strategic financial initiatives. Mr. Treadwell's responsibilities also include overall risk management and treasury management functions and SEC reporting as well as periodic internal reporting to management. Mr. Treadwell has over 17 years of accounting, financial and SEC reporting experience. Prior to joining AmREIT in August 2004, he served as a senior manager with PricewaterhouseCooper LLP and was previously with Arthur Andersen LLP. Mr. Treadwell received a Bachelor of Business Administration degree from Baylor University and is a Certified Public Accountant.

ITEM 6. EXECUTIVE COMPENSATION

117	1.1	C 1	D 4	1	1	1' '	4.	cc	1	4	1			, •
We are manage	ed hy our	(teneral	Partner	and v	we have	no directors	executive	officers of	r emnlo	vees to	whom w	na na	v com	nensation
TT C all lilalias	ca by bui	General	i ai ai ai cii ci ,	und	vv C mav C	no un coton	, checutive	OTTICCIS OF	cinpic	y CCS tO	***110111 ***	o pu	y COIII	pensanon

See "Item 7. Certain Relationships and Related Transactions" below for the fees and expenses we pay to our General Partner and its affiliates.

ITEM 7. CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS

Compensation Arrangements

Our General Partner and its affiliates received the payments and fees from us described below. These payments and fees were not negotiated and may be higher than payments and fees that would have resulted from an arm's length transaction with an unrelated entity. This does not include payments made by the joints venture in which we have an investment.

		From Inception (October 10, 2006) through
Type and Recipient	Determination of Amount	December 31, 2007
Dealer Manager Fees - AmREIT Securities Company	Organization and Offering Stage Up to 3.25% of the price of Units placed to AmREIT Securities Company, of which up to 0.75% may be reallowed to registered broker-dealers that are members of the Financial Industry Regulatory Authority, Inc. and selected by the Offering's dealer manager to participate in the Offering.	\$4,744,162
Organization and Offering Expenses - General Partner	Reimbursement of our organization and offering expenses, including legal and accounting fees, printing costs, filing fees and distribution costs. However, this amount is limited to 1.0% of the gross offering proceeds from the Offering.	\$609,672
	Operating Stage	
Asset Management Fee - General Partner	A fee equal to 1.0% of net invested capital under management for accounting related services, investor relations, facilitating the deployment of capital, and other services provided by our General Partner to us.	\$204,079
Acquisition Fees - General Partner	A fee not to exceed, for any single acquisition, 3% of the contract purchase price up to \$20 million, and 2% of the contract purchase price in excess of \$20 million.	No amounts have been paid
Development and Redevelopment Fees - General Partner	Development and redevelopment fees on properties we acquire an interest in and for which we intend to develop, redevelop or substantially renovate. These fees will be based on the total project costs, including the cost of acquiring the property, and will be paid as project costs are incurred. These fees shall not exceed, for any single property, 6% of the project cost up to \$10 million, 5% of the project cost between \$10 million and \$20 million, and 4% of the project cost in excess of \$20 million. We will not pay both acquisition fees and development or redevelopment fees on the same property.	No amounts have been paid
Property Management and Leasing Fees - Affiliates of the General Partner (1)	Property management fees not to exceed 4% of the gross revenues (including, without limitation, base rent, percentage rent and expense reimbursement) received from multi-tenant or multi-pad properties, for providing management, operating, maintenance and other services required to maintain a property. Leasing fees not to exceed 4% of base rent on a lease renewal and not to exceed 6% of base rent on an initial lease; provided, however, on leases of over 10,000 square feet, leasing commissions may be paid on the basis of square footage, not to exceed \$6.00 per square foot.	No amounts have been paid
Reimbursement of Operating Expenses - General Partner	We reimburse the actual expenses incurred by our General Partner for performing acquisition, development, management	\$46,341

and administrative functions for us, including construction and construction management fees for development and redevelopment projects.

Partner

Distributions During Operating Stage - General Our General Partner is entitled to distributions during our operating period. See "Item 11. Description of Registrant's Securities To Be Registered – Distributions and Allocations -Distributions to Partners."

\$14,000

Partner (2)

Real Estate Brokerage Commissions - General A fee not to exceed 6% of the sales price on co-brokered transactions and not to exceed 4% of the sales price on individually brokered transactions. Additionally, our General Partner and its affiliates will not be paid real estate brokerage commissions on the sale of a property if the property being sold has not generated an annual return of at least 8.5% per annum on the equity contributed to such property.

No amounts have been paid

Liquidating Stage

Distributions of Net Cash Flow - General Partner

Our General Partner is entitled to distributions after our operating period. See "Item 11. Description of Registrant's Securities to be Registered – Distributions and Allocations – Distributions to Partners."

No amounts have been paid

- The joint ventures paid property management fees of \$199,940 to AmREIT Realty Investment Corporation.
- The real estate brokerage commissions payable during the liquidation period will also be payable for Actively Managed properties sold (2) during the operation period.

Joint Ventures with Affiliates

As of December 31, 2007, we had entered into three joint ventures with our affiliates. In December 2006, we acquired a 50% interest in AmREIT Casa Linda, LP, which owns Casa Linda Plaza, the multi-tenant retail property in Dallas, Texas. The remaining 50% is owned by MIG III, our affiliate.

In August 2007, we acquired a 40% interest in AmREIT Woodlake, LP, which owns Woodlake Square, the multi-tenant retail property in Houston, Texas. Our affiliates, MIG III and ARIC, both own a 30% interest in AmREIT Woodlake, LP.

In November 2007, we acquired a 40% interest in AmREIT Westheimer Gessner, LP, which owns Westheimer Gessner, the multitenant retail property in Houston, Texas. Our affiliates, MIG III and ARIC, both own a 30% interest in AmREIT Woodlake, LP.

Directors

We have no directors and are managed by our General Partner. H. Kerr Taylor is the sole director of our General Partner.

AmREIT Construction Company

We have engaged and may continue to engage AmREIT Construction Company, an affiliate of our General Partner, to provide construction and construction management services for our development and redevelopment projects. Where AmREIT Construction Company is selected to provide construction and construction management services, such services will only be provided on terms and conditions no less favorable to us than can be obtained from independent third parties for comparable services in the same location.

AmREIT Securities Company

AmREIT Securities Company is a member firm of Financial Industry Regulatory Authority, Inc., or FINRA, and a wholly owned subsidiary of AmREIT. AmREIT Securities Company was organized in 1999 for the purpose of participating in and facilitating the distribution of securities of AmREIT affiliated entities. As the dealer manager for our Offering of Units, AmREIT Securities Company provided certain sales, promotional and marketing services to us in connection with the distribution of the Units. We paid selling commission and marketing reimbursements to AmREIT Securities Company in connection with our offering, a portion of which was reallowed to participating broker-dealers. Chad Braun is the President, Secretary and Treasurer of AmREIT Securities Company and H. Kerr Taylor serves as its Chairman and the sole member of its board of directors.

ITEM 8. LEGAL PROCEEDINGS

In the ordinary course of business, we may become subject to litigation or claims. Neither we nor our properties are the subject of any non-routine pending legal proceeding, nor are we aware of any proceeding that a governmental authority is contemplating against us.

ITEM 9. MARKET PRICE OF AND DIVIDENDS ON THE REGISTRANT'S COMMON EQUITY AND OTHER STOCKHOLDER MATTERS

Market Information and Holders

On November 15, 2006, we commenced the Offering. Our Offering was closed as of April 1, 2008 and, as of that date we had raised aggregate gross proceeds of approximately \$50.0 million through the sale of 2,000 Units to 822 Limited Partners.

There is no established trading market for our Units. As of December 31, 2007 none of the Units were subject to any outstanding options or warrants, and we had not issued any securities convertible into our Units. The Units, which are "restricted securities" as defined in Rule 144 promulgated by the SEC under the Securities Act, must be held indefinitely unless they are subsequently registered under the Securities Act and any applicable state securities laws or unless, upon the advice of counsel satisfactory to us, the Units are sold in a transaction that is exempt from the registration requirements of such laws. As of December 31, 2007, no Units were eligible for sale under Rule 144 or that we have agreed to register under the Securities Act for sale by Limited Partners and there were no Units that are being, or have been publicly proposed to be, publicly offered by us.

Distributions

The following table shows the distributions to our Limited Partners our General Partner has declared (including the total amount paid and the amount paid on a per Unit basis) since we commenced operations on December 8, 2006 and through the year ended December 31, 2007:

	1 otal Amount of Distributions Distribution	n
Period Declared	Paid Per Unit	
December 2006	\$ \$	
January 2007	1,878 156.25	
February 2007	14,069 156.25	
March 2007	27,946 156.25	
April 2007	46,375 156.25	
May 2007	79,854 156.25	
June 2007	108,679 156.25	
July 2007	136,461 156.25	
August 2007	153,375 156.25	
September 2007	170,430 156.25	
October 2007	192,316 156.25	
November 2007	210,919 156.25	
December 2007	247,119 156.25	
Total	\$1,389,421	

We have made monthly distributions to our Limited Partners at a rate of 7.5% per annum from January 1, 2007 through December 31, 2007. All of the distributions we have paid through December 31, 2007 constitute a return of capital to our Limited Partners. To date, our distributions have exceeded our cash flow from operations by approximately \$329,000. One of our primary goals is to pay regular (monthly) distributions to our Limited Partners. We expect to maintain the distribution rate of 7.5%, unless our results of operations, our general financial condition, general economic conditions or other factors prohibit us from doing so. The funds we receive from operations that are available for distribution may be affected by a number of factors, including the following:

Total

- our operating and interest expenses;
- the ability of tenants to meet their obligations under the leases associated with our properties;
- our ability to keep the properties in which we have investments occupied;
- our ability to maintain or increase rental rates when renewing or replacing current leases;
- capital expenditures and reserves for such expenditures;
- the issuance of additional shares; and
- financings and refinancings.

We qualified as a partnership for federal income tax purposes commencing with our taxable year ended December 31, 2006. For income tax purposes, distributions to our Limited Partners are characterized as ordinary income, capital gains, or as a return of a Limited Partner's invested capital.

Securities Authorized for Issuance Under Equity Compensation Plans

We do not have any compensation plans under which we are authorized to issue equity securities.

ITEM 10. RECENT SALES OF UNREGISTERED SECURITIES

AmREIT was our initial Limited Partner in connection with our organization on October 10, 2006 and received its initial partnership interest. AmREIT's interest was redeemed when we reached the initial minimum offering amount and accepted subscription from investors on January 12, 2007. As of December 31, 2007, we issued 1,796 Units to 748 investors in the Offering, which began on November 15, 2006.

The offering price for the aggregate Units sold in the Offering was \$25,000, and as of December 31, 2007, the aggregate fees paid to our General Partner and its affiliates in connection with the Offering were \$4,744,162. The net proceeds from the sale of the Units we received as of December 31, 2007 were \$44,812,422. We relied on the exemption from registration provided by Rule 506 under Regulation D and Section 4(2) of the Securities Act in connection with of the Offering. In each instance, the purchaser had access to sufficient information regarding us so as to make an informed investment decision. We had reasonable basis to believe that each purchaser was an accredited investor, as defined in Regulation D under the Securities Act, and was acquiring the Units for investment only and not with a view to distribute, sell or otherwise transfer the Units. The Units were distributed by AmREIT Securities Company, which acted as the dealer manager of our Offering, and did not involve a public offering or general solicitation.

ITEM 11. DESCRIPTION OF REGISTRANT'S SECURITIES TO BE REGISTERED

General

The Units represent limited partnership interests in us and entitle their holders to certain allocations and distributions. The parties who purchased Units from us became limited partners in the Partnership. Units may not be freely assigned and are subject to restrictions on transfer by law and by the Partnership Agreement. There is no public trading market for the Units and it is not anticipated that a public trading market for the Units will develop.

We were formed under the Delaware Revised Uniform Limited Partnership Act. Our General Partner is AmREIT Monthly Income & Growth IV Corporation, a Texas corporation and subsidiary of AmREIT. Our four properties are located in Texas. We are qualified to transact business in Texas. The character and general nature of the business to be conducted by us is the ownership, operation and eventual sale of our properties.

Distributions and Allocations

Distributions to Partners

We have made monthly distributions to our Limited Partners at a rate of 7.5% of invested capital per annum from January 1, 2007 to December 31, 2007. All of these distributions paid to date constitute a return of capital to our Limited Partners. We intend to pay regular (monthly) distributions to our Limited Partners at the distribution rate of 7.5%, unless our results of operations, our general financial condition, general economic conditions or other factors prohibit us from doing so. Distributions are made to our Partners of record as of the close of each calendar month immediately preceding the month in which the distribution is made. The amount of any distributions is pro rated in accordance with the number of days the Partner was a Partner during such month.

The actual amount and timing of distributions is determined by our General Partner in its discretion and typically will depend on the amount of funds available for distribution, our financial condition, whether available funds should be reinvested, capital expenditure and reserve requirements and general operational requirements. During the early stages of our operations, we have not had sufficient cash available from operations to pay distributions. Therefore, we have used proceeds from the Offering to make cash distributions, which reduce the amount of proceeds available for investments and operations. There can be no assurance as to when and in what amounts future distributions will be made

Cash from whatever source will be distributed during our operating period as follows: 1% to our General Partner and 99% to our Limited Partners.

Cash from whatever source will be distributed after our operating period as follows:

- First: 100% to our Limited Partners (in proportion to their unreturned actual invested capital) until such time as our Limited Partners have received cumulative distributions from all sources equal to 100% of their actual invested capital (calculated using the actual purchase price per Unit);
- Second: 100% to our General Partner until it has received cumulative distributions from all sources equal to 100% of its actual invested capital of \$1,000;

- Third: 1% to our General Partner and 99% to our Limited Partners on a per Unit basis until such time as our Limited Partners have received cumulative distributions from all sources equal to an 8.5% per annum, cumulative, uncompounded return on their unreturned deemed capital contributions (which will be equal to (i) the product of \$25,000 per Unit (regardless of the purchase price paid for a Unit) multiplied by the number of Units owned by a Partner, reduced by (ii) the aggregate amount of any distributions received that constitute a return of capital contributions);
- Fourth: 100% to our General Partner until it has received cumulative distributions from all sources (other than with respect to the Units it purchased) in an amount equal to 40% of the net cash flow paid to date to our Limited Partners in excess of their actual invested capital; and
- Thereafter: 60% to our Limited Partners on a per Unit basis and 40% to our General Partner.

All or a portion of each cash distribution may consist of a distribution of a Partner's original capital investment which is deemed to be that portion of cash distributions which is not attributable to Partnership net income for the period of the distribution, or to prior periods, irrespective of whether such distributions have in fact been paid from cash from current or past operations.

Allocations of Profits and Losses

Under the Partnership Agreement, except as otherwise required by the special allocation provisions of the Delaware Revised Uniform Limited Partnership Act, profits and losses shall be allocated in the following manner:

- Profits shall be allocated:
- First: 99% to our Limited Partners on a per Unit basis and 1% to our General Partner, until our Limited Partners receive their actual invested capital plus an 8.5% annual return;
- Second: 100% to our General Partner until its capital account balance equals 40% of our Partners' aggregate capital account balance, taking into account any proposed distributions and allocations; and
- o Thereafter: 60% to our Limited Partners on a per Unit basis and 40% to our General Partner.
- Losses shall be allocated:
- First: 60% to our Limited Partners on a per Unit basis and 40% to our General Partner until prior 60%-40% allocations of profits have been reversed;
- o Second: 100% to our General Partner until prior 100% allocations of profits have been reversed; and
- o Thereafter: 99% to our Limited Partners on a per Unit basis and 1% to our General Partner.

Under the Partnership Agreement, profits and losses are essentially allocated in a manner such that the adjusted capital account balance of each Partner prior to making liquidating distributions equals the amount each such Partner is entitled to receive in liquidation of its interest in the Partnership so that the liquidating distributions to each such Partner will result in each Partner's adjusted capital account balance being reduced to zero upon the termination of the Partnership.

Adjustments to Allocations

Our General Partner may alter the foregoing allocations of profits and losses, including any item thereof in any one year where our tax advisor advises that such allocation will likely not be in compliance with the Delaware Revised Uniform Limited Partnership Act or the regulations promulgated thereunder. In such event, our General Partner shall allocate any such item of profits or losses so that, in its sole judgment, it most closely achieves the original allocation.

Table of Contents

Partnership Agreement

The rights of our Limited Partners are governed by Delaware law as well as our Partnership Agreement. The following is a summary of the material provisions of the Partnership Agreement and is qualified in its entirety by the full text thereof, which is attached hereto as Exhibits 3.2 and 3.3.

Management

Our General Partner has the exclusive right to manage and control our affairs, except those limited matters upon which our Limited Partners must first consent or vote to approve. See "– Limited Partners' Voting Rights" below. Our General Partner may change our principal place of business from time to time but must provide written notice of any change to our Limited Partners.

Term

By November 15, 2013, our General Partner will market our properties for sale and begin an orderly liquidation of our assets. Our operating period may be extended to November 15, 2015 only with the consent of the majority of Units held by our Limited Partners. As a legal entity, however, unless sooner terminated by our General Partner or by our bankruptcy, insolvency, liquidation or dissolution, we will terminate on December 31, 2026.

Capital Contributions

Our Partners have made initial capital contributions as follows:

General Partner: \$1,000

Limited Partners: \$25,000 per Unit

AmREIT has also invested \$800,000 in us through the acquisition of Units. AmREIT's \$800,000 investment was made at a purchase price of \$22,250 per Unit, reflecting that no sales commissions or selling expenses were payable with respect to the investment. Our Partners are not subject to additional capital calls. The sharing of partnership distributions among our Partners and the allocations of partnership income and loss to our Partners is described above under "– Distributions and Allocations."

Compensation to our General Partner and Its Affiliates

Our General Partner and/or its affiliates will be compensated as provided in the Partnership Agreement and as described herein. See "Item 7. Certain Relationships and Related Transactions – Compensation Arrangements."

Partnership Expenses

Our General Partner will endeavor to have all our expenses billed directly to us. To the extent our General Partner incurs expenses on our behalf, our General Partner shall be reimbursed for its costs thereof. See "Item 7. Certain Relationships and Related Transactions – Compensation Arrangements."

Limited Partners' Voting Rights

Limited Partners may not take part in our management. However, our General Partner may not do any of the following without first obtaining a vote or consent of the Limited Partners owning a majority of the outstanding Units:

- amend the Partnership Agreement (except to admit a substituted Limited Partner, to perform certain other ministerial acts or as otherwise expressly permitted in the Partnership Agreement);
- (2) confess a judgment against us in an amount exceeding \$100,000;
- (3) possess any property or assign our rights in a specific property, for other than a Partnership purpose;
- (4) resign or withdraw as a General Partner;
- (5) sell its General Partner interest;

- other than indebtedness incurred in connection with the Casa Linda Plaza property, incur indebtedness on our behalf in an amount greater than 75% of our cash and cash equivalents plus the aggregate market value of our properties;
- (7) extend our operating period from November 15, 2013 to November 15, 2015;
- (8) permit us to invest more than 15% of the gross proceeds from the Offering in properties outside of the United States;
- (9) permit us to invest more than 20% of the gross proceeds from the Offering in mortgages; or
- (10) permit us to invest in the securities of other issues, except for permitted temporary investments pending utilization of our funds for investments in real properties.

For the purposes of the Partnership Agreement, our Limited Partners may only vote or give their written consent if they own Units of record as of the applicable record date, which shall not be more than 60 days preceding the date on which the meeting where the vote is to be taken or the last date on which written consents may be submitted for the matter being considered. Additionally, for purposes of obtaining vote of a majority of the outstanding Units, our General Partner's investment in Units are considered non-voting units. See "— Mergers, Consolidations and Conversions" below for additional voting rights of Limited Partners.

Indemnification of General Partner

Our Limited Partners, within the limits of their respective capital contributions and to the extent of their allocable portion of Partnership assets, agree to indemnify and hold our General Partner harmless from and against actual and reasonable third-party claims or lawsuits arising out of our activities and operations, except those instances in which our General Partner failed to adhere to its fiduciary obligations to us, or acted with gross negligence or willful or wanton misconduct. Amounts paid to indemnify our General Partner may be recouped to the extent such payments relate to a third-party claim or lawsuit arising out of a breach by our General Partner of its fiduciary obligations to us.

Restrictions on Transfer of Units

Units may be transferred of record only if properly executed assignment documents are delivered to us in the form approved by our General Partner and such transfer would not result in our being treated as an association taxable as a corporation. An assignee of a Unit may become a substituted Limited Partner only with the written consent of our General Partner, whose consent may be withheld for any or no reason, and after delivering the properly executed documents to our General Partner in the form prescribed by our General Partner. Transferees are also required to obtain a legal opinion acceptable to our General Partner that the transfer is exempt from registration under the Securities Act and any applicable state securities laws. Our General Partner may waive the requirement of a legal opinion in its sole discretion. A fee of \$150 or the actual cost to us to effect the transfer, whichever is greater, must be submitted with the foregoing. Transfers of Units will be recognized by us only after all of the transfer conditions have been satisfied.

Mergers, Consolidations and Conversions

The Partnership Agreement provides that, in order to adopt a plan of merger pursuant to which we are merged into any other entity, a plan of exchange pursuant to which all of the Units are exchanged for cash or securities of another entity, or a plan of conversion whereby we are converted into another entity, the transaction generally must be approved by Limited Partners owning at least 50% of the outstanding Units (or a higher percentage if required by our General Partner). Limited Partner approval is not required for any of the following transactions: (1) in the case of a plan of exchange, if we are the acquiring entity; (2) if the surviving entity in a merger or conversion is a limited partnership that has a substantially identical partnership agreement and Limited Partners holding Units immediately before the date of merger or conversion will hold the same interest in the same proportions immediately after the effective date of the merger or conversion; or (3) if the transaction involves our conversion to corporate, trust or association form and, as a consequence of the transaction, there will be no significant adverse change in investors' voting rights, the term of existence of the surviving or resulting entity, sponsor compensation or investment objectives. In the event that federal tax legislation is enacted that results in another form of organization having superior tax advantages to our limited partnership structure, our General Partner has the ability to cause us to convert to a corporation or other organizational form without Limited Partner approval, if our General Partner determines that such conversion is in our best interest.

Limited Redemption Program

Subject to and upon compliance with the provisions of the Partnership Agreement, at any time after November 15, 2009 and prior to November 15, 2013, any holder of Units who has held the Units for not less than three years may present all of those Units to us for redemption at any time. We may, at our sole option, redeem those Units presented for redemption for cash to the extent we have sufficient funds available thereof; provided that the sum of the percentage interests in Partnership capital or profits transferred during our taxable year does not exceed 9% of the total interests in partnership capital or profits as determined in the sole discretion of our General Partner. The redemption price paid to the holder of Units will be 92% of the unreturned invested capital. At no time during a 12-month period, however, may the number of Units we redeem exceed 2% of the number of Units outstanding at the beginning of that 12-month period unless such redemption is otherwise deemed to be a disregarded transfer for purposes of determining whether we are a "publicly traded partnership."

Amendments to the Partnership Agreement

The Partnership Agreement may be amended only by a vote of the majority of Units held by our Limited Partners. Our General Partner may amend the Partnership Agreement without action by our Limited Partners so long as such amendments do not directly affect the rights of our Limited Partners. Permissible changes include amendments to reflect (1) a change in our name or location of our principal place of business, (2) the disposition of any Limited Partner's interest in any one or more Units, (3) the admission of additional or Substitute Limited Partners, or (4) a clarification of the provisions of the Partnership Agreement, provided such clarification does not change the intent of the Partnership Agreement. Also, our General Partner is authorized to make certain amendments as may be necessary to assure compliance with applicable federal income tax law by our income and loss allocation provisions.

Special Power of Attorney

Each Limited Partner has appointed our General Partner (and any successor or substitute General Partner) as his, her or its true and lawful attorney-in-fact who may act in each Limited Partner's stead to execute, certify, acknowledge, swear to, file and record the Partnership Agreement and any and all amendments to the Partnership Agreement, which are adopted as provided in the Partnership Agreement. In addition, our General Partner has been designated as the Limited Partner's agent to execute, acknowledge and deliver all conveyance and other instruments that our General Partner deems appropriate, in accordance with the Partnership Agreement, to effect the transfer of Units, including assignments on the default of the Limited Partner, to admit, substitute or delete Partners, to sell, exchange or dispose of our assets or properties, to borrow money and otherwise enter into financing transactions and to execute all amendments and/or restatements of the Partnership Agreement. The purpose of this power of attorney is to relieve our General Partner of the requirement of obtaining the required number of signatures of Limited Partners to effect the filing of certificates, agreements, or amendments to agreements as may be required to effect our operation and business as described herein. This power of attorney is coupled with an interest (our Limited Partners' purchase of our Units) and is irrevocable.

Removal of a General Partner

Our General Partner may be removed, with or without cause by an affirmative vote or written consent of the Limited Partners owning at least a majority of the Units then outstanding (not counting any Units then owned by our General Partner). For the purposes of the foregoing, "cause" shall mean (1) the bankruptcy of our General Partner pursuant to an event of bankruptcy, (2) the incapacity of our General Partner, (3) the commission by our General Partner of an intentional material breach of this Agreement, or (4) conduct of our General Partner constituting gross negligence or willful or wanton misconduct harmful to us.

In the event that our Limited Partners vote to remove our General Partner, our General Partner shall have an additional 30 business days from the date of the notice of removal from our Limited Partners in which to cure or otherwise resolve the matter(s) resulting in the notice to remove to the reasonable satisfaction of our Limited Partners. Our Limited Partners shall not unreasonably withhold approval of the cure accomplished by our General Partner. In the event that our General Partner is removed, it hereby agrees to convert its General Partner's Capital Contribution into that of a Limited Partner having no voting rights but whose Partnership interest is unchanged in all other respects. The removed General Partner shall be removed from and indemnified against our liabilities including, without limitation, liabilities such General Partner has guaranteed. Further, if removed except for cause, we must purchase our General Partner's Interest and/or any successor General Partner's Partnership Interest for fair market value, which will be determined by the median of the independent appraisals of the General Partner Interest.

In the event of the dissolution and liquidation or removal of our General Partner and our continued existence in accordance with Section 8.3 of the Partnership Agreement, the successor of the said General Partner shall not be entitled to any distributions due the former General Partner under the Partnership Agreement. Our Limited Partners have agreed in advance to such a transfer and specifically authorize that the power of attorney provided for in their Subscription Agreements with us be utilized, if need be, on their behalf to effectuate fully the legal conversion or transfer, as the case may be, of the dissolved or removed General Partner's Interest hereunder.

Meetings

Our General Partner may call meetings of our Limited Partners. Meetings may also be held upon the written demand of Limited Partners holding more than 25% of the then outstanding Units.

Books and Records

Our General Partner will maintain at our principal place of business adequate books and records setting forth an account of all our business transactions and a list of all our Limited Partners. Limited Partners holding no less than 5% of the outstanding Units or their designated representatives have the right during normal business hours to inspect such books and records. Limited Partners holding less than 5% of the outstanding Units may inspect our books and records only with the consent of our General Partner, in its sole discretion.

Limited Liability

No Limited Partner will be liable for our debts or other obligations, except to the extent of such Limited Partner's share of undistributed profits, if any, and the amount of any distributions we have made to such Limited Partner, which constitute a return of such Limited Partner's capital contribution, unless such Partner takes part in the control of our business, which is not permitted under the Partnership Agreement. The Partnership Agreement requires our General Partner to cause us to operate in such manner as it deems appropriate to avoid unlimited liability for our Limited Partners.

Additional Units

Our General Partner may issue additional Units to existing Partners or any other persons for such consideration and on such terms and conditions as our General Partner may determine in its sole discretion, all without the approval of our Limited Partners. Any such additional Units may be issued in one or more classes, or in one or more series of any of such classes, with designations, preferences and relative participating, option or other special rights, powers and duties, including rights, powers and duties senior to Units previously issued to our Limited Partners, all as shall be determined by our General Partner in its sole discretion and without the approval of any Limited Partner.

ITEM 12. INDEMNIFICATION OF DIRECTORS AND OFFICERS

Under the Partnership Agreement, our Limited Partners, within the limits of their respective capital contributions and to the extent of their Partnership assets, agree to indemnify and hold our General Partner (and its affiliates that furnish services to us within the scope of our General Partner's authority) harmless from and against actual and reasonable third-party claims or lawsuits arising out of our activities and operations, except those instances in which our General Partner failed to adhere to its fiduciary obligations to us, or acted with gross negligence or willful or wanton misconduct. Amounts paid to indemnify our General Partner may be recouped to the extent such payments relate to a third-party claim or lawsuit arising out of a breach by our General Partner of its fiduciary obligations to us. It is the opinion of the SEC that indemnification for liabilities arising under the Securities Act is contrary to public policy, and therefore, unenforceable. In the case of liability arising from an alleged violation of securities laws, our General Partner may obtain indemnification if: (1) our General Partner is successful in defending the action; or (2) the court or our Limited Partners owning at least a majority of the Units then outstanding exclusive of any Units then owned by our General Partner specifically approve the indemnification.

ITEM 13. FINANCIAL STATEMENTS AND SUPPLEMENTARY DATA

See the Index to Financial Statements beginning on page F-1.

The following SEC filing is incorporated by reference: Quarterly Report Form 10-Q for the quarter ended June 30, 2008 and filed on August 14, 2008.

ITEM 14. CHANGES IN AND DISAGREEMENTS WITH ACCOUNTANTS ON ACCOUNTING AND FINANCIAL DISCLOSURE

There were no changes in or disagreements with our independent registered public accountants during the year ended December 31, 2007, or the period from October 10, 2006 (inception) through December 31, 2006.

ITEM 15. FINANCIAL STATEMENTS AND EXHIBTIS

(a) List separately all financial statements filed.

See Index to Financial Statements beginning on page F-1.

(b) Exhibits

See "Exhibit Index" below.

PART F/S

AMREIT MONTHLY INCOME & GROWTH FUND IV, L.P.

INDEX TO FINANCIAL STATEMENTS

FINANCIAL STATEMENTS:	Page
Report of Independent Registered Public Accounting Firm	F-1
Balance Sheet as of December 31, 2007 and 2006	F-2
Statements of Operations for the year ended December 31, 2007 and for the period from October 10, 2006 (inception) through	F-3
<u>December 31, 2006</u>	F-4
Statements of Partners' Capital for the year ended December 31, 2007 and for the period from October 10, 2006 (inception)	
through December 31, 2006	
Statements of Cash Flows for the year ended December 31, 2007 and for the period from October 10, 2006 (inception) through	F-5
<u>December 31, 2006</u>	
Notes to Financial Statements - December 31, 2007 and 2006	F-6
Consolidated Balance Sheet as of March 31, 2008 (unaudited) and December 31, 2007	F-9
Consolidated Statements of Operations for the three months ended March 31, 2008 and 2007 (unaudited)	F-10
Consolidated Statements of Partners' Capital for the three months ended March 31, 2008 (unaudited)	F-11
Consolidated Statements of Cash Flows for the three months ended March 31, 2008 and 2007 (unaudited)	F-12
Notes to Consolidated Financial Statements - March 31, 2008	F-13
FINANCIAL STATEMENT SCHEDULES:	
Report of Independent Registered Public Accounting Firm	S-1
Schedule III – Financial Statements and Notes for AmREIT Casa Linda, L.P.	S-2
Report of Independent Registered Public Accounting Firm	S-9
Schedule IV – Financial Statements and Notes for AmREIT Woodlake Square, L.P.	S-10
Report of Independent Registered Public Accounting Firm	S-17
Schedule V - Balance Sheet and Notes for AmREIT Monthly Income & Growth IV Corporation as of December 31, 2007	S-18

Report of Independent Registered Public Accounting Firm

The Partners

AmREIT Monthly Income & Growth Fund IV, L.P.:

We have audited the accompanying balance sheets of AmREIT Monthly Income & Growth Fund IV, L.P. (the "Partnership") as of December 31, 2007 and 2006, and the related statements of operations, partners' capital and cash flows for the year ended December 31, 2007 and for the period from October 10, 2006 (inception) through December 31, 2006. These financial statements are the responsibility of the Partnership's management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of AmREIT Monthly Income & Growth Fund IV, L.P. as of December 31, 2007 and 2006, and the results of their operations and their cash flows for the year ended December 31, 2007 and for the period from October 10, 2006 (inception) through December 31, 2006, in conformity with U.S. generally accepted accounting principles.

(signed) KPMG LLP

Houston, Texas

April 29, 2008

PART I - FINANCIAL INFORMATION

Item 1. Financial Statements

Amreit Monthly Income & Growth Fund IV, L. P. BALANCE SHEET as of December 31, 2007 and 2006

	December 31, 2007 (in	December 31 2006
ASSETS	thousands)	(in thousands)
Investment in non-consolidated entities	\$17,100	\$ 5,216
Cash and cash equivalents	19,349	796
Escrow Deposits	1,540	-
Accounts receivable	113	-
Other assets	10	<u> </u>
TOTAL ASSETS	\$38,112	\$ 6,012
LIABILITIES AND PARTNERS' CAPITAL		
Liabilities:		4.026
Notes payable - related party Accounts payable	104	4,926 4
Accounts payable - related party	1,208	763
TOTAL LIABILITIES	\$1,312	\$ 5,693
Partners' capital:		
General partner	_	_
Limited partners, 1,796 and 32 units outstanding at December 31,		
2007 and 2006, respectively	36,800	319
TOTAL PARTNERS' CAPITAL	36,800	319
TOTAL LIABILITIES AND PARTNERS' CAPITAL	\$38,112	\$ 6,012

See Notes to Financial Statements

Amreit Monthly income & growth fund IV, L. P. Statements of operations

For the Year Ended December 31, 2007 and for the Period from October 10, 2006 (inception) through December 31, 2006

	2007	2006
	(in thousands)	
Revenues	\$-	\$-
Expenses:		
General and administrative	156	_
Asset management fees - related party	204	-
Legal and professional	119	17
Total expenses	479	17
Operating loss	(479)	(17)
Other income (expense):		
Interest and other income	523	-
Interest expense	(101)	(26)
Income tax expense	(4)	-
(Loss) from non-consolidated entities	(1,123)	(28)
Total other income (expense)	(705)	(54)
Net loss	<u>\$(1,184</u>)	\$(71)

See Notes to Financial Statements

Amreit monthly income & growth fund IV, L.P. Statements of Partners' capital

For the Year Ended December 31, 2007 and for the Period from October 10, 2006 (inception) through December 31, 2006 (in thousands)

	GP	LP	Total
Balance at October 10, 2006 (inception)	\$-	\$-	\$-
Contributions, net of issuance costs of \$0.4 million	1	389	390
Net loss	-	(71) (71)
Distributions	-	-	-
		<u> </u>	_
Balance at December 31, 2006	\$1	\$318	\$319
		_	
Contributions, net of issuance costs of \$4.9 million	-	39,069	39,069
Net loss (1)	13	(1,197) (1,184)
Distributions	(14) (1,390) (1,404)
			_
Balance at December 31, 2007	\$-	\$36,800	\$36,800

(1) The allocation of net loss during 2007 includes a curative allocation to increase the GP capital account by \$25,000. The partnership agreement provides that no partner shall be required to fund a deficit balance in their capital account.

See Notes to Financial Statements.

Amreit Monthly income & growth fund IV, L. P. Statements of Cash flows

For the Year Ended December 31, 2007 and for the Period from October 10, 2006 (inception) through December 31, 2006

	2007 (in thousands)	2006
Cash flows from operating activities:		
Net loss	\$(1,184)	\$(71)
Adjustments to reconcile net loss to net cash provided by		
(used in) operating activities:		
Loss (income) from non-consolidated entities	1,123	28
Increase in accounts receivable	(113	-
Increase in other assets	(10	-
Increase in accounts payable and other liabilities	100	93
Increase in accounts payable - related party	444	664
Net cash provided by operating activities	360	714
Cash flows from investing activities:	,, <u>,</u> , , , , , , , , , , , , , , , , ,	
Investment in non-consolidated entities	(13,007)	(5,244)
Earnest money deposit made	(1,540	·
Net cash used in investing activities	(14,547	(5,244)
Cash flows from financing activities:		
Proceeds from notes payable - related party	_	4,926
Payments of notes payable - related party	(4,926	1,520
Contributions	44,011	801
Issuance costs	(4,942	(401)
Distributions	(1,403	-
Net cash provided by financing activities	32,740	5,326
1 7 8		
Net increase in cash and cash equivalents	18,553	796
Cash and cash equivalents, beginning of period	796	
Cash and cash equivalents, end of period	\$19,349	\$796
Supplemental schedule of cash flow information:		
Cash paid during the year for interest	\$101	\$26

See Notes to Financial Statements.

AMREIT MONTHLY INCOME & GROWTH FUND IV, L.P.

NOTES TO FINANCIAL STATEMENTS

December 31, 2007 and 2006

1. DESCRIPTION OF BUSINESS AND NATURE OF OPERATIONS

AmREIT Monthly Income & Growth Fund IV, Ltd., a Delaware limited partnership (the "Partnership" or "MIG IV"), was formed on October 10, 2006 to acquire, develop and operate directly or indirectly, a portfolio of commercial real estate consisting primarily of multitenant shopping centers and mixed-use developments. The General Partner of the Partnership is AmREIT Monthly Income & Growth IV Corporation, a Delaware Corporation (the "General Partner"), which is a subsidiary of AmREIT, a Texas real estate investment trust. The General Partner maintains its principal place of business in Houston, Texas.

We commenced our principal operations on December 8, 2006 when we acquired our first interest in a property. On January 12, 2007 we raised the minimum offering of \$1.0 million pursuant to the terms of our Offering Memorandum dated November 15, 2006 (the "Offering Memorandum") and issued the initial 40 Units. As of December 31, 2007, we had received \$44.0 million from the sale of 1,796 Units and we had acquired investment interests in four properties comprising of 612,280 square feet of gross leasable area.

Our limited partnership units were sold pursuant to exemptions from registration under the Securities Act of 1933 and are not currently listed on a national exchange. These units will be transferable only if we register them under such laws, and such registration is not expected. We do not anticipate that any public market for the limited partnership units will develop.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

BASIS OF PRESENTATION

Our financial records are maintained on the accrual basis of accounting whereby revenues are recognized when earned and expenses are recorded when incurred. The consolidated financial statements include our accounts as well as the accounts of any wholly- or majority-owned subsidiaries in which we have a controlling financial interest. Investments in joint ventures and partnerships where we have the ability to exercise significant influence but do not exercise financial and operating control, are accounted for using the equity method. All significant inter-company accounts and transactions have been eliminated in consolidation.

We were formed on October 10, 2006. Accordingly, the accompanying statements of operations, cash flows and changes in partners' capital related to 2006 represent activity for the period from October 10, 2006 (inception) through December 31, 2006. Unless otherwise noted, all references to the "2006 period" or "the period ended December 31, 2006" refer to the short period of operations from October 10, 2006 through December 31, 2006.

INVESTMENT IN NON-CONSOLIDATED ENTITIES

As of December 31, 2007, we have ownership interests in four real estate properties. Although we exercise significant influence over the activities of these properties, we do not have a controlling financial interest in them. Accordingly, all of our real estate investments are reported under the equity method of accounting pursuant to U.S. generally accepted accounting principles. Certain of the significant accounting policies below are applicable specifically to property-level reporting and represent policies that are therefore primarily relevant at the investee entity level as of December 31, 2007.

We own interests in four properties through joint ventures. The joint ventures lease space to tenants under agreements with varying terms. The majority of the leases are accounted for as operating leases with revenue being recognized on a straight-line basis over the terms of the individual leases. When our joint ventures acquire a property, the term of existing leases is considered to commence as of the acquisition date for the purposes of this calculation.

Accrued rents are included in tenant receivables. Revenue from tenant reimbursements of taxes, maintenance expenses and insurance is recognized in the period the related expense is recorded. Additionally, certain of the lease agreements contain provisions that grant additional rents based on tenants' sales volumes (contingent or percentage rent). We defer the recognition of contingent or percentage rental income until the specific targets as defined in lease agreements that trigger the contingent or percentage rental income are achieved. Cost recoveries from tenants are included in rental income in the period the related costs are incurred.

JOINT VENTURES ACCOUNTING POLICIES

The joint ventures in which we invest account for real estate acquisitions pursuant to SFAS No. 141, *Business* Combinations (SFAS No. 141). Accordingly, they allocate the purchase price of the acquired properties to tangible and intangible assets as well as liabilities acquired. The joint ventures compute depreciation using the straight-line method over an estimated life of up to 50 years for buildings, up to 20 years for site improvements and over the term of the lease for tenant improvements. The intangible assets are being amortized over the remaining term of their respective leases.

The joint ventures in which we invest assess their properties for impairment whenever events or changes in circumstances indicate that the carrying amount of the assets, including accrued rental income, may not be recoverable through operations. We determine whether an impairment value occured by comparing the estimate future cash flows (undiscounted and without interest charges), including the residual value of the property, with the carrying value of the asset exceeds its fair value. We do not believe that any of our joint venture investment have carrying values that are in excess of their respective fair values as of December 31, 2007.

ENVIRONMENTAL EXPOSURES

We are subject to numerous environmental laws and regulations as they apply to real estate pertaining to chemicals used by the dry cleaning industry, the existence of asbestos in older shopping centers, and underground petroleum storage tanks. We believe that the tenants who currently operate dry cleaning plants or gas stations do so in accordance with current laws and regulations. We believe that the ultimate disposition of currently known environmental matters will not have a material affect on our financial position, liquidity, or operations. However, we can give no assurance that existing environmental studies with respect to the shopping centers have revealed all potential environmental liabilities; that any previous owner, occupant or tenant did not create any material environmental condition not known to it; that the current environmental condition of the shopping centers will not be affected by tenants and occupants, by the condition of nearby properties, or by unrelated third parties; or that changes in applicable environmental laws and regulations or their interpretation will not result in additional environmental liability to the Partnership.

INCOME TAXES

<u>Federal</u> - No provision for U.S. federal income taxes is included in the accompanying consolidated financial statements. As a partnership, we are not subject to federal income tax, and the federal tax effect of our activities is passed through to our partners.

<u>State</u> - In May 2006, the State of Texas adopted House Bill 3, which modified the state's franchise tax structure, replacing the previous tax based on capital or earned surplus with one based on margin (often referred to as the "Texas Margin Tax" effective with franchise tax reports filed on or after January 1, 2008. The Texas Margin Tax is computed by applying the applicable tax rate (1% for the Partnership) to the profit margin, which, generally, will be determined for us as total revenue less a 30% standard deduction. Although House Bill 3 states that the Texas Margin Tax is not an income tax, the Partnership believes that SFAS No. 109, *Accounting for Income Taxes*, applies to the Texas Margin Tax. The Partnership has determined that the impact of the Texas Margin Tax is insignificant as of December 31, 2007 and December 31, 2006. The Texas Margin Tax accrued as of December 31, 2007 total \$3,661.

USE OF ESTIMATES

The preparation of consolidated financial statements in conformity with U.S. generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

RECENTLY ISSUED ACCOUNTING STANDARDS

In September 2006, the Financial Accounting Standards Board ("FASB") issued SFAS No. 157, *Fair Value Measurement* (SFAS No. 157). SFAS No. 157 defines fair value, establishes a framework for measuring fair value and requires enhanced disclosures about fair value measurements. SFAS No. 157 requires companies to disclose the fair value of its financial instruments according to a fair value hierarchy. Additionally, companies are required to provide certain disclosures regarding instruments within the hierarchy, including a reconciliation of the beginning and ending balances for each major category of assets and liabilities. SFAS No. 157 is effective for our fiscal year beginning January 1, 2008. The adoption of SFAS No. 157 is not expected to have a material effect on our results of operations or financial position.

In February 2007, the FASB issued SFAS No. 159, *The Fair Value Option for Financial Assets and Financial Liabilities* (SFAS No. 159). SFAS No. 159 expands opportunities to use fair value measurement in financial reporting and permits entities to choose to measure many financial instruments and certain other items at fair value. This statement is effective for fiscal years beginning after November 15, 2007. We currently do not plan to measure any eligible financial assets and liabilities at fair value under the provisions of SFAS No. 159.

In December 2007, the FASB issued SFAS No. 141 (Revised 2007), *Business Combinations* (SFAS No. 141R). SFAS No. 141R will change the accounting for business combinations. Under SFAS No. 141R, an acquiring entity will be required to recognize all the assets acquired and liabilities assumed in a transaction at the acquisition-date fair value with limited exceptions. SFAS No. 141R applies prospectively to business combinations for which the acquisition date is on or after the beginning of the first annual reporting period beginning on or after December 15, 2008. We are currently evaluating the potential impact of SFAS No. 141R on our financial position and results of operations beginning for fiscal year 2009.

In December 2007, the FASB issued SFAS No. 160, *Noncontrolling Interests in Consolidated Financial Statements—an amendment of ARB No. 51* (SFAS No. 160). SFAS No. 160 establishes accounting and reporting standards for the noncontrolling interest in a subsidiary and for the deconsolidation of a subsidiary. It clarifies that a noncontrolling interest in a subsidiary is an ownership interest in the consolidated entity that should be reported as equity in the consolidated financial statements. SFAS No. 160 is effective for fiscal years, and interim periods within those fiscal years, beginning on or after December 15, 2008. SFAS No. 160 requires retroactive adoption of the presentation and disclosure requirements for existing minority interests. All other requirements of SFAS No. 160 shall be applied prospectively. We are currently evaluating the potential impact of the adoption of SFAS No. 160 on our consolidated financial statements.

OFFERING COSTS

The General Partner funded all of the organization and offering costs on the Partnership's behalf. As of December 31, 2007, we had reimbursed the General Partner for approximately \$5,343,465 of organization and offering costs. The offering costs, which include items such as legal and accounting fees, marketing, and promotional printing costs are treated as a reduction of partners' capital along with sales commissions of 7.75% for the first \$1,000,000 of Unit sales and 7.25% thereafter and dealer manager fees of 3.25%. (See Note 5.)

CASH AND CASH EQUIVALENTS

For purposes of the consolidated statements of cash flows, we consider all highly liquid debt instruments purchased with an original maturity of three months or less to be cash equivalents. At December 31, 2007, cash and cash equivalents consisted of demand deposits at commercial banks of \$1.0 million and \$18.3 million of government security money market funds. At December 31, 2006, cash and cash equivalents consisted of \$796,000 of demand deposits at commercial banks.

INTEREST

Interest is charged to interest expense as it accrues. No interest was capitalized during the 2007 or 2006 periods.

3. INVESTMENTS IN NON-CONSOLIDATED ENTITIES

During 2006 and 2007, we made the following investments in four entities through which we own an interest:

- In December 2006, we acquired a 50% interest in AmREIT Casa Linda, LP which owns a multi-tenant retail property located in Dallas, Texas with a combined gross leasable area of 325,000 square feet. The remaining 50% is owned by AmREIT Monthly Income and Growth Fund III, Ltd., an affiliate AmREIT entity.
- In August 2007, we acquired a 40% interest in AmREIT Woodlake Square, LP which owns a multi-tenant retail property located in Houston, Texas with a combined gross leasable area of 206,000 square feet. The remaining 60% is owned by affiliated AmREIT entities, AmREIT Monthly Income and Growth Fund III, Ltd. and AmREIT Realty Investment Company.

- In November 2007, we acquired a 40% interest in AmREIT Westheimer Gessner, LP which owns a multi-tenant retail property located in Houston, Texas with a combined gross leasable area of 82,120 square feet. The remaining 60% is owned by affiliated AmREIT entities, AmREIT Monthly Income and Growth Fund III, Ltd. and AmREIT Realty Investment Company.
- In December 2007, we acquired a 50% interest in Cambridge Holcombe, LP which consists of raw land that will be developed into a multi-tenant retail property located in Houston, Texas. The remaining 50% is owned by an unaffiliated third party.

We report our investments in these entities using the equity method of accounting due to our ability to exercise significant influence over them. Combined condensed financial information for our non-consolidated entities (at 100%) is summarized as of and for the year ended December 31, 2007 and for the period from October 10, 2006 (inception) through December 31, 2006 as follows:

Combined Balance Sheets	As of December 31, 2007	As of December 31, 2006
	(In thousands)	(In thousands)
Assets		
Property		
Land & building	\$ 101,948	\$ 41,065
Tenant & leasehold improvements	1,607	442
Accumulated depreciation	(1,889)	(96)
Cash	785	0
Intangible Lease Cost	5,554	2,695
Other assets	10,052	7,133
Total Assets	\$118,057	\$51,239
Liabilities and partners' capital:		
Notes payable (1)	\$ 69,558	\$ 37,950
Accounts payable	3,792	337
Asset Retirement Obligations	1,500	800
Other liabilities	3,645	1,720
Partners capital	39,562	10,432
Total Liabilities and Partner's Capital	\$ 118,057	\$ 51,239
MIG IV share of net assets	§ 17,100	\$ 5,216

(1) 55% of the notes payable is a fixed rate of approximately 5.48% with a remaining term of 6 years. 45% of the notes payable is variable rate with an average rate of 6.91%.

Combined Statements of Operations	2007	200	6
	(In thousands)	(In thous	sands)
Revenue			
Total Revenue	\$6,222	\$396	
Expense			
Interest	2,737	142	
Depreciation and amortization	3,374		221
Property Expense	2,211	88	
Other	291	1	
Total expense	8,613	452	
Net loss	\$(2,391) \$(56	
MIG IV share of net loss	\$(1,123) \$(28)

4. PARTNERS' CAPITAL AND MINORITY INTEREST

AmREIT invested \$800,000 as a limited partner and \$1,000 as a general partner. We began raising capital in December 2006 and had raised approximately \$44 million at December 31, 2007. AmREIT's \$800,000 investment represents a 1.8% limited partner interest in the Partnership.

<u>Limited Optional Redemption</u> — Our limited partnership units were sold pursuant to exemptions from registration under the Securities Act of 1933 and are not currently listed on a national exchange. These units will be transferable only if we register them under such laws, and such registration is not expected. We do not anticipate that any public market for the limited partnership units will develop. In order to provide limited partners with the possibility of liquidity, limited partners who have held their shares for at least three years may receive the benefit of interim liquidity by presenting all of those units to the Partnership for redemption. At that time, we may, at our sole election and subject to the conditions and limitations described below, redeem the shares presented for cash to the extent that we have sufficient funds available to us to fund such redemption. The redemption price to be paid will be 92% of the limited partner's unreturned invested capital. At no time during a 12-month period, however, may the number of limited partner units redeemed by us exceed 2% of the number of units outstanding at the beginning of that 12-month period. We had no redemptions during the 2007 or 2006 periods.

<u>Distributions</u> — During the operating stage of the Partnership, net cash flow, as defined, will be distributed 99% to the limited partners and 1% to the General Partner. A current distribution of 7.5% per annum on invested capital has been paid to date during the operating stage. All distributions to date have been a return of capital. During the liquidation stage of the Partnership (commencing in November 2013, unless extended) net cash flow, as defined, will be distributed among the limited partners and the General Partner in the following manner:

- First 100% to the Limited Partners (in proportion to their unreturned actual invested capital) until such time as the Limited Partners have received cumulative distributions from all sources equal to 100% of their actual invested capital (calculated using the actual purchase price per unit);
- Second 100% to the General Partner until it has received cumulative distributions from all sources equal to 100% of its actual invested capital of \$1,000;
- Third 1% to the General Partner and 99% to the limited partners on a per unit basis until such time as the limited partners have received cumulative distributions from all sources equal to 8.5% per annum, cumulative, uncompounded return on their unreturned deemed capital contributions (which will be equal to (i) the product of \$25,000 per unit (regardless of the purchase price paid for a unit) multiplied by the number of units owned by a partner, reduced by (ii) the aggregate amount of any distributions received that constitute a return of capital contributions);
- Fourth 100% to the General Partner until it has received cumulative distributions from all sources (other than with respect to the Units it purchased) in an amount equal to 40% of the net cash flow paid to date to the Limited Partners in excess of their actual invested capital; and
- Thereafter 60% to the limited partners on a per unit basis and 40% to the General Partner.

5. RELATED PARTY TRANSACTIONS

We have no employees or officers. Certain of our affiliates received fees and compensation during the organizational stage of the Partnership, including securities commissions and due diligence reimbursements, marketing reimbursements and reimbursement of organizational and offering expenses. In the event that these companies are unable to provide us with the respective services, we would be required to find alternative providers of these services. The following table summarizes the amount of such compensation paid to our affiliates during the 2007 and 2006 periods:

Type of Service	Service Description & Compensation	2007	2006
Dealer Manager Fees	Dealer manager fees (3.25%) received for placement of the limited partnership units.	\$4,744,162	\$-
Organizational and Offering Cost Reimbursements	Reimbursement of the Partnership's organizational and offering costs, including legal and accounting fees, printing costs, filing fees and distribution costs.	198,784	410,888
Total		\$4,942,946	\$410,888

Additionally, certain of our affiliates receive fees and compensation during the operating stage of the Partnership, including compensation for providing services to us in the areas of asset management, development and acquisitions, property management and leasing, financing, brokerage and administration. The following table summarizes the amount of such compensation paid to our affiliates during the 2007 and 2006 periods:

Type of Service	Service Description & Compensation	2007	2006
Asset Management	A fee equal to 1.0% of net invested capital under management for accounting related services, investor relations, facilitating the deployment of capital, and other services provided by our General Partner to us.	\$202,621	¢ 450
Development and Redevelopment	Development and redevelopment fees on properties we acquire an interest in and for which we intend to develop, redevelop or substantially renovate. These fees will be based on the total project costs, including the cost of acquiring the property, and will be paid as project costs are incurred. These fees shall not exceed, for any single property, 6% of the project cost up to \$10 million, 5% of the project cost between \$10 million and \$20 million, and 4% of the project cost in excess of \$20 million. We will not pay both acquisition fees and development or redevelopment fees on the same property.	\$203,621	\$458
Property Management and Leasing	Property management fees not to exceed 4% of the gross revenues (including, without limitation, base rent, percentage rent and expense reimbursement) received from multi-tenant or multi-pad properties, for providing management, operating, maintenance and other services required to maintain a property. Leasing fees not to exceed 4% of base rent on a lease renewal and not to exceed 6% of base rent on an initial lease; provided, however, on leases of over 10,000 square feet, leasing commissions may be paid on the basis of square footage, not to exceed \$6.00 per square foot.		-
Acquisition	A fee not to exceed 6% of the sales price on co-brokered transactions and not to exceed 4% of the sales price on individually brokered transactions. Additionally, our General Partner and its affiliates will not be paid real estate brokerage commissions on the sale of a property if the property being sold has not generated an annual return of at least 8.5% per annum on the equity contributed to such property.	-	-
Reimbursement of Operating Expenses	We reimburse the actual expenses incurred by our General Partner for performing acquisition, development, management and administrative functions for us, including construction and construction management fees for development and redevelopment projects.	46 241	-
	Tatal	46,341 \$249,962	<u>-</u> \$458
	Total	φ 4 42,304	9430

See also Note 3 regarding investments in non-consolidated entities.

6. COMMITMENTS AND CONTINGENCIES

<u>Litigation</u> – In the ordinary course of business, we may become subject to litigation or claims. There are no material pending legal proceedings known to be contemplated against us.

<u>Environmental matters</u> – In connection with the ownership and operation of real estate, we may be potentially liable for costs and damages related to environmental matters. We have not been notified by any governmental authority of any non-compliance, liability or other claim.

In conjunction with our acquisitions of the Casa Linda and Woodlake Shopping Centers in December 2006 and August 2007, respectively, we identified environmental exposures caused by businesses that operated on the property prior to our ownership. The Casa Linda and Woodlake Shopping Centers joint ventures recorded asset retirement obligations of \$800,000 and \$700,000, respectively, on their acquisition dates related to these exposures. We believe that these matters will not have a material adverse effect on our consolidated financial position or results of operations, and we are aware of no other environmental exposures.

7. SUBSEQUENT EVENTS

In February 2008, we contributed approximately \$5.1 million to a joint venture owned 10% by us, 10% by an affiliate of ours and 80% by a third party institutional investor. Simultaneously, the joint venture acquired the Shadow Creek Ranch Town Center, a 616,000 square foot retail center located in Pearland, Texas.

In March 2008, we acquired a 36,000 square foot retail shopping center in Texas for \$8.8 million.

On April 1, 2008, we closed the Offering after having raised approximately \$50.0 million.

Amreit monthly income & growth fund IV, L.P. and subsidiary consolidated balance sheets

March 31, 2008 and December 31, 2007

(in thousands)

	March 31,	December 31,
ACCEPTEC	2008	2007
ASSETS	(unaudited)	
Real estate investments at cost:	¢2.000	¢.
Land	\$3,098	\$- -
Buildings Target improvements	5,319 172	-
Tenant improvements		
T 1/11 1/2 1 2 2	8,589	-
Less accumulated depreciation and amortization	(5)	_
	8,584	-
*	21.012	15 100
Investment in non-consolidated entities	21,813	17,100
Intangible lease cost, net	714	
Net real estate investments	31,111	17,100
Cash and cash equivalents	13,903	19,349
Escrow deposits	430	1,540
Accounts receivable	110	113
Accounts receivable - related party	1,111	-
Deferred costs, net	36	-
Other assets	50	10
TOTAL ASSETS	\$46,751	\$38,112
LIABILITIES AND PARTNERS' CAPITAL		
Liabilities:		
Notes payable	\$6,200	\$-
Accounts payable	209	104
Accounts payable - related party	595	1,208
Below market leases, net	155	-
Security deposits	48	_
TOTAL LIABILITIES	7,207	1,312
Partners' capital:		
General partner		
Limited partners, 1,956 and 1,796 units outstanding at	-	-
March 31, 2008 and December 31, 2007, respectively	39,544	36,800
TOTAL PARTNERS' CAPITAL	39,544	36,800
TOTAL LIABILITIES AND PARTNERS' CAPITAL	\$46,751	\$38,112
	4 10,72	

Amreit monthly income & growth fund IV, L.P. and subsidiary consolidated statements of operations

(in thousands) (unaudited)

Three months ended March

	31,		,11
	2008	2007	
D			
Revenues:	Φ 1 4	¢.	
Rental income from operating leases	\$14	<u>\$-</u>	
Total revenues	14	-	
Expenses:			
General and administrative	16	-	
General and administrative - related party	30	-	
Asset management fees - related party	105	12	
Legal and professional	50	29	
Depreciation and amortization	12	-	
Total expenses	213	41	
Operating loss	(199) (41)
Other income (expense):			
Interest and other income	154	3	
Interest expense	(6) -	
Interest expense - related party	<u>-</u>	(101)
Equity in losses from non-consolidated entities	(705) (198)
Margin tax expense	(1) -	
Total other income (expense)	(558) (296)
Net loss	\$(757) \$(337)

Amreit monthly income & growth fund IV, L.P. and subsidiary consolidated statements of partners' capital

(in thousands) (unaudited)

	General partner	Limited partners	Total	_
Balance at December 31, 2007	<u>\$-</u>	\$36,800	\$36,800	_
Contributions, net of offering costs Net loss (1) Distributions	- 9 (9	4,368 (766) (858	4,368) (757) (867)
Balance at March 31, 2008	\$ -	\$39,544	\$39,544	

(1) The allocation of net loss includes a curative allocation to increase the GP capital account by \$17. The cumulative curative allocation since inception is \$42. The partnership agreement provides that no partner shall be required to fund a deficit balance in their capital account.

Amreit Monthly Income & Growth fund IV, L.P. and Subsidiary consolidated statements of cash flows

(in thousands) (unaudited)

Three months ended March

	31,	
	2008	2007
Cash flows from operating activities:		
Net loss	\$(757) \$(337
Adjustments to reconcile net loss to net cash		
used in operating activities:		
Equity in losses from non-consolidated entities	705	198
Depreciation and amortization	11	-
Decrease in accounts receivable	3	=
Increase in accounts receivable - related party	(211) -
Increase in escrow deposits	(390) -
Increase in other assets	(40) -
Increase in accounts payable	105	23
Decrease in accounts payable - related party	(613) (60
Increase in security deposits	48	=
Net cash used in operating activities	(1,139) (176
Cash flows from investing activities:		
Acquisition of investment properties	(2,954) -
Investment in non-consolidated entities, net	(4,818) -
Net cash used in investing activities	(7,772) -
Cash flows from financing activities:		
Proceeds from notes payable - related party	-	101
Contributions	4,899	9,268
Issuance costs	(531) (979
Loan acquisition costs	(36) -
Distributions	(867) (44
Net cash provided by financing activities	3,465	8,346
Net (decrease) increase in cash and cash equivalents	(5,446) 8,170
Cash and cash equivalents, beginning of period	19,349	796
Cash and cash equivalents, end of period	\$13,903	\$8,966
Supplemental schedule of cash flow information:		
Supplemental senerate of easi non information.		
Cash paid during the period for interest	\$6	\$-

Supplemental schedule of noncash investing and financing activities:

In conjunction with our acquisition of the Village on the Green property in March 2008, we placed \$6.2 million in debt that was funded directly to the buyer.

In conjunction with our investment in the Shadow Creek Ranch property, we placed \$1.5 million in escrow deposits in December 2007. When we acquired the property in February 2008, an affiliated AmREIT entity purchased an interest in the property, resulting in \$900,000 of the escrow amount being reclassified to accounts receivable - related party. The remaining \$600,000, which represents our portion of the escrow, was reclassified to investment in non-consolidated entities. We were reimbursed the \$900,000 during the second quarter of 2008.

Amreit Monthly income & growth fund IV, L.P. and Subsidiary

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

March 31, 2008

(unaudited)

1. DESCRIPTION OF BUSINESS AND NATURE OF OPERATIONS

AmREIT Monthly Income & Growth Fund IV, L.P., a Delaware limited partnership (herein after referred to as the "Partnership," "MIG IV," "we," "us" or "our"), was formed on October 10, 2006 to acquire, develop and operate directly or indirectly, a portfolio of commercial real estate consisting primarily of multi-tenant shopping centers and mixed-use developments. The General Partner of the Partnership is AmREIT Monthly Income & Growth IV Corporation, a Delaware Corporation (the "General Partner"), which is a subsidiary of AmREIT, a Texas real estate investment trust. The General Partner maintains its principal place of business in Houston, Texas.

We commenced our principal operations on December 8, 2006 when we acquired our first interest in a property. On January 12, 2007 we raised the minimum offering of \$1.0 million pursuant to the terms of our Offering Memorandum dated November 15, 2006 (the "Offering Memorandum"), and issued the initial 40 limited partnership units (the "Units"). As of March 31, 2008, we had received \$48.9 million from the sale of 1,956 Units and closed the Offering. At March 31, 2008, we directly owned one property comprising 36,000 square feet of gross leasable area and owned an investment interest in five other properties comprising 1,231,000 square feet of gross leasable area.

Our Units were sold pursuant to exemptions from registration under the Securities Act of 1933, as amended, and are not currently listed on a national exchange. These Units will be transferable only if we register them under such laws, and such registration is not expected, or pursuant to an exemption under the Securities Act of 1933, as amended, and applicable state securities laws. We do not anticipate that any public market for the Units will develop.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

BASIS OF PRESENTATION

Our financial records are maintained on the accrual basis of accounting whereby revenues are recognized when earned and expenses are recorded when incurred. The consolidated financial statements include our accounts as well as the accounts of any wholly- or majority-owned subsidiaries in which we have a controlling financial interest (See Note 9). Investments in joint ventures and partnerships where we have the ability to exercise significant influence but do not exercise financial and operating control, are accounted for using the equity method. All significant inter-company accounts and transactions have been eliminated in consolidation.

The consolidated financial statements included in this report are unaudited; however, amounts presented in the consolidated balance sheet as of December 31, 2007 are derived from our audited financial statements at that date. In our opinion, all adjustments necessary for a fair presentation of such financial statements have been included. Such adjustments consisted of normal recurring items.

REVENUE RECOGNITION

We lease space to tenants under agreements with varying terms. Our leases are accounted for as operating leases with revenue being recognized on a straight-line basis over the terms of the individual leases. Accrued rents are included in tenant receivables. Revenue from tenant reimbursements of taxes, maintenance expenses and insurance is recognized in the period the related expense is recorded. Additionally, certain of the lease agreements contain provisions that provide for additional rents based on tenants' sales volumes (contingent or percentage rent). Percentage rents are recognized when the tenants achieve the specified targets as defined in their lease agreements. During the three months ended March 31, 2008 and 2007, there were no percentage rents recognized. We recognize lease termination fees in the period that the lease is terminated and collection of the fees is reasonably assured. During the three months ended March 31, 2008 and 2007, there were no lease termination fees recognized.

REAL ESTATE INVESTMENTS

<u>Development Properties</u> - Land, buildings and improvements are recorded at cost. Expenditures related to the development of real estate are carried at cost which includes capitalized carrying charges, acquisition costs and development costs. Carrying charges, primarily interest, real estate taxes and loan acquisition costs, and direct and indirect development costs related to buildings under construction, are capitalized as part of construction in progress. The capitalization of such costs ceases at the earlier of one year from the date of completion of major construction or when the property, or any completed portion, becomes available for occupancy. We capitalize acquisition costs as incurred.

Such costs are expensed if and when the acquisition becomes no longer probable. During the three months ended March 31, 2008 and 2007, we did not capitalize any interest or taxes on properties under development.

Acquired Properties and Acquired Lease Intangibles - We account for real estate acquisitions pursuant to Statement of Financial Accounting Standards No. 141, Business Combinations (SFAS No. 141). Accordingly, we allocate the purchase price of the acquired properties to land, building and improvements, identifiable intangible assets and to the acquired liabilities based on their respective fair values. Identifiable intangibles include amounts allocated to acquired out-of-market leases, the value of in-place leases and customer relationship value, if any. We determine fair value based on estimated cash flow projections that utilize appropriate discount and capitalization rates and available market information. Estimates of future cash flows are based on a number of factors including the historical operating results, known trends and specific market and economic conditions that may affect the property. Factors considered by management in our analysis of determining the as-if-vacant property value include an estimate of carrying costs during the expected lease-up periods considering market conditions, and costs to execute similar leases. In estimating carrying costs, management includes real estate taxes, insurance and estimates of lost rentals at market rates during the expected lease-up periods, tenant demand and other economic conditions. Management also estimates costs to execute similar leases including leasing commissions, tenant improvements, legal and other related expenses. Intangibles related to out-of-market leases and in-place lease value are recorded as acquired lease intangibles and are amortized as an adjustment to rental revenue or amortization expense, as appropriate, over the remaining terms of the underlying leases. Premiums or discounts on acquired out-of-market debt are amortized to interest expense over the remaining term of such debt.

<u>Depreciation</u> - Depreciation is computed using the straight-line method over an estimated useful life of up to 50 years for buildings, up to 20 years for site improvements and over the term of lease for tenant improvements.

<u>Properties Held for Sale</u> - Properties will be classified as held for sale if we have decided to market the property for immediate sale in its present condition with the belief that the sale will be completed within one year. Operating properties held for sale will be carried at the lower of cost or fair value less cost to sell. Depreciation and amortization will be suspended during the held for sale period. As of March 31, 2008 and December 31, 2007, we had no properties held for sale.

<u>Impairment</u> - We review our properties for impairment whenever events or changes in circumstances indicate that the carrying amount of the assets, including accrued rental income, may not be recoverable through operations. We determine whether an impairment in value occurred by comparing the estimated future cash flows (undiscounted and without interest charges), including the residual value of the property, with the carrying value of the individual property. If impairment is indicated, a loss will be recorded for the amount by which the carrying value of the asset exceeds its fair value. No impairment charges were recognized during the three months ended March 31, 2008 and 2007.

ENVIRONMENTAL EXPOSURES

We are subject to numerous environmental laws and regulations as they apply to real estate pertaining to chemicals used by the dry cleaning industry, the existence of asbestos in older shopping centers, and underground petroleum storage tanks. We believe that the tenants who currently operate dry cleaning plants or gas stations do so in accordance with current laws and regulations. We believe that the ultimate disposition of current known environmental matters will not have a material affect on our financial position, liquidity, or operations (See Note 10). However, we can give no assurance that existing environmental studies with respect to the shopping centers have revealed all potential environmental liabilities; that any previous owner, occupant or tenant did not create any material environmental condition not known to it; that the current environmental condition of the shopping centers will not be affected by tenants and occupants, by the condition of nearby properties, or by unrelated third parties; or that changes in applicable environmental laws and regulations or their interpretation will not result in additional environmental liability to the Partnership.

RECEIVABLES

<u>Accounts Receivable-Related Party</u> - included in accounts receivable related party are short-term cash advances provided to certain of our affiliated investment entities primarily for their working capital needs. These cash advances are due upon demand.

DEFERRED COSTS

Deferred costs include deferred leasing costs and loan acquisition costs, net of amortization. Loan acquisition costs are incurred in obtaining financing and are amortized to interest expense over the term of the debt agreements. Deferred leasing costs consist of external commissions associated with leasing our properties and are amortized to expense over the lease term. We had no accumulated amortization related to loan acquisition costs or leasing costs as of March 31, 2008 and December 31, 2007.

INCOME TAXES

<u>Federal</u> - No provision for U.S. federal income taxes is included in the accompanying consolidated financial statements. As a partnership, we are not subject to federal income tax, and the federal tax effect of our activities is passed through to our partners.

<u>State</u> - In May 2006, the State of Texas adopted House Bill 3, which modified the state's franchise tax structure, replacing the previous tax based on capital or earned surplus with one based on margin (often referred to as the "Texas Margin Tax") effective with franchise tax reports filed on or after January 1, 2008. The Texas Margin Tax is computed by applying the applicable tax rate (1% for the Partnership) to the profit margin, which, generally, will be determined for us as total revenue less a 30% standard deduction. Although House Bill 3 states that the Texas Margin Tax is not an income tax, the Partnership believes that SFAS No. 109, *Accounting for Income Taxes*, applies to the Texas Margin Tax. We have recorded a tax provision of approximately \$1,000 and \$0 for the Texas Margin Tax for three month periods ended March 31, 2008 and 2007, respectively.

USE OF ESTIMATES

The preparation of consolidated financial statements in conformity with U.S. generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

FAIR VALUE OF FINANCIAL INSTRUMENTS

Our consolidated financial instruments consist primarily of cash and cash equivalents, accounts receivable, accounts receivable-related party, accounts payable, accounts payable-related party, and notes payable. The carrying value of cash and cash equivalents, accounts receivable, accounts receivable-related party, accounts payable, and accounts payable-related party, are representative of their respective fair values due to the short-term nature of these instruments. As of March 31, 2008 and December 31, 2007, the carrying value of our total debt obligations was \$6.2 million and \$0, respectively, all of which represent fixed-rate obligations. The carrying value of these obligations approximated their fair value, based on a discounted cash flow analysis using current market rates of interest.

CONSOLIDATION OF VARIABLE INTEREST ENTITIES

In December 2003, the Financial Accounting Standards Board ("FASB") reissued Interpretation No. 46, *Consolidation of Variable Interest Entities* (FIN No. 46R), as revised. FIN No. 46R addresses how a business enterprise should evaluate whether it has a controlling financial interest in an entity through means other than voting rights. FIN No. 46R requires a variable interest entity to be consolidated by a company that is subject to a majority of the risk of loss from the variable interest entity's activities or entitled to receive a majority of the entity's residual returns or both. Disclosures are also required about variable interest entities in which a company has a significant variable interest but that it is not required to consolidate. As of March 31, 2008 and December 31, 2007, we were not invested in any entities that qualify as variable interest entities pursuant to FIN No. 46R.

NEW ACCOUNTING STANDARDS

In September 2006, the FASB issued SFAS No. 157, *Fair Value Measurements* (SFAS No. 157). SFAS No. 157 defines fair value, establishes a framework for measuring fair value and requires enhanced disclosures about fair value measurements. SFAS No. 157 requires companies to disclose the fair value of its financial instruments according to a fair value hierarchy. Additionally, companies are required to provide certain disclosures regarding instruments within the hierarchy, including a reconciliation of the beginning and ending balances for each major category of assets and liabilities. SFAS No. 157 is effective for our fiscal year beginning January 1, 2008. The adoption of SFAS No. 157 is not expected to have a material effect on our results of operations or financial position.

In February 2007, the FASB issued SFAS No. 159, *The Fair Value Option for Financial Assets and Financial Liabilities* (SFAS No. 159). SFAS No. 159 expands opportunities to use fair value measurement in financial reporting and permits entities to choose to measure many financial instruments and certain other items at fair value. This Statement is effective for fiscal years beginning after November 15, 2007. We currently do not plan to measure any eligible financial assets and liabilities at fair value under the provisions of SFAS No. 159.

In December 2007, the FASB revised SFAS No. 141 (Revised 2007), *Business Combinations* (SFAS No. 141R). SFAS No. 141R will change the accounting for business combinations. Under SFAS No. 141R, an acquiring entity will be required to recognize all the assets acquired and liabilities assumed in a transaction at the acquisition-date fair value with limited exceptions. SFAS No. 141R applies prospectively to business combinations for which the acquisition date is on or after the beginning of the first annual reporting period beginning on or after December 15, 2008. We currently capitalize acquisition costs as part of the basis of the asset acquired. Upon effectiveness of SFAS No. 141R we will expense acquisition costs as incurred.

In December 2007, the FASB issued SFAS No. 160, *Noncontrolling Interests in Consolidated Financial Statements—an amendment of ARB No. 51* (SFAS No. 160). SFAS No. 160 establishes accounting and reporting standards for the noncontrolling interest in a subsidiary and for the deconsolidation of a subsidiary. It clarifies that a noncontrolling interest in a subsidiary is an ownership interest in the consolidated entity that should be reported as equity in the consolidated financial statements. SFAS No. 160 is effective for fiscal years, and interim periods within those fiscal years, beginning on or after December 15, 2008. SFAS No. 160 requires retroactive adoption of the presentation and disclosure requirements for existing minority interests. All other requirements of SFAS No. 160 shall be applied prospectively. We are currently evaluating the potential impact of the adoption of SFAS No. 160 on our consolidated financial statements.

OFFERING COSTS

The General Partner funded all of the organization and offering costs on the Partnership's behalf. As of March 31, 2008, we had reimbursed the General Partner for approximately \$625,000 of organization and offering costs. The offering costs, which include items such as legal and accounting fees, marketing, and promotional printing costs are treated as a reduction of partners' capital along with sales commissions and dealer manager fees of 7.75% and 3.25%, respectively. (See Note 8.)

CASH AND CASH EQUIVALENTS

For purposes of the consolidated statements of cash flows, we consider all highly liquid debt instruments purchased with an original maturity of three months or less to be cash equivalents. Cash and cash equivalents consist of demand deposits at commercial banks and money market funds.

INTEREST

Interest is charged to interest expense as it accrues. No interest has been capitalized since inception of the Partnership.

SEGMENT REPORTING

The FASB issued SFAS No. 131, *Disclosures about Segments of an Enterprise and Related Information*, which establishes standards for reporting financial and descriptive information about an enterprise's reportable segments. We determined that we have one reportable segment, with activities related to investing in real estate. Our investments in real estate generate rental revenue and other income through the leasing of multi-tenant retail properties, which comprised 100% of our total consolidated revenues for all periods presented. We evaluate operating performance on an individual property level. However, as each of our properties have similar economic characteristics, tenants and products and services, our properties have been aggregated into one reportable segment.

3. INVESTMENTS IN NON-CONSOLIDATED ENTITIES

Since inception, we made the following investments in five entities through which we own an interest in five properties:

- In February 2008, we acquired a 10% interest in Shadow Creek Holding Company LLC which owns a multi-tenant retail property located in Pearland, Texas with a combined gross leasable area of 616,370 square feet. The remaining 90% is owned by an unaffiliated third party and AmREIT Realty Investment Company, an affiliated AmREIT entity.
- In December 2007, we acquired a 50% interest in Cambridge Holcombe, LP which consists of raw land that will be developed into a multi-tenant retail property located in Houston, Texas. The remaining 50% is owned by an unaffiliated third party.
- In November 2007, we acquired a 40% interest in AmREIT Westheimer Gessner, LP which owns a multi-tenant retail property located in Houston, Texas with a combined gross leasable area of 82,120 square feet. The remaining 60% is owned by affiliated AmREIT entities, AmREIT Monthly Income & Growth Fund III, Ltd. and AmREIT Realty Investment Company.
 - In August 2007, we acquired a 40% interest in AmREIT Woodlake, LP which owns a multi-tenant retail property located in Houston, Texas with a combined gross leasable area of 206,000 square feet. In December 2007, AmREIT Woodlake, LP
- entered into an interest rate swap to hedge the interest rate risk associated with the property's variable rate loan. The remaining 60% is owned by affiliated AmREIT entities, AmREIT Monthly Income & Growth Fund III, Ltd. and AmREIT Realty Investment Company. (See Note 11).
- In December 2006, we acquired a 50% interest in AmREIT Casa Linda, LP which owns a multi-tenant retail property located in Dallas, Texas with a combined gross leasable area of 325,000 square feet. The remaining 50% is owned by AmREIT Monthly Income & Growth Fund III, Ltd., an affiliated AmREIT entity.

We report our investments in these entities using the equity method of accounting due to our ability to exercise significant influence over them. Combined condensed financial information for our significant equity entities (at 100%) is summarized as of and for the three months ended March 31, 2008 and 2007 as follows (in thousands):

Three months ended March 31.

	11241 011 0 19			
	2008	2007		
Revenue	\$830	\$-		
Depreciation and amortization	(506) -		
Net loss	\$(1,324) \$-		

4. ACQUIRED LEASE INTANGIBLES

In accordance with SFAS No. 141, we have identified and recorded the value of intangibles at the property acquisition date. Such intangibles include the value of in-place leases and out-of-market leases. Acquired lease intangible assets (in-place leases and above-market leases) are amortized over the leases' remaining terms, which range from 9 months to 7 years. The amortization of above-market leases is recorded as a reduction of rental income and the amortization of in-place leases is recorded to amortization expense. The amortization expense related to in-place leases was approximately \$7,000 and \$0 for the three months ended March 31, 2008 and 2007, respectively. The amortization of above-

market leases, which was recorded as a reduction of rental income, was approximately \$0 during the three months ended March 31, 2008 and 2007, respectively.

In-place and above-market lease amounts and their respective accumulated amortization are as follows (in thousands):

	March 31, 2008		December	31, 2007
		Above-market		Above-
	In-Place leases	leases	In-Place leases	market leases
Cost	\$707	\$14	\$-	\$-
Accumulated amortization	_(7)			
Intangible lease cost, net	\$700	\$14	\$-	\$-

Acquired lease intangible liabilities (below-market leases) of \$155,000 and \$0 as of March 31, 2008 and December 31, 2007, respectively, are net of previously accreted minimum rent of approximately \$1,000 and \$0 at March 31, 2008 and December 31, 2007, respectively. Below-market leases are accreted over the leases' remaining terms, which range from 9 months to 7 years. Accretion of below market leases was approximately \$1,000 and \$0 during the three months ended March 31, 2008 and 2007, respectively. Such accretion is recorded as an increase to rental income.

5. NOTES PAYABLE

Our outstanding debt at March 31, 2008 consisted entirely of a fixed-rate mortgage loan of \$6.2 million. We had no outstanding debt as of December 31, 2007. Our mortgage loan is secured by a real estate property and may be prepaid, but could be subject to a yield-maintenance premium or prepayment penalty. Our mortgage loan is due in monthly installments of interest and principal and matures in April 2017.

As of March 31, 2008, the weighted-average interest rate on our fixed-rate debt is 5.5%, and the weighted average remaining life of such debt is 9.02 years. As of March 31, 2008, scheduled principal repayments on notes payable were as follows (in thousands):

Scheduled Principal Payments	Term-Loan Maturities	Total Payments
\$-	\$-	\$-
51	-	51
81	=	81
86	-	86
90	-	90
449	5,443	5,892
\$757	\$5,443	\$6,200
	Principal Payments \$- 51 81 86 90 449	Principal Payments Term-Loan Maturities \$- \$- 51 - 81 - 86 - 90 - 449 5,443

6. CONCENTRATIONS

As of March 31, 2008, our single consolidated property individually comprised greater than 20% of our consolidated total assets. Consistent with our strategy of investing in areas that we know well, this property is located in San Antonio, Texas' third largest metropolitan area.

Following are the base rents generated by our top tenants during the three months ended March 31, 2008 and 2007 (in thousands):

Three months ended
March 31,

Tenant	2008	2007
Paesano's Restaurant	\$3	\$-
Alamo Heights Pediatrics	1	-
Theo & Herb Designer Shoes	1	=
Rouse Dental Office	1	-
The Mutual Fund Store	1	<u> </u>
	\$7	\$-

7. PARTNERS' CAPITAL

AmREIT invested \$800,000 as a limited partner and \$1,000 as a general partner in MIG IV. We began raising capital in December 2006. We closed the Offering on March 31, 2008 when we had raised approximately \$49 million. AmREIT's \$800,000 investment represents a 1.6% limited partner interest in the Partnership.

<u>Limited Optional Redemption</u> — Our limited partnership units were sold pursuant to exemptions from registration under the Securities Act of 1933 and are not currently listed on a national exchange. These units will be transferable only if we register them under such laws, and such registration is not expected. We do not anticipate that any public market for the limited partnership units will develop. In order to provide limited partners with the possibility of liquidity, limited partners who have held their shares for at least three years may receive the benefit of interim liquidity by presenting all of those units to the Partnership for redemption. At that time, we may, at our sole election and subject to the conditions and limitations described below, redeem the shares presented for cash to the extent that we have sufficient funds available to us to fund such redemption. The redemption price to be paid will be 92% of the limited partner's unreturned invested capital. At no time during a 12-month period, however, may the number of limited partner units redeemed by us exceed 2% of the number of units outstanding at the beginning of that 12-month period. We had no redemptions during the 2008 or 2007 periods.

<u>Distributions</u> — During the operating stage of the Partnership, net cash flow, as defined, will be distributed 99% to the limited partners and 1% to the General Partner. A current distribution of 7.5% per annum on invested capital has been paid to date during the operating stage. All distributions to date have been a return of capital. During the liquidation stage of the Partnership (commencing in November 2013, unless extended) net cash flow, as defined, will be distributed among the limited partners and the General Partner in the following manner:

- First 100% to the Limited Partners (in proportion to their unreturned actual invested capital) until such time as the Limited Partners have received cumulative distributions from all sources equal to 100% of their actual invested capital (calculated using the actual purchase price per unit);
- Second 100% to the General Partner until it has received cumulative distributions from all sources equal to 100% of its actual invested capital of \$1,000;
- Third 1% to the General Partner and 99% to the limited partners on a per unit basis until such time as the limited partners have received cumulative distributions from all sources equal to 8.5% per annum, cumulative, uncompounded return on their unreturned deemed capital contributions (which will be equal to (i) the product of \$25,000 per unit (regardless of the purchase price paid for a unit) multiplied by the number of units owned by a partner, reduced by (ii) the aggregate amount of any distributions received that constitute a return of capital contributions);
- Fourth 100% to the General Partner until it has received cumulative distributions from all sources (other than with respect to the Units it purchased) in an amount equal to 40% of the net cash flow paid to date to the Limited Partners in excess of their actual invested capital; and
- Thereafter 60% to the limited partners on a per unit basis and 40% to the General Partner.

8. RELATED PARTY TRANSACTIONS

Certain of our affiliates received fees and compensation during the organizational stage of the Partnership, including securities commissions and due diligence reimbursements, marketing reimbursements and reimbursement of organizational and offering expenses. In the event that these companies are unable to provide us with the respective services, we would be required to find alternative providers of these services. The following table summarizes the amount of such compensation paid to our affiliates during the three months ended March 31, 2008 and 2007 (in thousands):

	Three months ended March 31,			
Type of Service	2008	2007		
Securities Commissions, Due Diligence and Marketing				
Reimbursements	\$515	\$935		
Organizational and Offering Cost Reimbursements	15	59		
Asset Management Fees	105	12		
Development and Acquisition Fees	264	-		
Administrative Costs Reimbursements	30	-		
	\$929	\$1,006		

In addition to the above fees paid by us, the non-consolidated entities in which we have investments pay property management, leasing, acquisition and certain other real estate fees to one of our affiliated entities. See also Note 3 regarding investments in non-consolidated entities.

9. REAL ESTATE ACQUISITIONS AND DISPOSITIONS

During the three months ended March 31, 2008, we invested approximately \$3.2 million through the acquisition of one property. On March 25, 2008, we acquired Village on the Green, a multi-tenant retail property in San Antonio, Texas with a gross leasable area of 36,000 square feet. The acquisition was accounted for as a purchase and the results of its operations are included in the accompanying consolidated financial statements from the date of acquisition.

See Note 3 for a discussion of our investment activity since our inception with respect to our non-consolidated entities.

10. COMMITMENTS AND CONTINGENCIES

<u>Litigation</u> - In the ordinary course of business, we may become subject to litigation or claims. There are no material pending legal proceedings known to be contemplated against us.

<u>Environmental matters</u> - In connection with the ownership and operation of real estate, we may be potentially liable for costs and damages related to environmental matters. We have not been notified by any governmental authority of any non-compliance, liability or other claim.

11. SUBSEQUENT EVENTS

On May 30, 2008, we acquired a 60% interest in an additional tract of land adjacent to AmREIT Westheimer Gessner, LP.

On June 6, 2008, we acquired an additional 20% interest in AmREIT Woodlake, LP and AmREIT Westheimer Gessner, LP for \$2.5 million and \$3.5 million, respectively. The additional ownership interests were acquired at their respective carrying values from AmREIT Realty Investment Company, an affiliated AmREIT entity, which still maintains a 10% interest in both partnerships. Due to these additional 20% ownership investments, we own a 60% majority interest in each partnership.

MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS FOR THE OUARTER ENDED MARCH 31, 2008.

The following discussion and analysis should be read in conjunction with the accompanying consolidated financial statements and notes thereto.

Our disclosure and analysis in these financial statements contain forward-looking statements, which provide our current expectations or forecasts of future events. Words such as "believe," "anticipate," "expect," and "intend" and similar expressions may identify forward-looking statements, but the absence of these words does not necessarily mean that a statement is not forward-looking. Forward-looking statements are subject to known and unknown risks and uncertainties and are based on potentially inaccurate assumptions that could cause actual results to differ materially from those expected or implied by the forward-looking statements. Our actual results could differ materially from those anticipated in the forward-looking statements for many reasons. You should not unduly rely on these forward-looking statements, which speak only as of the date of these financial statements. We undertake no obligation to publicly revise any forward-looking statement to reflect circumstances or events after the date of these financial statements or to reflect the occurrence of unanticipated events, except as may be required by law. You should, however, review the factors and risks we describe in the reports we file from time to time with the Securities and Exchange Commission, or SEC, after the date of these financial statements.

Overview

We are a Delaware limited partnership formed on October 10, 2006 to acquire, develop and operate, directly or indirectly through joint venture arrangements, a portfolio of commercial real estate consisting primarily of multi-tenant shopping centers and mixed-use developments throughout the Southern and Southwestern United States. Our investment strategy is to (1) sell approximately 40% of our properties when appropriate and re-invest the net sales proceeds into additional properties and (2) retain the remaining properties as income-producing assets during our entire operating period. We focus on properties characterized by high automobile traffic counts, high populations, high household incomes and limited opportunities for competition.

We have no employees and are managed by AmREIT Monthly Income & Growth IV Corporation, our General Partner, pursuant to our Partnership Agreement. Our General Partner is a wholly owned subsidiary of AmREIT, a Texas real estate investment trust with shares of common stock traded on the American Stock Exchange (AMEX: AMY). We qualify as a partnership for federal income tax purposes.

As of March 31, 2008, we directly owned one property comprising 36,000 square feet of gross leasable area and owned an investment interest in five other properties comprising 1,231,000 square feet of gross leasable area. A majority of our properties are located in highly populated, suburban communities in Texas. We derive a substantial portion of our revenue from rental income from these properties. As a result, our operating results and cash flows are primarily influenced by rental income from our properties and interest expense on our property acquisition indebtedness. Rental income accounted for 100% of our total revenue during the three months ended March 31, 2008. As of March 31, 2008, our properties were 63% leased and the average debt leverage ratio of the properties in which we have an investment was approximately 53%, with 83% of such debt carrying a fixed rate of interest.

Summary of Critical Accounting Policies

The preparation of the consolidated financial statement information contained herein requires management to make estimates and assumptions that affect the reported amounts of assets, liabilities, net sales and expenses, and related disclosure of contingent assets and liabilities. Management bases its estimates on historical experience and on various other assumptions that are believed to be reasonable under the circumstances, the results of which form the basis for making judgments about the carrying values of assets and liabilities that are not readily apparent from other sources. Actual results may differ from these estimates under different assumptions or conditions. An accounting policy is deemed to be critical if it requires an accounting estimate to be made based on assumptions about matters that are highly uncertain at the time the estimate is made, and if different estimates that reasonably could have been used, or changes in the accounting estimates that are reasonably likely to occur periodically, could materially impact the financial statements.

Revenue Recognition

We lease space to tenants under agreements with varying terms. Our leases are accounted for as operating leases with revenue being recognized on a straight-line basis over the terms of the individual leases. When we acquire a property, the term of existing leases is considered to commence as of the acquisition date for the purposes of this calculation. Accrued rents are included in tenant receivables. Revenue from tenant reimbursements of taxes, maintenance expenses and insurance is recognized in the period the related expense is recorded. Additionally, certain of the lease agreements contain provisions that grant additional rents based on tenants' sales volumes (contingent or percentage rent). We defer the recognition of contingent or percentage rental income until the specific targets as defined in lease agreements that trigger the contingent or percentage rental income are achieved. Cost recoveries from tenants are included in rental income in the period the related costs are incurred.

Valuation of Real Estate Assets

We continually monitor events and changes in circumstances that could indicate that the carrying amounts of our real estate assets, including accrued rental income, may not be recoverable through operations. When indicators of potential impairment are present that indicate that the carrying amounts of real estate and related intangible assets may not be recoverable, we assess the recoverability of the assets by determining whether the carrying value of the assets will be recovered through the undiscounted future operating cash flows expected from the use of the assets and their eventual disposition. If impairment is indicated, a loss will be recorded for the amount by which the carrying value of the asset exceeds its estimated fair value.

The estimation of expected future cash flows is inherently uncertain and relies on subjective assumptions about future and current market conditions and events that affect the ultimate value of the property. It requires us to estimate future market rental income amounts subsequent to the expiration of current lease agreements, property operating expenses, discount rates, the number of months it takes to re-lease the property and the number of years the property is held for investment. The use of inappropriate assumptions in the future cash flow analysis would result in an incorrect assessment of the property's future cash flow and fair value and could result in the overstatement of the carrying value of our real estate and related intangible assets and net income.

We have not recorded any impairment losses since the inception of the Partnership.

Investment in Real Estate Assets

Land, buildings and improvements are recorded at cost. Expenditures related to the development of real estate are carried at cost which includes capitalized carrying charges, acquisition costs and development costs. Carrying charges, primarily interest and loan acquisition costs, and direct and indirect development costs related to buildings under construction are capitalized as part of construction in progress. The capitalization of such costs ceases at the earlier of one year from the date of completion of major construction or when the property, or any completed portion, becomes available for occupancy. We capitalize acquisition costs as incurred. Such costs are expensed if and when the acquisition becomes no longer probable.

We are required to make subjective assessments as to the useful lives of our depreciable assets. We consider the period of future benefit of the asset to determine the appropriate useful lives. These assessments, which are based on estimates, have a direct impact on net income. Depreciation is computed using the straight-line method over an estimated useful life of up to 50 years for buildings, up to 20 years for site improvements and over the life of lease for tenant improvements and intangible lease costs.

Valuation of Receivables

We determine an appropriate allowance for the uncollectible portion of tenant receivables and accounts receivable based upon an analysis of balances outstanding, historical payment history, tenant credit worthiness, additional guarantees and other economic trends. Balances outstanding include base rents, tenant reimbursements and receivables attributed to the accrual of straight line rents. Additionally, we consider estimates of the expected recovery of pre-petition and post-petition claims with respect to tenants in bankruptcy in our assessment of the likelihood of collecting the related receivables.

Real Estate Acquisitions

We account for real estate acquisitions pursuant to Statement of Financial Accounting Standards No. 141, *Business Combinations* (SFAS No. 141). Accordingly, we allocate the purchase price of the acquired properties to land, building and improvements, identifiable intangible assets and to the acquired liabilities based on their respective fair values. Identifiable intangibles include amounts allocated to acquired out-of-market leases, the value of in-place leases and customer relationships, if any.

We determine fair value based on estimated cash flow projections that utilize appropriate discount and capitalization rates and available market information. Estimates of future cash flows are based on a number of factors including the historical operating results, known trends and specific market and economic conditions that may affect the property. Factors considered by management in our analysis of determining the as-if-vacant property value include an estimate of carrying costs during the expected lease-up periods considering market conditions, and costs to execute similar leases. In estimating carrying costs, management includes real estate taxes, insurance and estimates of lost rentals at market rates during the expected lease-up periods, tenant demand and other economic conditions. Management also estimates costs to execute similar leases including leasing commissions, tenant improvements, legal and other related expenses. Intangibles related to out-of-market leases and in-place lease value are recorded as acquired lease intangibles and are amortized as an adjustment to rental revenue or amortization expense, as appropriate, over the remaining terms of the underlying leases. Any premiums or discounts on acquired out-of-market debt are amortized to interest expense over the remaining term of such debt.

The fair values of above-market and below-market in-place lease values are recorded based on the present value (using an interest rate which reflects the risks associated with the leases acquired) of the difference between (1) the contractual amounts to be paid pursuant to the in-place leases and (2) our estimate of fair market lease rates for the corresponding in-place leases, measured over a period equal to the remaining non-cancelable term of the lease. The above-market and below-market lease values are capitalized as intangible lease assets or liabilities and are amortized as an adjustment of rental income over the remaining terms of the respective leases.

The fair values of in-place leases include costs associated with obtaining a new tenant, opportunity costs associated with lost rentals which are avoided by acquiring an in-place lease, and tenant relationships. Direct costs associated with obtaining a new tenant include commissions, tenant improvements, and other direct costs and are estimated based on management's consideration of current market costs to execute a similar lease. These direct costs are included in intangible lease assets in the accompanying consolidated balance sheet and are amortized to expense over the remaining terms of the respective leases. The value of opportunity costs is calculated using the contractual amounts to be paid pursuant to the in-place leases over a market absorption period for a similar lease. Customer relationships are valued based on expected renewal of a lease or the likelihood of obtaining a particular tenant for other locations. These intangibles are included in intangible lease assets in the accompanying consolidated balance sheet and are amortized to expense over the remaining term of the respective leases.

Recently Issued Accounting Pronouncements

In September 2006, the Financial Accounting Standards Board ("FASB") issued SFAS No. 157, *Fair Value Measurements* (SFAS No. 157). SFAS No. 157 defines fair value, establishes a framework for measuring fair value and requires enhanced disclosures about fair value measurements. SFAS No. 157 requires companies to disclose the fair value of its financial instruments according to a fair value hierarchy. Additionally, companies are required to provide certain disclosures regarding instruments within the hierarchy, including a reconciliation of the beginning and ending balances for each major category of assets and liabilities. SFAS No. 157 is effective for our fiscal year beginning January 1, 2008. The adoption of SFAS No. 157 is not expected to have a material effect on our results of operations or financial position.

In February 2007, the FASB issued SFAS No. 159, *The Fair Value Option for Financial Assets and Financial Liabilities* (SFAS No. 159). SFAS No. 159 expands opportunities to use fair value measurement in financial reporting and permits entities to choose to measure many financial instruments and certain other items at fair value. This Statement is effective for fiscal years beginning after November 15, 2007. We currently do not plan to measure any eligible financial assets and liabilities at fair value under the provisions of SFAS No. 159.

In December 2007, the FASB revised SFAS No. 141 (Revised 2007), *Business Combinations* (SFAS No. 141R). SFAS No. 141R will change the accounting for business combinations. Under SFAS No. 141R, an acquiring entity will be required to recognize all the assets acquired and liabilities assumed in a transaction at the acquisition-date fair value with limited exceptions. SFAS No. 141R applies prospectively to business combinations for which the acquisition date is on or after the beginning of the first annual reporting period beginning on or after

December 15, 2008. We currently capitalize acquisition costs as part of the basis of the asset acquired. Upon effectiveness of SFAS No. 141R we will expense acquisition costs as incurred.

In December 2007, the FASB issued SFAS No. 160, *Noncontrolling Interests in Consolidated Financial Statements—an amendment of ARB No. 51* (SFAS No. 160). SFAS No. 160 establishes accounting and reporting standards for the noncontrolling interest in a subsidiary and for the deconsolidation of a subsidiary. It clarifies that a noncontrolling interest in a subsidiary is an ownership interest in the consolidated entity that should be reported as equity in the consolidated financial statements. SFAS No. 160 is effective for fiscal years, and interim periods within those fiscal years, beginning on or after December 15, 2008. SFAS No. 160 requires retroactive adoption of the presentation and disclosure requirements for existing minority interests. All other requirements of SFAS No. 160 shall be applied prospectively. We are currently evaluating the potential impact of the adoption of SFAS No. 160 on our consolidated financial statements.

Results of Operations

We commenced our principal operations on December 8, 2006, when we acquired our first interest in a property. On January 12, 2007 we raised the minimum offering of \$1.0 million pursuant to the terms of our Offering Memorandum and issued the initial 40 Units to investors. During 2007, we made investments in three joint ventures through which we obtained an ownership interest in three properties. During the first three months ended March 31, 2008, we acquired an interest in Shadow Creek on February 29, 2008 through a joint venture and acquired a direct interest in Village on the Green on March 25, 2008. As of March 31, 2008, we directly owned one property comprising 36,000 square feet of gross leasable area and owned an investment interest in five other properties comprising 1,231,000 square feet of gross leasable area.

Our direct property acquisition was accounted for as a purchase and the results of its operations are included in our consolidated financial statements from its date of acquisition. We report our investments in joint ventures under the equity method of accounting given our ability to exercise significant influence over them.

Three Months Ended March 31, 2008 versus Three Months Ended March 31, 2007

Revenue. Revenue increased to approximately \$14,000 during the three months ended March 31, 2008 compared to \$0 for the three months ended March 31, 2007. This increase was due to rental income earned from the acquisition of Village on the Green during March 2008.

General & Administrative. General and administrative expenses increased to approximately \$16,000 during the three months ended March 31, 2008 compared to approximately \$0 for the three months ended March 31, 2007. This increase was primarily due to an increase in insurance costs related to the acquisition of Village on the Green.

General & Administrative – Related Party. General and administrative expenses paid to our affiliate increased to approximately \$30,000 during the three months ended March 31, 2008 compared to \$0 for the three months ended March 31, 2007. This increase is due to an increase in administrative cost reimbursements, which is due to higher activity within the fund during the three months ended March 31, 2008 as compared to the three months ended March, 31 2007.

Asset Management Fees. Asset management fees paid to our affiliate increased approximately \$93,000 to approximately \$105,000 during the three months ended March 31, 2008 compared to approximately \$12,000 for the three months ended March 31, 2007. Asset management fees have increased commensurate with the increase in assets under management as a result of our capital-raising efforts which ceased on March 31, 2008.

Legal and Professional Fees. Legal and professional fees increased approximately \$21,000 to approximately \$50,000 during the three months ended March 31, 2008 compared to approximately \$29,000 for the three months ended March 31, 2007. The additional costs are primarily attributable to the increased financial reporting requirements associated with being a public registrant.

Depreciation and Amortization Expense. Depreciation and amortization expense increased to approximately \$12,000 during the three months ended March 31, 2008 compared to approximately \$0 for the three months ended March 31, 2007. This increase was due to the acquisition of Village on the Green during March 2008.

Interest and Other Income. Interest and other income increased approximately \$151,000 to approximately \$154,000 during the three months ended March 31, 2008 compared to approximately \$3,000 for the three months ended March 31, 2007. This increase was primarily due to earning interest on a higher balance of investable funds during the first quarter of 2008 as compared to the first quarter of 2007 as we continued our capital-raising efforts through March 31, 2008. We invest our excess cash in short-term investments or overnight funds until properties suitable for acquisition can be identified and acquired.

Interest Expense. Interest expense increased to approximately \$6,000 during the three months ended March 31, 2008 compared to \$0 for the three months ended March 31, 2007. Interest expense incurred during the three months ended March 31, 2008 is related to the note payable on Village on the Green, which was acquired on March 25, 2008.

Interest Expense – Related Party. Interest expense – related party decreased approximately \$101,000 during the three months ended March 31, 2008 compared to the three months ended March 31, 2007. Interest expense for the three months ended March 31, 2007 was incurred due to a note payable – related party that was established during December 2006 and was not repaid until the second quarter of 2007.

Equity in Losses From Non-Consolidated Entities. Equity in losses from non-consolidated entities increased approximately \$507,000 to approximately \$705,000 for the three months ended March 31, 2008 compared to approximately \$198,000 for the three months ended March 31, 2007. These amounts represent our ownership portion of our joint ventures' net income or loss for the period. The increased loss is primarily attributable to our 40% ownership interest in Woodlake Square, which was acquired in August 2007, and our 40% ownership interest in Westheimer Gessner, which was acquired in November 2007. These two properties have been operating at net losses given their vacancies, and Woodlake Square's loss on derivative.

Liquidity and Capital Resources

We expect to meet our short-term liquidity requirements with cash on hand as well as through net cash provided by property operations. We expect to meet our long-term liquidity requirements through proceeds from secured or unsecured financings from banks and other lenders, the selective and strategic sale of properties and net cash flows from operations.

As of March 31, 2008 and December 31, 2007, our cash and cash equivalents totaled approximately \$13.9 million and approximately \$19.3 million, respectively. Cash flows provided by (used in) operating activities, investing activities and financing activities for the three months ended March 31, 2008 and 2007 are as follows (in thousands):

Three months ended

	Mai	March 31,		
	2008	2007		
Operating activities	\$(1,139	\$(176)	
Investing activities	\$(7,772) \$-		
Financing activities	\$3,465	\$8,346		

Net cash flows used in operating activities increased approximately \$963,000 to approximately \$1.1 million for the three months ended March 31, 2008 compared to approximately \$176,000 for the three months ended March 31, 2007. The increase in operating outflows is primarily attributable to an increase during the period of \$208,000 in receivables, coupled with a decrease in accounts payable of approximately \$471,000 and an increase of approximately \$390,000 in escrow deposits. This was partially offset by an increase of approximately \$98,000 in our income before the effects of losses from non-consolidated entities and depreciation and amortization expenses.

Net cash flows used in investing activities increased to approximately \$7.8 million for the three months ended March 31, 2008 compared to \$0 for the three months ended March 31, 2007. This increase in investing outflows was primarily due to our investment of \$4.8 million in the Shadow Creek Ranch property and our acquisition of the Village on the Green property for \$3.0 million.

Net cash flows provided by financing activities decreased approximately \$4.8 million to approximately \$3.5 million for the three months ended March 31, 2008 compared to approximately \$8.3 million for the three months ended March 31, 2007. This decrease in financing inflows was primarily due to a reduction in contributions received of approximately \$4.4 million coupled with an increase in distributions paid to investors of \$823,000. Additionally, payments related to issuance costs decreased by \$448,000.

Report of Independent Registered Public Accounting Firm

The Partners AmREIT Casa Linda, L.P.:

We have audited the accompanying balance sheets of AmREIT Casa Linda, L.P. (the "Partnership") as of December 31, 2007 and 2006, and the related statements of operations, partners' capital and cash flows for the year ended December 31, 2007 and for the period from December 8, 2006 (inception) through December 31, 2006. These financial statements are the responsibility of the Partnership's management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures on the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of AmREIT Casa Linda, L.P. as of December 31, 2007 and 2006 and the results of its operations and its cash flows for the year ended December 31, 2007 and for the year ended December 31, 2007 and for the period from December 8, 2006 (inception) through December 31, 2006, in conformity with U.S. generally accepted accounting principles.

(signed) KPMG LLP

Houston, Texas

March 31, 2008

AmREIT Casa Linda, L.P. BALANCE SHEETS December 31, 2007 and December 31, 2006 (in thousands)

	December 31,	December 31,
	2007	2006
ASSETS		
Real estate investments at cost:		
Land	\$11,420	\$11,420
Buildings	29,834	29,645
Tenant improvements	530	442
	41,784	41,507
Less accumulated depreciation and amortization	(1,510	(96)
	40,274	41,411
Intangible lease cost, net	3,152	4,403
Net real		
estate		
investments	43,426	45,814
Cash and cash equivalents	250	-
Tenant receivables	519	113
Deferred costs, net	401	428
Other assets	5,655	4,883
TOTAL ASSETS	\$50,251	\$51,238
LIABILITIES AND PARTNERS' CAPITAL		
Liabilities:		
Notes payable	\$37,950	\$37,950
Accounts payable and accrued liabilities	1,558	75
Accounts payable - related party	117	262
Asset retirement obligations	800	800
Below market leases, net	945	1,599
Security deposits	108	120
TOTAL LIABILITIES	\$41,478	\$40,806
Partners' capital:		
General partner	-	-
Limited partner	8,773	10,432
TOTAL PARTNERS' CAPITAL	\$8,773	\$10,432
TOTAL LIABILITIES AND PARTNERS' CAPITAL	\$50,251	\$51,238

See Notes to Financial Statements

AmREIT Casa Linda, L.P. STATEMENTS OF OPERATIONS

For the year ended December 31, 2007 and for the period from December 8, 2006 (inception) through December 31, 2006 (in thousands)

	2007	2006
Revenues:		
Rental income from operating leases	\$5,004	\$391
Total revenues	5,004	391
T.		
Expenses:		
General and administrative	54	-
Property expense	1,670	80
Property management fees - related party	163	8
Legal and professional	113	1
Depreciation and amortization	2,497	221
Total expenses	4,497	310
	507	0.1
Operating income	507	81
Other income (expense):		
Interest and other income	42	5
Interest expense	(2,172) (142
Income tax expense	(36)
Total other income (expense)	(2,166) (137
Net loss	\$(1,650	\$(56
Net ioss	\$(1,659) \$(56

See Notes to Financial Statements.

AmREIT Casa Linda, L.P. STATEMENTS OF PARTNERS' CAPITAL

For the year ended December 31, 2007 and for the period from December 8, 2006 (inception) through December 31, 2006 (in thousands)

	GP		LP	Total
Balance at December 8, 2006 (inception)	\$	-	\$ _	\$ <u>-</u>
Contributions		-	10,488	10,488
Net loss (1)		(1)	(55)	(56)
Curative allocation		1	(1)	-
Balance at December 31, 2006	\$	-	\$ 10,432	\$ 10,432
Net loss (1)		(17)	(1,642)	(1,659)
Curative allocation		17	(17)	-
			, ,	
Balance at December 31, 2007	\$	-	\$ 8,773	\$ 8,773

(1) The allocation of net loss includes a curative allocation to increase the GP capital account by \$17,000 and \$1,000 for the 2007 and 2006 periods, respectively. The partnership agreement provides that no partner shall be required to fund a deficit balance in their capital account.

See Notes to Financial Statements

AmREIT Casa Linda, L.P. STATEMENTS OF CASH FLOWS

For the year eneded December 31, 2007 and for the period from December 8, 2006 (inception) through December 31, 2006 (in thousands)

	2007	2006
Cash flows from operating activities:		
Net loss	\$(1,659) \$(56
Adjustments to reconcile net loss to net cash		
provided by (used in) operating activities:		
Depreciation and amortization	2,081	134
Increase in tenant receivables	(405) (113
Increase in deferred costs	(44) -
Increase in other assets	(772) (4,308
Increase in accounts payable and accrued liabilities	1,483	75
(Decrease) increase in accounts payable - related party	(145) 262
(Decrease) increase in security deposits	(12) 120
Net cash provided by (used in) operating activities	527	(3,886
Cash flows from investing activities:		
Improvements to real estate	(277) -
Acquisition of investment properties	-	(6,172
Net cash used in investing activities	(277) (6,172
Cash flows from financing activities:		
Contributions	-	10,488
Loan acquisition costs	-	(430
Net cash provided by financing activities	-	10,058
Net increase in cash and cash equivalents	250	-
Cash and cash equivalents, beginning of period	-	_
Cash and cash equivalents, end of period	\$250	\$-
Supplemental schedule of cash flow information:	Ф1 0 2 0	Ф120
Cash paid during the year for interest	\$1,929	\$139

Supplemental schedule of non-cash transactions:

In conjunction with our acquistion of the Property in December 2006, we placed \$37,950,000 in debt that was funded directly to the buyer. Additionally, we incurred a liability in the amount of \$800,000 related to an asset retirement obligation. Included in other assets is \$575,000 which was previously placed in an escrow account by the seller and was assiged to us as part of the acquisition.

See Notes to Financial Statements.

AmREIT Casa Linda, L.P. Notes to Financial Statements December 31, 2007 and 2006

1. DESCRIPTION OF BUSINESS

AmREIT Casa Linda, L.P., a Texas limited partnership (the "Partnership"), was formed on December 8, 2006 to acquire Casa Linda, (the "Property") a multi-tenant retail property located in Dallas, Texas with a combined gross leasable area of 324,638 square feet. The property is in the planning stages of a significant redevelopment project that is expected to be completed during 2008. The General Partner of the Partnership is AmREIT Casa Linda GP, Inc. The General Partner maintains its principal place of business in Houston, Texas.

The limited partners of the Partnership are AmREIT Monthly Income & Growth Fund III, Ltd. (MIG III) and AmREIT Monthly Income & Growth Fund IV, Ltd. (MIG IV). We have contracted with AmREIT Realty Investment Corporation ("ARIC") to provide property management and leasing services. ARIC is a wholly-owned subsidiary of AmREIT (AMEX: AMY), a Texas real estate investment trust, and is an affiliate of MIG III and MIG IV.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

BASIS OF PRESENTATION

Our financial records are maintained on the accrual basis of accounting whereby revenues are recognized when earned and expenses are recorded when incurred. We were formed on December 8, 2006. Accordingly, the accompanying statements of operations, changes in partners' capital and cash flows related to 2006 represent activity for the period from December 8, 2006 (inception) through December 31, 2006. Unless otherwise noted, all references to the "2006 period" or "the period ended December 31, 2006" refer to the short period of operations from December 8, 2006 through December 31, 2006.

REVENUE RECOGNITION

We lease space to tenants under agreements with varying terms. Our leases are accounted for as operating leases with revenue being recognized on a straight-line basis over the terms of the individual leases. Accrued rents are included in tenant receivables. Revenue from tenant reimbursements of taxes, maintenance expenses and insurance is recognized in the period the related expense is recorded. Additionally, certain of the lease agreements contain provisions that provide for additional rents based on tenants' sales volumes (contingent or percentage rent). Percentage rents are recognized when the tenants achieve the specified targets as defined in their lease agreements. During the periods ended December 31, 2007 and 2006, we recognized percentage rents of \$26,000 and \$0, respectively. We recognize lease termination fees in the period that the lease is terminated and collection of the fees is reasonably assured. During the periods ended December 31, 2007 and 2006, we recognized no lease termination fees.

REAL ESTATE INVESTMENTS

Acquired Properties and Acquired Lease Intangibles — We account for real estate acquisitions pursuant to Statement of Financial Accounting Standards No. 141, Business Combinations ("SFAS No. 141"). Accordingly, we allocate the purchase price of the acquired properties to land, building and improvements, identifiable intangible assets and to the acquired liabilities based on their respective fair values. Identifiable intangibles include amounts allocated to acquired out-of-market leases, the value of in-place leases and customer relationship value, if any. We determine fair value based on estimated cash flow projections that utilize appropriate discount and capitalization rates and available market information. Estimates of future cash flows are based on a number of factors including the historical operating results, known trends and specific market and economic conditions that may affect the property. Factors considered by management in our analysis of determining the as-if-vacant property value include an estimate of carrying costs during the expected lease-up periods considering market conditions, and costs to execute similar leases. In estimating carrying costs, management includes real estate taxes, insurance and estimates of lost rentals at market rates during the expected lease-up periods, tenant demand and other economic conditions. Management also estimates costs to execute similar leases including leasing commissions, tenant improvements, legal and other related expenses. Intangibles related to out-of-market leases and in-place lease value are recorded as acquired lease intangibles and are amortized as an adjustment to rental revenue or amortization expense, as appropriate, over the remaining terms of the underlying leases.

<u>Depreciation</u> - Depreciation is computed using the straight-line method over an estimated useful life of up to 30 years for buildings, up to 6 years for site improvements and over the term of lease for tenant improvements.

<u>Impairment</u> - We review the property for impairment whenever events or changes in circumstances indicate that the carrying amount of the assets, including accrued rental income, may not be recoverable through operations. We determine whether an impairment in value occurred by comparing the estimated future cash flows (undiscounted and without interest charges), including the residual value of the property, with the carrying value of the individual property. If impairment is indicated, a loss will be recorded for the amount by which the carrying value of the asset exceeds its fair value. No impairment charges were recognized during the 2007 and 2006 periods.

ENVIRONMENTAL EXPOSURES

We are subject to numerous environmental laws and regulations as they apply to real estate pertaining to chemicals used by the dry cleaning industry, the existence of asbestos in older shopping centers, and underground petroleum storage tanks. We believe that the tenants who currently operate dry cleaning plants or gas stations do so in accordance with current laws and regulations. We believe that the ultimate disposition of currently known environmental matters will not have a material affect on our financial position, liquidity, or operations. However, we can give no assurance that existing environmental studies with respect to the shopping centers have revealed all potential environmental liabilities; that any previous owner, occupant or tenant did not create any material environmental condition not known to it; that the current environmental condition of the shopping center will not be affected by tenants and occupants, by the condition of nearby properties, or by unrelated third parties; or that changes in applicable environmental laws and regulations or their interpretation will not result in additional environmental liability to the Partnership.

RECEIVABLES AND ALLOWANCE FOR UNCOLLECTIBLE ACCOUNTS

<u>Tenant Receivables</u> - Included in tenant receivables are base rents, tenant reimbursements and receivables attributable to recording rents on a straight-line basis. An allowance for the uncollectible portion of accrued rents and accounts receivable is determined based upon customer credit-worthiness (including expected recovery of our claim with respect to any tenants in bankruptcy), historical bad debt levels, and current economic trends. Bad debt expenses and any related recoveries are included in general and administrative expense. As of December 31, 2007 and 2006, we did not have an allowance for uncollectible accounts related to our tenant receivables.

DEFERRED COSTS

Deferred costs include deferred leasing costs and loan acquisition costs, net of amortization. Loan acquisition costs are incurred in obtaining financing and are amortized to interest expense over the term of the debt agreements. Deferred leasing costs consist of external commissions associated with leasing our properties and are amortized to expense over the lease term. Accumulated amortization related to loan acquisition costs as of December 31, 2007 and 2006 totaled \$67,000 and \$3,000, respectively. Accumulated amortization related to leasing costs as of December 31, 2007 and 2006 totaled \$6,000 and \$0, respectively.

INCOME TAXES

<u>Federal</u> - No provision for U.S. federal income taxes is included in the accompanying financial statements. As a partnership, we are not subject to federal income tax, and the federal tax effect of our activities is passed through to our partners.

<u>State</u> - In May 2006, the State of Texas adopted House Bill 3, which modified the state's franchise tax structure, replacing the previous tax based on capital or earned surplus with one based on margin (often referred to as the "Texas Margin Tax") effective with franchise tax reports filed on or after January 1, 2008. The Texas Margin Tax is computed by applying the applicable tax rate (1% for the Partnership) to the profit margin. The Texas Margin Tax accrued as of December 31, 2007 and 2006 totaled \$36,000 and \$0, respectively.

USE OF ESTIMATES

The preparation of financial statements in conformity with U.S. generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

FAIR VALUE OF FINANCIAL INSTRUMENTS

Our financial instruments consist primarily of cash and cash equivalents, tenant receivables, note payable, accounts payable and accrued liabilities, and accounts payable-related party. The carrying value of cash, cash equivalents, tenant receivables, accounts payable and accrued liabilities, and accounts payable-related party are representative of their respective fair values due to the short-term nature of these instruments. As of December 31, 2007 and 2006, the carrying value of our total debt obligations was \$38.0 and \$38.0 million, respectively; all of which represents fixed-rate obligations with an estimated fair value of \$38.2 and \$36.7 million, respectively, based on a discounted cash flow analysis using current market rates of interest.

NEWACCOUNTING STANDARDS

In September 2006, The Financial Accounting Standards Board ("FASB") issued SFAS No. 157, *Fair Value Measurement* (SFAS No. 157). SFAS No. 157 defines fair value, establishes a framework for measuring fair value and requires enhanced disclosures about fair value measurements. SFAS No. 157 requires companies to disclose the fair value of its financial instruments according to a fair value hierarchy. Additionally, companies are required to provide certain disclosures regarding instruments within the hierarchy, including a reconciliation of the beginning and ending balances for each major category of assets and liabilities. SFAS No. 157 is effective for our fiscal year beginning January 1, 2008. The adoption of SFAS No. 157 is not expected to have a material effect on our results of operations or financial position.

In February 2007 the FASB issued SFAS No. 159, *The Fair Value Option for Financial Assets and Financial Liabilities* (SFAS No. 159). SFAS No. 159 expands opportunities to use fair value measurement in financial reporting and permits entities to choose to measure many financial instruments and certain other items at fair value. This Statement is effective for fiscal years beginning after November 15, 2007. We currently do not plan to measure any eligible financial assets and liabilities at fair value under the provisions of SFAS No. 159.

In December 2007, the FASB issued SFAS No. 141 (Revised 2007), *Business Combinations* (SFAS No. 141R). SFAS No. 141R will change the accounting for business combinations. Under SFAS No. 141R, an acquiring entity will be required to recognize all the assets acquired and liabilities assumed in a transaction at the acquisition-date fair value with limited exceptions. SFAS No. 141R applies prospectively to business combinations for which the acquisition date is on or after the beginning of the first annual reporting period beginning on or after December 15, 2008. We are currently evaluating the potential impact of SFAS No. 141R on our financial position and results of operations beginning for fiscal year 2009.

In December 2007, the FASB issued SFAS No. 160, *Noncontrolling Interests in Consolidated Financial Statements—an amendment of ARB No. 51* (SFAS No. 160). SFAS No. 160 establishes accounting and reporting standards for the noncontrolling interest in a subsidiary and for the deconsolidation of a subsidiary. It clarifies that a noncontrolling interest in a subsidiary is an ownership interest in the consolidated entity that should be reported as equity in the consolidated financial statements. SFAS No. 160 is effective for fiscal years, and interim periods within those fiscal years, beginning on or after December 15, 2008. SFAS No. 160 requires retroactive adoption of the presentation and disclosure requirements for existing minority interests. All other requirements of SFAS No. 160 shall be applied prospectively. We are currently evaluating the potential impact of the adoption of SFAS No. 160 on our consolidated financial statements.

CASH AND CASH EQUIVALENTS

For purposes of the statements of cash flows, we consider all highly liquid debt instruments purchased with an original maturity of three months or less to be cash equivalents. Cash and cash equivalents consist of demand deposits at commercial banks and money market funds.

INTEREST

Interest is charged to interest expense as it accrues. No interest was capitalized during the 2007 or 2006 period.

3. OPERATING LEASES

Our operating leases generally range from one month to thirteen years and generally include one or more five year renewal options. A summary of minimum future base rentals to be received, exclusive of any renewals, under non-cancelable operating leases in existence at December 31, 2007 is as follows (in thousands):

2008	\$3,077
2009	2,119
2010	1,522
2011	965
2012	638
Thereafter	4,374
	\$12,695

Future minimum rental revenue excludes amounts that may be received from tenants for reimbursements of operating costs, real estate taxes and insurance. Expense reimbursements recognized as revenue totaled \$1.3 million and \$0.1 million during the 2007 and 2006 periods, respectively.

4. ACQUIRED LEASE INTANGIBLES

In accordance with SFAS No. 141, we have identified and recorded the value of intangibles at the property acquisition date. Such intangibles include the value of in-place leases and out-of-market leases. Acquired lease intangible assets (in-place leases and above-market leases) are amortized over the leases' remaining terms, which range from 1 month to 13 years. The amortization of above-market leases is recorded as a reduction of rental income and the amortization of in-place leases is recorded to amortization expense. The amortization expense related to in-place leases was \$1,077,000 and \$81,000 during 2007 and 2006, respectively. The amortization of above-market leases, which was recorded as a reduction of rental income, was \$173,000 and \$14,000 during 2007 and 2006, respectively.

In-place and above-market lease amounts and their respective accumulated amortization are as follows (in thousands):

		December 31, 2007		December		31, 2006		
		In-Place		Above-market	'		Above-ma	ırket
		leases		leases	In-Place	leases	leases	
Cos	st	\$2,757		\$1,718	\$2,776		\$1,722	
Ac	cumulated amortization	(1,140	_)	(183	(81)	(14)
	Intangible lease cost, net	\$1,617		\$1,535	\$2,695		\$1,708	

Acquired lease intangible liabilities (below-market leases) are net of previously accreted minimum rent of \$702,000 and \$58,000 at December 31, 2007 and 2006, respectively and are accreted over the leases' remaining terms, which range from 1 month to 13 years. Accretion of below market leases was approximately \$653,000 and \$58,000 during the 2007 and 2006 periods, respectively. Such accretion is recorded as an increase to rental income.

The estimated aggregate amortization amounts from acquired lease intangibles for each of the next five years are as follows (in thousands):

Year Ending December 31,	Amortization Expense (inplace lease value)	Rental Income (out- of-market leases)
2008	\$695	\$(357)
2009	376	(112)
2010	199	58
2011	83	122
2012	45	134
	\$1,398	\$(155)

5. NOTES PAYABLE

Our outstanding debt at December 31, 2007 and 2006 consisted entirely of a \$38.0 million fixed-rate mortgage loan, which requires monthly interest-only payments and matures in January 2014. Our mortgage loan is secured by certain real estate properties and may be prepaid, but is subject to a yield-maintenance premium or prepayment penalty.

As of December 31, 2007, the interest rate on our fixed-rate debt is 5.5%, and the remaining life is 6 years. As of December 31, 2007, scheduled principal repayments on our note payable were as follows (in thousands):

Scheduled Payments by Year	Scheduled Principal Payments	Term-Loan Maturities	Total Payments
2008	\$-	\$ -	\$-
2009	-	-	-
2010	-	-	-
2011	-	-	-
2012	-	-	-
Thereafter	-	37,950	37,950
Total	\$-	\$37,950	\$37,950

6. CONCENTRATIONS

As of December 31, 2007, Casa Linda consisted of 63 tenants; none of which comprised more than 10% of our total base rent. Following are the rental income generated by our top tenants during the period ended December 31 (\$ in thousands):

	2007	2006
Albertson's	\$287	\$19
24 Hour Fitness	229	23
El Fenix Restaurant	181	12
PETCO	178	11
Chili's Restaurant	144	12
	\$1,019	\$77

7. RELATED PARTY TRANSACTIONS

We have no employees or offices. We rely on our General Partner to manage our business and affairs. Our General Partner utilizes the services of AmREIT and its affiliates in performing its duties to us. These services primarily include the supervision of the management and leasing of the Property. As a result, we are dependent upon AmREIT and its affiliates. In the event that these companies are unable to provide us with the respective services, we would be required to find alternative providers of these services. During 2007 and 2006 we incurred property management and leasing fees of \$207,000 and \$8,000, respectively.

8. COMMITMENTS AND CONTINGENCIES

<u>Litigation</u> - In the ordinary course of business, we may become subject to litigation or claims. There are no material pending legal proceedings known to be contemplated against us.

<u>Environmental matters</u> - In connection with the ownership and operation of real estate, we may be potentially liable for costs and damages related to environmental matters. We have not been notified by any governmental authority of any non-compliance, liability or other claim.

In conjunction with our acquisition of Casa Linda in December 2006, we identified an environmental exposure caused by a dry cleaning business that operated on the property prior to our ownership. We estimate that our obligation related to remediation of this exposure will be approximately \$800 thousand. Such amount has been placed in escrow and represents the deductible amount under an environmental insurance policy that we placed on the property upon acquisition. We have recorded an asset retirement obligation for \$800 thousand and

~S-8		

Report of Independent Registered Public Accounting Firm

The Partners
AmREIT Woodlake Square, L.P.:

We have audited the accompanying balance sheet of AmREIT Woodlake Square, L.P. (the "Partnership") as of December 31, 2007, and the related statements of operations, partners' capital and cash flows for the period from August 31, 2007 (inception) through December 31, 2007. These financial statements are the responsibility of the Partnership's management. Our responsibility is to express an opinion on these financial statements based on our audit.

We conducted our audit in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audit provides a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of AmREIT Woodlake Square, L.P. as of December 31, 2007, and the results of its operations and its cash flows for the period from August 31, 2007 (inception) through December 31, 2007, in conformity with U.S. generally accepted accounting principles.

(signed) KPMG LLP

Houston, Texas

April 29, 2008

Schedule IV

AmREIT Woodlake Square, L.P. BALANCE SHEET December 31, 2007 (in thousands)

	December 31, 2007
ASSETS	
Real estate investments at cost:	
Land	\$18,949
Buildings	18,096
Tenant improvements	824
	37,869
Less accumulated depreciation and amortization	(347
	37,522
Intangible lease cost, net	2,670
Net real estate investments	40,192
	,
Cash and cash equivalents	488
Tenant receivables, net	131
Accounts receivable - related party	767
Deferred costs, net	238
Other assets	433
TOTAL ASSETS	\$42,249
LIABILITIES AND PARTNERS' CAPITAL	
Liabilities:	#22.000
Note payable	\$23,800 947
Accounts payable and accrued liabilities Accounts payable - related party	1,130
Asset Retirement Obligations	700
Below market leases, net	2,338
Security deposits	84
TOTAL LIABILITIES	\$28,999
Posturand conitalis	
Partners' capital:	
General partner Limited Partner	12 250
	13,250
TOTAL PARTNERS' CAPITAL	13,250
TOTAL LIABILITIES AND PARTNERS' CAPITAL	\$42,249

See Notes to Financial Statements

AmREIT Woodlake Square, L.P. STATEMENT OF OPERATIONS For the period from August 31, 2007 through December 31, 2007 (in thousands)

	2007
Revenues:	
Rental income from operating leases	\$ 1,088
Total revenues	1,088
Expenses:	
General and administrative	8
Property expense	323
Property management fees - related	37
party	
Legal and professional	72
Depreciation and amortization	788
Total expenses	1,228
Operating (loss)	(140)
Other income (expense):	
Interest and other income	12
Interest expense	(565)
Income Tax Expense	(8)
Total other income (expense)	(561)
Net Loss	\$ (701)

See Notes to Financial Statements

AmREIT Woodlake Square, L.P. STATEMENT OF PARTNERS' CAPITAL For the period from August 31, 2007 (inception) to December 31, 2007 (in thousands)

	GP	LP	Total
Balance at August 31, 2007 (inception)	<u>\$-</u>	\$-	\$-
Contributions	-	13,951	13,951
Net loss	(7) (694) (701)
Curative allocation (1)	7	(7) -
Balance at December 31, 2007	<u>\$-</u>	\$13,250	\$13,250

(1) The allocation of net loss includes a curative allocation to increase the GP capital account by \$7,000 for the 2007 period. The partnership agreement provides that no partner shall be required to fund a deficit balance in the capital account.

See Notes to Financial Statements.

AmREIT Woodlake Square, L.P. STATEMENT OF CASH FLOWS

For the period from August 31, 2007 (inception) to December 31, 2007 (in thousands)

	2007
Cash flows from operating activities:	
Net loss	\$(701
Adjustments to reconcile net loss to net cash	
provided by operating activities:	
Depreciation and amortization	592
Increase in tenant receivables	(131
Increase in accounts receivable - related party	(767
Increase in intangible lease costs and other assets	(33
Increase in accounts payable	1,247
Increase in accounts payable - related party	1,130
Increase in below market leases	(2
Increase in security deposits	84
Net cash provided by operating activities	1,420
Cook flows from investing activities.	
Cash flows from investing activities:	(14.616
Acquisition of investment properties	(14,616
Net cash used in investing activities	(14,616
Cash flows from financing activities:	
Contributions	13,951
Loan acquisition costs	(267
Net cash provided by financing activities	13,684
Net increase in cash and cash equivalents	488
Cash and cash equivalents, beginning of period	-
Cash and cash equivalents, end of period	\$488
	+ 133
Supplemental schedule of cash flow information:	
Cash paid during the year for interest	\$400
- man same we fear for moreon	Ψ.00

Supplemental schedule of cash flow information

In conjunction with our acquisition of the Property in August 2007, we placed \$23,800,000 in debt that was funded directly to the buyer.

See Notes to Financial Statements

AmREIT Woodlake Square, L.P. Notes to Financial Statements December 31, 2007

1. DESCRIPTION OF BUSINESS

AmREIT Woodlake Square, L.P., a Texas limited partnership (the "Partnership"), was formed on August 31, 2007 to acquire Woodlake Square (the "Property"), a multi-tenant retail property located in Houston, Texas with a combined gross leasable area of 206,000 square feet. The property is in the planning stages of a significant redevelopment project that is expected to be complete in 2010. The General Partner of the Partnership is AmREIT Woodlake GP, LLC. The General Partner maintains its principal place of business in Houston, Texas.

The limited partners of the Partnership are AmREIT Monthly Income & Growth Fund III, Ltd. (MIG III), AmREIT Monthly Income & Growth Fund IV, Ltd. (MIG IV), and AmREIT Realty Investment Corporation (ARIC). We have contracted with ARIC to provide property management and leasing services. ARIC is a wholly-owned subsidiary of AmREIT (AMEX: AMY), a Texas real estate investment trust, and is an affiliate of MIG III and MIG IV.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

BASIS OF PRESENTATION

Our financial records are maintained on the accrual basis of accounting whereby revenues are recognized when earned and expenses are recorded when incurred. We were formed on August 31, 2007. Accordingly, the accompanying statements of operations, changes in partners' capital and cash flows related to 2007 represent activity for the period from August 31, 2007 (inception) through December 31, 2007. Unless otherwise noted, all references to the "2007 period" or "the period ended December 31, 2007" refer to the short period of operations from August 31, 2007 through December 31, 2007.

REVENUE RECOGNITION

We lease space to tenants under agreements with varying terms. Our leases are accounted for as operating leases with revenue being recognized on a straight-line basis over the terms of the individual leases. Accrued rents are included in tenant receivables. Revenue from tenant reimbursements of taxes, maintenance expenses and insurance is recognized in the period the related expense is recorded. Additionally, certain of the lease agreements contain provisions that provide for additional rents based on tenants' sales volumes (contingent or percentage rent). Percentage rents are recognized when the tenants achieve the specified targets as defined in their lease agreements. During the period ended December 31, 2007 there were no percentage rents recognized. We recognize lease termination fees in the period that the lease is terminated and collection of the fees is reasonably assured. During the period ended December 31, 2007, we recognized no lease termination fees.

REAL ESTATE INVESTMENTS

Acquired Lease Intangibles — We account for real estate acquisitions pursuant to Statement of Financial Accounting Standards No. 141, Business Combinations ("SFAS No. 141"). Accordingly, we allocate the purchase price of the acquired properties to land, building and improvements, identifiable intangible assets and to the acquired liabilities based on their respective fair values. Identifiable intangibles include amounts allocated to acquired out-of-market leases, the value of in-place leases and customer relationship value, if any. We determine fair value based on estimated cash flow projections that utilize appropriate discount and capitalization rates and available market information. Estimates of future cash flows are based on a number of factors including the historical operating results, known trends and specific market and economic conditions that may affect the property. Factors considered by management in our analysis of determining the as-if-vacant property value include an estimate of carrying costs during the expected lease-up periods considering market conditions, and costs to execute similar leases. In estimating carrying costs, management includes real estate taxes, insurance and estimates of lost rentals at market rates during the expected lease-up periods, tenant demand and other economic conditions. Management also estimates costs to execute similar leases including leasing commissions, tenant improvements, legal and other related expenses. Intangibles related to out-of-market leases and in-place lease value are recorded as acquired lease intangibles and are amortized as an adjustment to rental revenue or amortization expense, as appropriate, over the remaining terms of the underlying leases.

<u>Depreciation</u> - Depreciation is computed using the straight-line method over an estimated useful life of up to 28 years for buildings, up to 4 years for site improvements and over the term of lease for tenant improvements.

Impairment - We review the property for impairment whenever events or changes in circumstances indicate that the carrying amount of the assets, including accrued rental income, may not be recoverable through operations. We determine whether an impairment in value occurred by comparing the estimated future cash flows (undiscounted and without interest charges), including the residual value of the property, with the carrying value of the individual property. If impairment is indicated, a loss will be recorded for the amount by which the carrying value of the asset exceeds its fair value. No impairment charges were recorded during the 2007 period.

ENVIRONMENTAL EXPOSURES

We are subject to numerous environmental laws and regulations as they apply to real estate pertaining to chemicals used by the dry cleaning industry, the existence of asbestos in older shopping centers, and underground petroleum storage tanks. We believe that the tenants who currently operate dry cleaning plants or gas stations do so in accordance with current laws and regulations. We believe that the ultimate disposition of currently known environmental matters will not have a material affect on our financial position, liquidity, or operations. However, we can give no assurance that existing environmental studies with respect to the shopping center have revealed all potential environmental liabilities; that any previous owner, occupant or tenant did not create any material environmental condition not known to it; that the current environmental condition of the shopping center will not be affected by tenants and occupants, by the condition of nearby properties, or by unrelated third parties; or that changes in applicable environmental laws and regulations or their interpretation will not result in additional environmental liability to the Partnership.

RECEIVABLES AND ALLOWANCE FOR UNCOLLECTIBLE ACCOUNTS

<u>Tenant Receivables</u> - Included in tenant receivables are base rents, tenant reimbursements and receivables attributable to recording rents on a straight-line basis. An allowance for the uncollectible portion of accrued rents and accounts receivable is determined based upon customer credit-worthiness (including expected recovery of our claim with respect to any tenants in bankruptcy), historical bad debt levels, and current economic trends. Bad debt expenses and any related recoveries are included in general and administrative expense. As of December 31, 2007, we did not have an allowance for uncollectible accounts related to our tenant receivables.

<u>Accounts Receivable-Related Party</u> - Accounts receivable-related party represents amounts due from affiliates for their portion of earnest money on the acquisition of the property.

DEFERRED COSTS

Deferred costs include deferred leasing costs and loan acquisition costs, net of amortization. Loan acquisition costs are incurred in obtaining financing and are amortized to interest expense over the term of the debt agreements. Deferred leasing costs consist of external commissions associated with leasing our properties and are amortized to expense over the lease term. Accumulated amortization related to loan acquisition costs as of December 31, 2007 totaled \$30,000. Accumulated amortization related to leasing costs as of December 31, 2007, totaled \$0.

INCOME TAXES

<u>Federal</u> - No provision for U.S. federal income taxes is included in the accompanying financial statements. As a partnership, we are not subject to federal income tax, and the federal tax effect of our activities is passed through to our partners.

<u>State</u> - In May 2006, the State of Texas adopted House Bill 3, which modified the state's franchise tax structure, replacing the previous tax based on capital or earned surplus with one based on margin (often referred to as the "Texas Margin Tax") effective with franchise tax reports filed on or after January 1, 2008. The Texas Margin Tax is computed by applying the applicable tax rate (1% for the Partnership) to the profit margin. At December 31, 2007, we had accrued State Taxes related to the Margin Tax of \$7,600.

USE OF ESTIMATES

The preparation of financial statements in conformity with U.S. generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

FAIR VALUE OF FINANCIAL INSTRUMENTS

Our financial instruments consist primarily of cash and cash equivalents, tenant receivables, accounts receivable-related party, note payable, accounts payable, and accounts payable-related party. The carrying value of cash, cash equivalents, tenant receivables, accounts receivable-related party, accounts payable, and accounts payable-related party are representative of their respective fair values due to the short-term maturity of these instruments. As of December 31, 2007, the carrying value of our total debt obligations was \$23.8 million, all of which represents variable-rate obligations. At December 31, 2007, the fair value of our debt approximated the carrying value.

NEWACCOUNTING STANDARDS

In September 2006, The Financial Accounting Standards Board ("FASB") issued SFAS No. 157, Fair Value Measurement (SFAS No. 157). SFAS No. 157 defines fair value, establishes a framework for measuring fair value and requires enhanced disclosures about fair value measurements. SFAS No. 157 requires companies to disclose the fair value of its financial instruments according to a fair value hierarchy. Additionally, companies are required to provide certain disclosures regarding instruments within the hierarchy, including a reconciliation of the beginning and ending balances for each major category of assets and liabilities. SFAS No. 157 is effective for our fiscal year beginning January 1, 2008. The adoption of SFAS No. 157 is not expected to have a material effect on our results of operations or financial position.

In February 2007 the FASB issued SFAS No. 159, The Fair Value Option for Financial Assets and Financial Liabilities (SFAS No. 159). SFAS No. 159 expands opportunities to use fair value measurement in financial reporting and permits entities to choose to measure many financial instruments and certain other items at fair value. This Statement is effective for fiscal years beginning after November 15, 2007. We currently do not plan to measure any eligible financial assets and liabilities at fair value under the provisions of SFAS No. 159.

In December 2007, the FASB issued SFAS No. 141 (Revised 2007), Business Combinations (SFAS No. 141R). SFAS No. 141R will change the accounting for business combinations. Under SFAS No. 141R, an acquiring entity will be required to recognize all the assets acquired and liabilities assumed in a transaction at the acquisition-date fair value with limited exceptions. SFAS No. 141R applies prospectively to business combinations for which the acquisition date is on or after the beginning of the first annual reporting period beginning on or after December 15, 2008. We are currently evaluating the potential impact of SFAS No. 141R on our financial position and results of operations beginning for fiscal year 2009.

In December 2007, the FASB issued SFAS No. 160, Noncontrolling Interests in Consolidated Financial Statements—an amendment of ARB No. 51 (SFAS No. 160). SFAS No. 160 establishes accounting and reporting standards for the noncontrolling interest in a subsidiary and for the deconsolidation of a subsidiary. It clarifies that a noncontrolling interest in a subsidiary is an ownership interest in the consolidated entity that should be reported as equity in the consolidated financial statements. SFAS No. 160 is effective for fiscal years, and interim periods within those fiscal years, beginning on or after December 15, 2008. SFAS No. 160 requires retroactive adoption of the presentation and disclosure requirements for existing minority interests. All other requirements of SFAS No. 160 shall be applied prospectively. We are currently evaluating the potential impact of the adoption of SFAS No. 160 on our consolidated financial statements.

CASH AND CASH EQUIVALENTS

For purposes of the statements of cash flows, we consider all highly liquid debt instruments purchased with an original maturity of three months or less to be cash equivalents. Cash and cash equivalents consist of demand deposits at commercial banks and money market funds.

INTEREST

Interest is charged to interest expense as it accrues. No interest was capitalized during the 2007 period.

3. OPERATING LEASES

Our operating leases range from three months to thirteen years and generally include one or more five year renewal options. A summary of minimum future base rentals to be received, exclusive of any renewals, under non-cancelable operating leases in existence at December 31, 2007 is as follows (in thousands):

Thereafter	16,368
2012	837
2011	995
2010	1,104
2009	1,332
2008	\$1,783

Future minimum rental revenue excludes amounts that may be received from tenants for reimbursements of operating costs, real estate taxes and insurance. Expense reimbursements totaled \$296,000 during the 2007 period.

4. ACQUIRED LEASE INTANGIBLES

In accordance with SFAS No. 141, we have identified and recorded the value of intangibles at the property acquisition date. Such intangibles include the value of in-place leases and out-of-market leases. Acquired lease intangible assets (in-place leases and above-market leases) are amortized over the leases' remaining terms, which range from 3 months to 13 years. The amortization of above-market leases is recorded as a reduction of rental income and the amortization of in-place leases is recorded to amortization expense. The amortization expense related to in-place leases was approximately \$423,000 for the 2007 period. The amortization of above-market leases, which was recorded as a reduction of rental income, was approximately \$17,000 during the 2007 period.

In-place and above-market lease amounts and their respective accumulated amortization are as follows (in thousands):

	Decemb	er 31, 2007
	In-Place	Above-
	leases	market leases
Cost	\$2,853	\$178
Accumulated amortization	(344) (17
Intangible lease cost, net	\$2,509	\$161

Acquired lease intangible liabilities (below-market leases) are net of previously accreted minimum rent of approximately \$146,000 at December 31, 2007 and are accreted over the leases' remaining terms, which range from 3 months to 13 years. Accretion of below-market leases was approximately \$146,000 during the 2007 period. Such accretion is recorded as an increase to rental income.

The estimated aggregate amortization amounts from acquired lease intangibles for each of the next five years are as follows (in thousands):

Year Ending December 31,	Amortization Expense (inplace lease value)	Rental Income (out- of-market leases)
2008	\$820	\$384
2009	448	360
2010	335	325
2011	282	314
2012	247	310
	\$2,132	\$1,693

5. NOTES PAYABLE

Our outstanding debt at December 31, 2007 consists entirely of a \$23.8 million variable rate mortgage loan which requires monthly interest-only payments and matures in 2010 and is secured by the property. Interest-only loans are generally due in full at maturity. The interest related to this debt is at a variable interest rate of LIBOR plus 1.35%. At December 31, 2007, the variable interest rate was 6.575%.

6. CONCENTRATIONS

As of December 31, 2007, Woodlake Square consisted of 33 tenants. At December 31, 2007, two tenants comprised more than 10% of our total base rent- Walgreens (17%) and, Randall's (12%). Following are the base rents generated by our top tenants during the period ended December 31, 2007 (\$ in thousands):

	2007
Walgreens	\$112
Randall's	81
Jos A. Bank	70
Hooters	39
Amazon Grill	37
	\$339

7. RELATED PARTY TRANSACTIONS

We have no employees or offices. We rely on our General Partner to manage our business and affairs. Our General Partner utilizes the services of AmREIT and its affiliates in performing its duties to us. These services primarily include the supervision of the management and leasing of the Property. As a result, we are dependent upon AmREIT and its affiliates. In the event that these companies are unable to provide us with the respective services, we would be required to find alternative providers of these services. During 2007, we incurred property management and leasing fees of \$37,000. Additionally, we incurred acquisition fees in the amount of \$1.7 million as part the acquisition of the Property which we paid to an affiliate. Such acquisition fees were included in the purchase price of the Property and have therefore been recorded to the various tangible and intangible assets as part of the purchase price allocation.

8. COMMITMENTS AND CONTINGENCIES

<u>Litigation</u> - In the ordinary course of business, we may become subject to litigation or claims. There are no material pending legal proceedings known to be contemplated against us.

<u>Environmental matters</u> - In connection with the ownership and operation of real estate, we may be potentially liable for costs and damages related to environmental matters. We have not been notified by any governmental authority of any non-compliance, liability or other claim.

In conjunction with our acquisition of Woodlake Square in August 2007, we identified an environmental exposure caused by a dry cleaning business that operated on the property prior to our ownership. We estimate that our obligation related to remediation of this exposure will be approximately \$300 thousand. We received a credit at closing from the seller for \$300 thousand, which we have recorded as an asset retirement obligation. Additionally, there is a monitoring well on site that has shown a heightened reading of hazardous material believed to an anomaly. We estimate that our obligation related to remediation of this exposure will be approximately \$400 thousand, and such amount was placed in escrow by the seller. Accordingly, we have recorded an asset retirement obligation for the \$400 thousand, and have included the escrow amount in Other Assets in our accompanying balance sheet. We believe that this matter will not have an adverse effect on our financial position or results of operations, and we are aware of no other environmental exposures

Index to Financial Statements

The Shareholders

AmREIT Monthly Income & Growth IV Corporation:

We have audited the accompanying balance sheet of AmREIT Monthly Income & Growth IV Corporation (the "Company") as of December 31, 2007. This financial statement is the responsibility of the Company's management. Our responsibility is to express an opinion on this financial statement based on our audit.

We conducted our audit in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the balance sheet is free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the balance sheet, assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall balance sheet presentation. We believe that our audit of the balance sheet provides a reasonable basis for our opinion.

In our opinion, the balance sheet referred to above presents fairly, in all material respects, the financial position of AmREIT Monthly Income & Growth IV Corporation as of December 31, 2007 in conformity with U.S. generally accepted accounting principles.

(signed) KPMG LLP Houston, Texas August 29, 2008

AmREIT Monthly Income & Growth IV Corporation Balance Sheet December 31, 2007

	December 31,
	2007
ASSETS	
Accounts receivable - related party	11,125
TOTAL ASSETS	\$11,125
LIABILITIES AND PARTNERS' CAPITAL	
Liabilities:	
Excess of distributions and losses over investment in Monthly Income & Growth	
IV Fund	25,627
TOTAL LIABILITIES	25,627
Shareholder's Equity:	
Common Stock	5
Paid in Capital	1,995
Retained Earnings	(16,502
TOTAL SHAREHOLDER'S EQUITY	(14,502
TOTAL LIABILITIES AND SHAREHOLDER'S EQUITY	\$11,125

See Notes to Balance Sheet.

AmREIT Monthly Income & Growth IV Corporation Notes to Balance Sheet December 31, 2007

1. DESCRIPTION OF BUSINESS

AmREIT Monthly Income & Growth IV Corporation, a Delaware Corporation (the "Company"), was formed in 2006 for the sole purpose of serving as the General Partner of Monthly Income & Growth IV Fund, L.P. ("MIG IV"). We are wholly-owned by AmREIT, a Texas real estate investment trust which is traded on the American Stock Exchange under the symbol "AMY" We maintain our principal place of business in Houston, Texas. The Company invested \$1,000 as a general partner in MIG IV.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

BASIS OF PRESENTATION

Our financial records are maintained on the accrual basis of accounting whereby revenues are recognized when earned and expenses are recorded when incurred.

<u>Accounts Receivable-Related Party</u> - Accounts receivable-related party represents amounts on loan to our parent company. As part of AmREIT's corporate treasury management function, any excess cash is automatically transferred to AmREIT for investment or other uses at the corporate level.

Income Taxes -

<u>Federal</u> - We are treated for Federal income tax purposes as a taxable REIT subsidiary. We believe that it is not more likely than not that we will realize the tax benefits associated with our net loss for the year ended December 31, 2007. Accordingly, we have fully reserved for any such benefit as of December 31, 2007. Further, we have elected to be taxed as a qualified REIT subsidiary effective January 1, 2008 at which point we will not be subject to federal income tax provided that we comply with the organizational and operational requirements.

<u>State</u> - In May 2006, the State of Texas adopted House Bill 3, which modified the state's franchise tax structure, replacing the previous tax based on capital or earned surplus with one based on margin (often referred to as the "Texas Margin Tax") effective with franchise tax reports filed on or after January 1, 2008. The Texas Margin Tax is computed by applying the applicable tax rate (1% for the Company) to the profit margin. At December 31, 2007, we had accrued no state taxes related to the Margin Tax given the loss generated by our investment in MIG IV which represents our sole business activity.

USE OF ESTIMATES

The preparation of financial statements in conformity with U.S. generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

NEW ACCOUNTING STANDARDS

In February 2007 the FASB issued SFAS No. 159, *The Fair Value Option for Financial Assets and Financial Liabilities* (SFAS No. 159). SFAS No. 159 expands opportunities to use fair value measurement in financial reporting and permits entities to choose to measure many financial instruments and certain other items at fair value. This Statement is effective for fiscal years beginning after November 15, 2007. We currently do not plan to measure any eligible financial assets and liabilities at fair value under the provisions of SFAS No. 159.

3. INVESTMENT IN AMREIT MONTHLY INCOME AND GROWTH IV FUND, L.P.

The Company invested \$1,000 as a general partner in MIG IV. MIG IV began raising money in November 2006, and, as of December 31, 2007, had raised approximately \$44 million. We account for our investment in MIG IV under the equity method as we exercise significant influence over, but do not control, the investee. The limited partners of MIG IV have the right, with or without cause, to remove and replace us by a vote of the limited partners owning a majority of the outstanding units. At December 31, 2007 we have recorded distributions and

have been allocated losses in excess of our investment by \$26,000. We believe such amounts are realizable out of the value of the assets of MIG IV. MIG IV was formed to develop, own, manage and add value to commercial real estate properties with an average holding period of two to four years. Our interest in MIG IV as of December 31, 2007 is 1%. MIG IV has invested in a joint venture that owns a real estate property. In 2008, MIG IV acquired control of this joint venture

Pursuant to the MIG IV limited partnership agreement, net sales proceeds from its liquidation (expected in 2013) will be allocated to the limited partners, and to us as the general partner as, if and when the annual return thresholds have been achieved by the limited partners. Certain members of our management team have been assigned a 28.5% aggregate residual interest in any promoted interest that we receive from MIG IV in our capacity as general partner.

Combined condensed financial information for MIG IV (at 100%) is summarized as follows:

Combined Balance Sheets (in thousands)	December 31, 2007
	2007
Assets Investment in non-consolidated subsidiaries	\$17,100
Cash	19,349
Accounts	17,347
Receivable	113
Other assets	1,550
Total Assets	38,112
Liabilities and partners' capital:	
Other liabilities	1,312
Partners	
capital	36,800
Total Liabilities and Partners	
Capital	\$38,112
MIG IV Corp's share of Partners' Capital	\$-
	December 31,
Combined Statement of Operations (in thousands)	2007
Revenue	
Total Revenue	-
Expense	
Interest	
expense	(101
Loss from	
non- consolidated	
entities	(1.123
Other	(1,123 40
Total Expense	(1,184
Tomi Emperior	(1,104
Net (loss) income	\$(1,184
MIG IV Corple chara of Nat (loca) income (1)	\$(12
MIG IV Corp's share of Net (loss) income (1)	\$(12

(1) According to the partnership agreement, no partner shall be required to fund a deficit balance in their capital account. The allocation of net loss during 2007 is prior to a curative allocation of \$25,000 to restore the capital account to zero.

4. RELATED PARTY TRANSACTIONS

We have no employees or offices. We rely on AmREIT and its affiliates in performing these services to us. Services primarily include the supervision of the management. As a result, we are dependent upon AmREIT and its affiliates. In the event that these companies are unable to provide us with the respective services, we would be required to find alternative providers of these services.

MIG IV paid to AmREIT and its affiliates fees and compensation during the organizational stage of the Partnership, including securities commissions and due diligence reimbursements, marketing reimbursements and reimbursement of organizational and offering expenses. As of December 31, 2007, we had no related party receivables or payables related to services provided by AmREIT and its affiliates.

5. COMMITMENTS AND CONTINGENCIES

<u>Litigation</u> - In the ordinary course of business, we may become subject to litigation or claims. There are no material pending legal proceedings known to us.

SIGNATURES

Pursuant to the requirements of Section 12 of the Securities and Exchange Act of 1934, the registrant caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized.

AmREIT Monthly Income & Growth Fund IV, L.P.

By: AmREIT Monthly Income & Growth IV Corporation, its General Partner

Date: August 29, 2008

By: <u>/s/ K. Kerr Taylor</u>
H. Kerr Taylor
President, Chief Executive Officer, and Director

By: <u>/s/ Chad C. Braun</u>
Chad C. Braun
Executive Vice President, Chief Financial Officer, Treasurer and Secretary

PART III ITEM 1. INDEX TO EXHIBITS

INDEX TO EXHIBITS

The following exhibits are filed as part of this Amendment No. 2 to Form 10 Registration Statement and numbered in accordance with Part III of Form 10.

Exhibit	Description
No.	
3.1*	Certificate of Limited Partnership of AmREIT Monthly Income & Growth Fund IV, L.P., dated October 10, 2006.
3.2*	Agreement of Limited Partnership of AmREIT Monthly Income & Growth Fund IV, L.P., dated October 10, 2006, between AmREIT Monthly Income & Growth IV Corporation and the limited partners thereto.
3.3*	Amendment No. 1 to Agreement of Limited Partnership of AmREIT Monthly Income & Growth Fund IV, L.P. dated December 7, 2006.
10.1*	Promissory Note, dated December 8, 2006, between AmREIT Casa Linda, LP and Morgan Stanley Mortgage Capital, Inc.

^{*} Previously Filed

CERTIFICATE OF LIMITED PARTNERSHIP

OF

AMREIT MONTHLY INCOME & GROWTH FUND IV, L.P.

- 1. The name of the limited partnership is AmREIT Monthly Income & Growth Fund IV, L.P.
- 2. The registered office of the Company in the State of Delaware is c/o The Corporation Trust Company, 1209 Orange Street, City of Wilmington, County of New Castle, State of Delaware 19801.
- 3. The name and address of the registered agent for service of process on the Company in the State of Delaware is The Corporation Trust Company, 1209 Orange Street, City of Wilmington, County of New Castle, State of Delaware 19801.
- 4. The address of the principle office in the United States where records of the partnership are to be kept or made available is 8 Greenway Plaza, Suite 1000, Houston, Texas 77046.
 - 5. The name and mailing address, and the street address of the business or residence of the sole general partner is as follows:

Name: AmREIT Monthly Income & Growth IV Corporation

Street and Mailing Address: 8 Greenway Plaza Suite 1000 Houston, Texas 77046

IN WITNESS WHEREOF, the below has caused this Certificate of Limited Partnership to be executed as of the 10th day of October 2006.

AMREIT MONTHLY INCOME & GROWTH IV CORPORATION, the general partner

By: /s/ Chad C. Braun

Name: Chad C. Braun

Title: Vice President, Chief Financial Officer, Secretary and Treasurer

AGREEMENT OF LIMITED PARTNERSHIP

AMREIT MONTHLY INCOME & GROWTH FUND IV, L.P.

A Delaware Limited Partnership

ARTICLE I	NAME, PURPOSE, PLACE OF BUSINESS AND TERM	2
	<u>OF PARTNERSHIP</u>	
1.1	Name	2
1.2	Purpose	2
1.3	Place of Business and Agent	2
1.4	Term	2
		_
ARTICLE II	DEFINITIONS	2
ARTICEE II	DEFINITIONS	_
ARTICLE III	LIMITED PARTNERS	4
3.1	Initial Limited Partner	4
3.2	No Management by Limited Partners	4
3.3	Access to Records	4
3.4		4
3.5	Liability No With drawal on Dissolution	-
	No Withdrawal or Dissolution	4
3.6	Consent	4
3.7	Power of Attorney	4
		_
ARTICLE IV	CAPITAL CONTRIBUTIONS, ACCOUNTS AND	5
	WITHDRAWALS	
4.1	Initial Capital Contributions	5
4.2	Limited Partner Capital Contributions	5
4.3	No Interest on Capital Contributions	5
4.4	Liability Limited to Capital	5
4.5	Issuance of Additional Units	5
ARTICLE V	DUTIES AND POWERS OF PARTNERS, PARTNERSHIP	6
	EXPENSES	
5.1	General Partner	6
5.2	General Authority and Powers of General Partner	6
5.3	Limitations on Authority and Powers of General Partner	6
5.4	Loans to the Partnership	6
5.5	Liability	7
5.6	Outside Activities of the General Partner	8
5.7	Indemnification and Liability of General Partner	8
5.8	Partnership Expenses	8
5.9	Compensation	8
5.10		8
3.10	General Partner Investment	o
E 11	Joint Ventures	8
5.11		
5.12	Merger, Exchange and Conversion	9
A DELCT E VI	DICEDIDIFICATO	10
ARTICLE VI	DISTRIBUTIONS	10
6.1	Share of Distributions	10
6.2	Source of Funds	10
6.3	Liquidation or Dissolution	10
6.4	Amount and Time of Distributions	10
ARTICLE VII	PROFITS AND LOSSES	11
7.1	Capital Account	11
7.2	Deficit Capital Account Balances	11
7.3	Allocations	11
7.4	Taxable Gain on Exchange	11

7.5	Depreciation Recapture	11
7.6	Allocation between Assignor and Assignee	11
7.7	Express Consent to Allocations	11
7.8	Adjustments to Gain and Loss	12
7.9	Tax Allocations: Code Section 704(c)	12
7.10	General Provisions Relating to Distributions, Net Profit and Net Loss Allocations	12
7.11	Adjustments to Allocations	12
7.12	Fractions Rule	12
ARTICLE VIII	TERMINATION AND DISSOLUTION OF THE PARTNERSHIP	13
8.1	No Termination by Admission or Incapacity of Limited Partners	13
8.2	Termination of the Partnership	13
8.3	Continuation of Partnership upon Certain Events	13
8.4	Dissolution and Liquidation	13
ARTICLE IX	WITHDRAWAL AND TRANSFER BY PARTNERS	14
9.1	Voluntary Withdrawal or Assignment by the General Partner	14
9.2	Involuntary Withdrawal and Transfer of a General Partner's Interest	14
9.3	Transfer by Limited Partners	14
9.4	Put Right	14
ARTICLE X	BOOKS, RECORDS AND ACCOUNTING	15
10.1	Books and Records	15
10.2	Accounting Method; Audits and Reports	15
10.3	Bank Accounts	15
ARTICLE XI	SPECIAL TAX MATTERS	16
11.1	Tax Elections	16
11.2	Tax Audit Expenses	16
ARTICLE XII	AMENDMENTS AND MEETINGS	17
12.1	General	17
12.2	Action Without a Meeting	17
12.3	Amendments on Admission or Withdrawal of Partners and	17
	Clarification Amendments	
ARTICLE XIII	MISCELLANEOUS	18
13.1	Notices	18
13.2	Further Assurances	18
13.3	Representations of the Partners	18
13.4	Survival of Representations and Agreements	18
13.5	Agreement in Counterparts	18
13.6	Captions and Context	18
13.7	Construction	18
13.8	Governing Law, Successors	18
13.9	Binding Effect	18
13.10 13.11	Validity of Agreement Creditors Not Benefited	18 18
13.11	CI cultura Mut Dellenteu	10

AMREIT MONTHLY INCOME & GROWTH FUND IV, L.P.

A Delaware Limited Partnership

AGREEMENT OF LIMITED PARTNERSHIP

AMREIT MONTHLY INCOME & GROWTH FUND IV, L.P., a Delaware limited partnership, (the "Partnership") is hereby formed as a limited partnership under the laws of the state of Delaware pursuant to this Agreement of Limited Partnership entered into by and between Amreit Monthly Income & Growth IV Corporation, a Delaware corporation, as the General Partner (the "General Partner") and Amreit, the Initial Limited Partner. The Certificate of Limited Partnership of the Partnership shall be promptly recorded with the Secretary of State for the State of Delaware as required by the Delaware Revised Uniform Limited Partnership Act.

ARTICLE I

NAME, PURPOSE, PLACE OF BUSINESS AND TERM OF PARTNERSHIP

- 1.1. Name. The name of the Partnership is "AMREIT MONTHLY INCOME & GROWTH FUND IV, L.P."
- 1.2. <u>Purpose</u>. The purpose of the Partnership shall be to acquire, develop, own, operate, improve, lease, sell, manage and hold for investment real property and to make and/or invest in loans for the acquisition, development or construction of real property, either alone or in association with others, and to conduct any other activity in connection therewith permitted by law, any of which will be permitted regardless of whether any Partner or Affiliate has a direct or indirect interest in the activity.
- 1.3. Place of Business and Agent. The principal place of business and office of the Partnership shall be at c/o AmREIT, 8 Greenway Plaza, Suite 1000, Houston, Texas 77046 or such other location as may hereafter be determined by the General Partner on prior notice to the Limited Partners. The name and address of the registered agent for service of process on the Partnership in the State of Delaware shall be Capitol Services, Inc., 615 South Dupont Highway, County of Kent, Dover, Delaware 19901.
- 1.4. <u>Term.</u> The term of the Partnership commenced as of the date of filing of the original Certificate and shall continue until December 31, 2026, or until sooner dissolved or terminated as provided in this Agreement.

ARTICLE II

DEFINITIONS

As used herein the following terms shall have the following meanings:

"Act" shall mean the Delaware Revised Uniform Limited Partnership Act, as amended.

"Adjusted Capital Account" shall mean the Capital Account maintained for each Partner which is determined as of the end of each Partnership Fiscal Year, or other appropriate occasion, (i) increased by any amounts which such Partner is obligated to restore pursuant to any provision of this Agreement or Regulations Section 1.704-1(b)(2)(ii)(c) or is deemed to be obligated to restore pursuant to the penultimate sentences of Regulations Sections 1.704-2(g)(1) and 1.704-2(i)(5) and (ii) decreased by the items described in Regulations Sections 1.704-1(b)(2)(ii)(d)(4), 1.704-1(b)(2)(ii)(d)(5), and 1.704-1(b)(2)(ii)(d)(6). The foregoing definition of Adjusted Capital Account is intended to comply with the provisions of Regulations Section 1.704-1(b)(2)(ii)(d) and shall be interpreted consistently therewith.

"Adjusted Capital Account Deficit" shall mean, with respect to a Partner, the deficit balance, if any, in that Partner's Adjusted Capital Account.

"Affiliate" shall mean, when used with reference to a specified Person, (i) any Person directly or indirectly controlling, controlled by or under common control with another Person, (ii) any Person owning or controlling ten percent or more of the outstanding voting securities

of such other Person, (iii) any officer, director, member, manager or partner of such Person or any Person specified in (i) or (ii) above, and (iv) any company in which any officer or director of such Person is an officer, director, member, manager or partner; provided, however, that for purposes of this definition the term "Affiliate" will not be deemed to include any Person providing legal, accounting or other professional services to the Partnership or General Partner.

"Agreement" shall mean this Agreement of Limited Partnership and any amendments hereto.

"Annual Return" shall mean with respect to each Partner, an annual, cumulative, non-compounded return on such Partners' aggregate Unreturned Deemed Capital Contribution calculated like interest using the average monthly balance of such Partners' aggregate Unreturned Deemed Capital Contribution over the time period in question.

"Applicable Percentage" shall mean with respect to a particular year the annual distribution percentage determined by the General Partner, targeted to begin at 7.5% per annum.

"Capital Account" shall have the meaning assigned to such term in Section 7.1 hereof.

"Capital Account Balance" shall mean with respect to a Partner, the balance in that Partner's Capital Account.

"Capital Account Deficit" shall mean with respect to a Partner, the deficit balance, if any, in that Partner's Capital Account.

"Capital Contribution" shall mean, with respect to any Partner, the amount which such Partner contributes to the capital of the Partnership in accordance with the provisions of this Agreement, including without limitation any initial Capital Contribution and any additional Capital Contributions made by the Partner to the Partnership

"Certificate" shall mean the Certificate of Limited Partnership of the Partnership to be filed with the Secretary of State for the State of Delaware.

"Code" shall mean the Internal Revenue Code of 1986, as amended.

"Consent" shall mean the consent of a Person to do the act or thing for which the consent is solicited, or the act of granting such consent, as the context may require. The Consent of the Limited Partners will be deemed to have been obtained if Limited Partners identified in the Partnership's records affirmatively vote the number of outstanding Units required for such action in favor of such action.

"Deemed Capital Contribution" shall mean the product of \$25,000 per Unit (regardless of the purchase price paid for a Unit) multiplied by the number of Units owned by a Partner.

"Event of Bankruptcy" shall mean with respect to any Person: (i) the entry of an "order for relief," as defined in Title 11 of the United States Code ("Bankruptcy Code"), in either a voluntary case when instituted by such Person or in an involuntary case when instituted against such Person if such Person may be a debtor under the chapter of the Bankruptcy Code under which such case is commenced and which case shall not be dismissed or closed within ninety (90) days of such commencement; or a comparable action against or by such Person under any other Federal, state or foreign law related to bankruptcy or insolvency; or (ii) the filing by such Person of a petition or answer or consent seeking debtor relief under any other applicable federal, state or foreign law; or (iii) the entry of an order or decree appointing, or consent by such Person to the appointment of, a receiver, liquidator, assignee, trustee, sequestrator (or other similar official) of such Person or of all or a substantial part of the property of such Person or ordering the winding up or liquidation of the affairs of such Person; or (iv) the making by such Person of any assignment for the benefit of creditors or the admission by such Person of his or its inability to pay his or its debts generally as they come due, or the taking of any corporate or other action by such Person in furtherance of any such action; provided, in any event that the debtor's Partnership Interest in the Partnership must be subject to the bankruptcy or liquidation proceedings and not exempt from creditors.

"Event of Dissolution" shall mean the liquidation, dissolution, removal, or the occurrence of an Event of Bankruptcy with respect to a Partner or any successor.

"General Partner" shall mean the Person or Persons designated as General Partner in the introductory paragraph hereof and any successor Person named as a General Partner according to the terms of this Agreement.

"General Partner Investment" shall have the meaning assigned to such term in Section 4.2(a) hereof.

"GP Excess Distributions" shall have the meaning assigned to such term in Section 6.1(b)(4) hereof.

"Initial Limited Partner" shall mean AmREIT, whose address is set forth above in Section 1.3.

"Invested Capital" means the gross purchase price for his or her Unit(s) originally paid by a Limited Partner multiplied by the number of Units owned by a Limited Partner.

"Joint Ventures" shall have the meaning assigned to such term in Section 5.11 hereof.

"Limited Partners" shall mean the Persons identified as Limited Partners on the signature pages to this Agreement and/or each Person admitted to the Partnership as a Limited Partner upon executing a Subscription Agreement and contributing capital to the Partnership in exchange for Units, and such successor or additional Persons admitted to the Partnership as Limited Partners according to the terms of this Agreement.

"Limited Partner List" shall have the meaning assigned to such term in Section 3.3(a) hereof.

"LP Excess Distributions" shall mean cumulative distributions received by Limited Partners pursuant to Sections 6.1(a), 6.1(b)(1), and 6.1(b)(3) in excess of such Limited Partner's Invested Capital.

"LP Target Distribution Amount" shall mean an amount computed for the Limited Partners, as of any determination date, equal to the amount of cash, which if distributed to the Limited Partners, would be sufficient such that no additional amounts would be required to be distributed to the Limited Partners pursuant to Sections 6.1(b)(1) and 6.1(b)(3).

"Majority in Interest" shall mean a majority of the Units held by all Limited Partners.

Table of Contents

"Net Cash" shall mean cash funds derived from all sources by the Partnership (including, but not limited to, cash from operations, cash released from reserves, interest, other investment income, and net proceeds generated from the disposition of properties), but excluding (i) funds used to pay or to provide for the payment of all operating expenses of the Partnership and each Partnership property and debt service, if any, capital improvements and replacements, without deduction for depreciation or amortization and (ii) cash reserves (a) required by any loan agreements or similar arrangements to which the Partnership is subject, (b) necessary to satisfy contingencies reasonably anticipated for, or associated with, the Partnership's business, or (c) considered necessary or desirable by the General Partner, in its sole discretion (including for the purchase of additional assets).

"Net Profits" and "Net Losses" shall mean the profits or losses, as the case may be, of the Partnership for federal income tax purposes for each Partnership Fiscal Year determined in accordance with the Code, the Regulations and the accounting method followed by the Partnership for such purposes.

"Nonrecourse Deduction" has the meaning assigned to it in Regulations Section 1.704-2(c).

"Notification" or "Notice" shall mean a writing containing the information required by this Agreement to be communicated to any Person in the manner provided in Section 13.1 hereof; provided, however, that any communication containing such information sent or delivered to such Person and actually received by such Person constitutes Notification for all purposes under this Agreement.

"Offering" means that certain offering by the Partnership of an initial 2,000 Units (at an offering price of \$25,000 per Unit) pursuant to the Private Offering Memorandum, which may be increased in the discretion of the General Partner by an additional 2,000 units.

"Operating Period" means the period ending on the seventh anniversary of the date of the Private Offering Memorandum, unless extended pursuant to Section 5.3(a)(6).

"Partner" shall mean either the General Partner or a Limited Partner and "Partners" shall mean the General Partner and the Limited Partners, collectively.

"Partner Nonrecourse Debt" has the meaning assigned to it in Regulations Section 1.704-2(b)(4).

"Partner Nonrecourse Debt Minimum Gain" has the meaning assigned to it in Regulations Section 1.704-2(i)(2) and (3).

"Partner Nonrecourse Deductions" has the meaning assigned to it in Regulations Section 1.704-2(i)(2).

"Partnership" shall mean the limited partnership formed pursuant to the execution of this Agreement and the filing of the Certificate.

"Partnership Fiscal Year" shall mean the calendar year, unless otherwise determined by the Partnership's tax advisors and approved by the General Partner upon notice to the Limited Partners.

"Partnership Interest," "Interest" or "Interest of a Partner" shall mean the particular Partner's interest in the Partnership including, without limitation, the Partner's Units and interest in the Partnership's Net Profits, Net Losses, income, gains, losses, deductions and distributions.

"Partnership Minimum Gain" shall have the meaning set forth in Regulations Section 1.704-2(b)(2) and the amount of Partnership Minimum Gain, as well as any net increase or decrease in Partnership Minimum Gain, for a Partnership Fiscal Year shall be determined in accordance with the rules of Regulations Section 1.704-2(d).

"Person" shall mean an individual, trust, estate, partnership, joint venture, company, corporation or other entity.

"Private Offering Memorandum" means that certain Private Offering Memorandum dated November 15, 2006, as supplemented and amended from time to time during the Offering, pursuant to which the Partnership shall engage in the Offering.

"Project" shall mean any real property, including any interest therein, which is developed and/or owned by the Partnership.

"Put Partner" shall have the meaning assigned to such term in Section 9.4 hereof.

"Regulations" shall mean the Treasury Regulations promulgated under the Code.

"Representative" shall mean the executor, administrator, guardian, trustee, or other personal representative of a Partner.

- "Requested Redemption Date" shall have the meaning assigned to such term in Section 9.4 hereof.
- "Retirement Trusts" shall have the meaning assigned to such term in Section 4.2(d) hereof.
- "Subscription Agreements" shall mean the subscription agreements executed by the Limited Partners for Units.
- "Substitute Limited Partner" shall mean a Limited Partner admitted to the Partnership pursuant to Section 9.3.
- "Unit" shall mean an interest in the Partnership as a Limited Partner, which is acquired either by means of a Capital Contribution to the Partnership or by transfer. A qualified person may acquire more than one Unit and/or a fraction of a Unit upon approval of the General Partner.
- "Unreturned Deemed Capital Contribution" means each Partner's Deemed Capital Contribution less cumulative distributions to date made to the Partner pursuant to Section 6.1(b)(1) or which otherwise constitute a return of capital. For purposes of determining "Unreturned Deemed Capital Contribution" the Limited Partners' Unreturned Deemed Capital Contribution shall be zero at such time as there has been a return of all of the Limited Partners' Invested Capital.

"Unreturned Invested Capital" means each Partner's Invested Capital less cumulative distributions to date made to the Partner pursuant to Section 6.1(b)(1) or which otherwise constitute a return of capital.

ARTICLE III

LIMITED PARTNERS

- 3.1. <u>Initial Limited Partner</u>. Upon the admission of additional Limited Partners, the Limited Partner interest of the Initial Limited Partner shall be repurchased in full, and it is recognized and acknowledged that the capital of \$200 contributed by the Initial Limited Partner will be returned at that time.
- 3.2. No Management by Limited Partners. Limited Partners as such shall take no part in, or at any time interfere in any manner with, the management, conduct, or control of the Partnership's business and operations and shall have no right or authority to act or bind the Partnership in any manner whatsoever. Except as expressly provided for in this Agreement, the Limited Partners do not have the right to vote on or consent to any matter, nor is their vote or consent required for any action by the Partnership or the General Partner.
- 3.3. Access to Records. Each Limited Partner holding not less than 5% of the outstanding Units shall, at reasonable business hours, have access to the records of the Partnership subject to Article X and at its expense may inspect any of them; provided that the Limited Partner holding not less than 5% of the outstanding Units exercising such right (including, without limitation, any right to access or receive the Limited Partner List) shall not unreasonably interfere with or disrupt Partnership business. With respect to access of the list of Limited Partners, the Partnership shall prepare and make available the Limited Partner List upon request as follows:
- (a) An alphabetical list of the names, addresses and business telephone numbers of the Limited Partners along with the number of Units held by each of them (the "Limited Partner List") shall be maintained as a part of the books and records of the Partnership and shall be available for inspection by Limited Partners holding not less than 5% of the outstanding Units at the home office of the Partnership upon the request of such Limited Partners holding not less than 5% of the outstanding Units;
 - (b) The Limited Partner List shall be updated at least quarterly to reflect changes in the information contained therein; and
- (c) The General Partner may, in its sole discretion, grant access to the records of the Partnership and the Limited Partner List to a Limited Partner holding less than 5% of the outstanding Units upon written request by such Limited Partner stating the reasons for such request.
- 3.4. <u>Liability</u>. The Limited Partners shall be liable with respect to the Partnership only to the extent of the amount of the contributions to capital made by such Limited Partners as provided in Section 4.2. The Units shall be nonassessable.
- 3.5. <u>No Withdrawal or Dissolution</u>. No Limited Partner shall at any time withdraw from the Partnership except as provided in this Agreement. No Limited Partner shall have the right to have the Partnership dissolved or to have his contribution to the capital of the Partnership returned except as provided in this Agreement. The death, incompetency, disability, dissolution, termination, or bankruptcy of a Limited Partner shall not dissolve or terminate the Partnership.
- 3.6. <u>Consent</u>. To the fullest extent permitted by law, each of the Limited Partners hereby consents to the exercise by the General Partner of all the rights and powers conferred on the General Partner by this Agreement and the Act.
- 3.7. Power of Attorney. Each of the Limited Partners, by the execution of this Agreement, does hereby irrevocably constitute and appoint the General Partner, with full power of substitution, as his or its true and lawful attorney-in-fact and agent, with full power and authority in the Limited Partner's name, place and stead, to execute, acknowledge, swear to, deliver, file and record in the appropriate public offices (i) all certificates and other instruments (including counterparts of this Agreement and amendments hereto) that the General Partner deems appropriate to form, qualify or continue the Partnership as a limited partnership, or a partnership in which the Limited Partners have limited liability, in all jurisdictions in which the Partnership may conduct business or own properties, (ii) all instruments that the General Partner deems appropriate to reflect any amendment, change or modification of this Agreement in accordance with the terms hereof (including, without limitation, one or more amendments to this Agreement to reflect any assignments or issuances of Interests in the Partnership), (iii) all conveyances and other instruments that the General Partner deems appropriate to reflect the dissolution and termination of the Partnership pursuant to the terms of this Agreement and (iv) instruments relating to the admission of any Partner pursuant to Article IX. The power of attorney granted herein is hereby declared irrevocable and, being a power coupled with an interest, shall survive the death, incompetency, disability, bankruptcy, dissolution or other termination of any Partner and shall extend to such Partner's heirs, successors and assigns. Each Partner hereby agrees to be bound by any representations, covenants, or agreements made by the General Partner acting in good faith pursuant to such power of attorney, and each Partner hereby waives any and all defenses that may be available to contest, negate or disaffirm any action of the General Partner taken in good faith under such power of attorney. Each Partner hereby agrees to execute and deliver to the General Partner, within fifteen (15) days after receipt of the General Partner's written request therefor, such other and further statements of interest and holding, designations, powers of attorney and other instruments as the General Partner deems necessary to effectuate this Agreement and the purposes of this Partnership. If a

Limited Partner assigns his interest in the Partnership, as provided in Article IX, the foregoing power of attorney shall survive the delivery of the instruments effecting such assignment for the purpose of enabling the General Partner to sign, swear to, execute and acknowledge and file any and all amendments to this Agreement or the Certificate and other instruments and documents necessary to effectuate the substitution of the assignee as a Limited Partner.

ARTICLE IV

CAPITAL CONTRIBUTIONS, ACCOUNTS AND WITHDRAWALS

4.1. <u>Initial Capital Contributions</u>.

- (a) <u>General Partner's Capital Contribution</u>. The General Partner shall contribute \$1,000 to the capital of the Partnership for its Interest in the Partnership as a General Partner.
- (b) <u>Initial Limited Partner's Capital Contribution</u>. The Initial Limited Partner shall contribute \$200 to the capital of the Partnership for its Interest in the Partnership as Limited Partner.

4.2. <u>Limited Partner Capital Contributions</u>.

- (a) <u>Capital Contributions</u>. Prior to or contemporaneously with the Initial Closing Date (as defined in the Private Offering Memorandum), the General Partner shall purchase 36 Units in exchange for cash in the amount of \$800,000 (the "General Partner Investment"), which purchase shall not be deemed a part of the Offering. There shall be available for subscription in the Offering by prospective Limited Partners an aggregate initial amount of up to 2,000 Units, each Unit requiring such initial Capital Contribution as provided in the Private Offering Memorandum (the foregoing shall not be deemed to limit the General Partner's ability to issue fractional Units). Following the sale of the initial 2,000 Units, the General Partner shall have the right, in its sole discretion, to cause the Partnership to issue and sell up to an additional 2,000 Units subject to the same terms and conditions. Each Limited Partner investing pursuant to the Offering shall make a Capital Contribution by submitting to the registered broker-dealer designated by the General Partner the executed Subscription Agreement, as well as any other documents required by the Private Offering Memorandum, together with a check payable to the applicable account established by the Partnership in the amount of \$25,000 for each Unit to which the Subscription Agreement relates (or the pro rata amount of the Unit purchase price for fractional Units) less any discounts authorized in the Private Offering Memorandum. The Initial Limited Partner (prior to its redemption pursuant to Section 3.1), all subscribers for Units whose subscriptions are accepted by the General Partner and the General Partner (with respect to any Units purchased or otherwise held by the General Partner) are collectively referred to as the "Limited Partners."
- (b) Requirements for Limited Partner Status. A Limited Partner shall be admitted to the Partnership not later than the first day of the month following his purchase of Units, provided that his Capital Contribution has been received by the 20th day of the preceding month.
- (c) <u>Additional Capital Contributions</u>. The Limited Partners shall not be obligated to make any additional Capital Contributions to the capital of the Partnership.
- (d) Admission of Retirement Trusts as Limited Partners. The General Partner shall establish a fixed maximum percentage of Units permitted to be held by an "employee benefit plan," as described in Section 3(3) of ERISA, or an "individual retirement account," as described in Section 408 of the Code (each, a "Retirement Trust") or other tax exempt entities and prohibit transfers of Units pursuant to Sections 9.3(a)(2) and/or 9.3(b) in connection therewith, in order to permit the Units to fall within the insignificant investment exclusion from the definition of plan assets contained in Section 2510.3-101(f) of Title 29 of the Code of Federal Regulations.
 - 4.3. No Interest on Capital Contributions. No interest shall be paid by the Partnership to any Partner on any Capital Contribution.
- 4.4. <u>Liability Limited to Capital</u>. The liability of Limited Partners shall be limited to the amount of Capital Contributions which Limited Partners actually make or are required to make in accordance with the provisions of Section 4.2 of this Agreement. Limited Partners shall not have any further personal liability to contribute money to the Partnership with respect to the liabilities or obligations of the Partnership, nor shall the Limited Partners be personally liable for any obligations of the Partnership. However, if any distribution or distributions constituting a return of capital shall have been made to a Limited Partner, then such Limited Partner may, to the extent provided by the Act, be obligated to repay all or part of any such distributions. Any repayment of distributions shall be made to the Partnership within 30 days after the General Partner shall have delivered to such Limited Partner written Notice asserting such repayment, together with a statement of the amount chargeable to such Limited Partner to be repaid and an explanation of the reasons for such repayment.
- 4.5. <u>Issuances of Additional Units</u>. The General Partner is hereby authorized to cause the Partnership to issue additional Units for any Partnership purpose, at any time or from time to time, to Partners (including the General Partner) or to other Persons for such consideration and on such terms and conditions as shall be established by the General Partner in its sole discretion and absolute discretion, all without the approval of the Limited Partners. Any additional Units issued thereby may be issued in one or more classes, or in one or more series of any of such classes, with designations, preferences and relative participating, option or other special rights, powers and duties, including rights, powers

and duties senior to Units previously issued to the Limited Partners, all as shall be determined by the General Partner in its sole discretion and without the approval of any Limited Partner, subject to Delaware law, including without limitation, (A) the allocation or items of Partnership income, gain, loss, deduction and credit to each such class or series of Units; (B) the right of each such class or series of Units to share in Partnership distributions; and (C) the rights of each such class or series of Units upon dissolution and liquidation of the Partnership; provided, however, that no additional Units will be issued to the General Partner unless: (1) the additional Units are issued in exchange for property or other assets owned by the General Partner with a fair market value, as determined by the General Partner, in good faith, equal to the value to the Units; or (2) the additional Units are issued to all Partners in proportion to their respective percentage ownership of the Units.

ARTICLE V

DUTIES AND POWERS OF PARTNERS, PARTNERSHIP EXPENSES

- 5.1. <u>General Partner</u>. The General Partner is AmREIT Monthly Income & Growth IV Corporation, a Delaware corporation, whose address is 8 Greenway Plaza, Suite 1000, Houston, Texas 77046.
- 5.2. General Authority and Powers of General Partner. Subject to the terms and conditions of this Agreement, the General Partner shall have complete authority over and exclusive control and management of the business and affairs of the Partnership and shall devote such time to the Partnership as may be reasonably required for the achievement of its purposes, in addition to all other responsibilities of the General Partner. Without limiting the foregoing, the General Partner is empowered to spend Partnership funds on non-refundable options, fees to third parties or Affiliates to investigate, study and evaluate possible land acquisition and development opportunities and other similar expenditures which may be written off in full if the Partnership is unable to acquire and develop a property, to contract for the purchase or development of commercial real estate assets and to dispose and liquidate the assets. In connection with such management, the General Partner may, subject to this Article V, employ on behalf of the Partnership any other Persons to perform services for the Partnership, including Persons employed by, affiliated with, or related to any Partner.
 - 5.3. <u>Limitations on Authority and Powers of General Partner</u>.
- (a) Notwithstanding any other provision of this Agreement to the contrary, the General Partner shall not, without the prior vote or written Consent of a Majority in Interest of the Limited Partners, do any of the following:
 - 1. amend this Agreement except:
 - A. to admit a Substitute Limited Partner or additional Limited Partners pursuant to Section 4.2 and 9.3;
 - B. to reflect the withdrawal of a Limited Partner;
- C. to make a clarification to the Agreement or in connection with the performance of ministerial acts; provided that, such amendment does not change the intent of the Agreement or materially impair a Limited Partner's Interest or rights in the Partnership;
- D. to change the name or location of the principal place of business, registered agent or registered office of the Partnership, provided that notice of such change is provided in writing to the Limited Partners;
 - E. to reflect the disposition of any Limited Partner's Interest; or
 - F. as otherwise expressly permitted in this Agreement, including without limitation, Section 12.1;
 - 2. confess a judgment against the Partnership in an amount exceeding \$100,000;
- 3. possess any property or assign the rights of the Partnership in specific property, including a Project, for other than a Partnership purpose;
- 4. except as provided in Sections 9.1 or 9.2, resign as the General Partner, transfer its General Partner Interest or admit a Person as a substitute General Partner;
- 5. incur indebtedness on behalf of the Partnership in an amount greater than 75% of the Partnership's cash and cash equivalents plus the aggregate market value of the Projects, which market values shall be determined by the General Partner based on a capitalization rate approval applied to the net operating income of the Projects;
- 6. extend the Operating Period from seven years after the date of the Private Offering Memorandum to nine years after the date of the Private Offering Memorandum;
 - 7. invest more than 15% of the gross proceeds from the Offering in properties outside of the U.S.;
 - 8. invest more than 20% of the gross proceeds from the Offering in investments in mortgages; or

- 9. invest in the securities of other issuers, except for permitted temporary investments pending utilization of Partnership funds for investments in real property and except for investments in Joint Ventures and investments in mortgages pursuant to Section 5.3(a)(8) above.
- (b) Notwithstanding any other provision in this Agreement to the contrary, the General Partner shall not, without the express vote or prior written Consent of each Limited Partner adversely affected by such action, do any of the following:
- 1. perform any act in contravention of this Agreement which would make it impossible to carry on the business of the Partnership;
 - 2. perform any act which would subject the Limited Partners to liability as general partners in any jurisdiction; or
- 3. amend this Agreement to (i) convert the interest of a Limited Partner into that of a general partner; (ii) adversely affect the limited liability of a Limited Partner; or (iii) alter the interests of the Partners in distributions, except to the extent that distributions are altered solely as a result of the issuance of Units pursuant to Article IV.
- (c) Notwithstanding any other provision of this Agreement to the contrary, the General Partner shall not vote its limited partner interest in matters which affect the General Partner or the economic interest of the General Partner.
- 5.4. Loans to the Partnership. In the event that additional funds are required by the Partnership for any valid purpose relating to the business of the Partnership, or for any of its obligations, expenses, or costs, including operating deficits, the Partnership may borrow such funds as are needed from any Person (including, without limitation, a Partner) on terms and conditions agreeable between the Partnership and the lender; provided that if such loan is from a General Partner or an Affiliate thereof, the simple rate of interest shall not exceed the then prevailing regional prime interest rate at the Bank of America (or its successor) plus one percent, but in no event to exceed the interest rate that would be charged to the General Partner by said bank. No financing charges, fees, penalties or similar provisions will be included in any General Partner loan to the Partnership unless the General Partner must itself borrow such funds, in which event any financing charges, fees or penalties to the Partnership shall be no greater than those imposed on the General Partner by the lending institution.

Table of Contents

- 5.5. <u>Liability</u>. The General Partner, its Affiliates, employees and agents shall not be liable, responsible or accountable in damages or otherwise to any Partner for any act or omission performed or omitted by the General Partner or its duly authorized Affiliates, employees or agents in good faith on behalf of the Partnership and in a manner reasonably believed to be within the scope of the authority granted by this Agreement and in the best interests of the Partnership, except for gross negligence or intentional misconduct.
- 5.6. Outside Activities of the General Partner. The General Partner, during the term of this Partnership, may engage in and possess an interest for its own account in other business ventures of every nature and description, independently or with others, including, but not limited to, the ownership, financing, leasing, operation, management, syndication, brokerage, investment in and development of real estate; and neither the Partnership nor any Partner, by virtue of this Agreement shall have any right in and to said independent ventures or any income or profits derived therefrom. With respect to activities permitted by the preceding sentence, the General Partner may engage in such ownership, businesses and activities in direct competition with the Partnership without having or incurring any obligation to offer any such interest in such properties, business or activities to the Partnership or the Limited Partners, and no other provision of this Agreement shall be deemed to prohibit the General Partner from conducting such businesses and activities. The General Partner shall not be required under this Agreement to devote its financial capabilities or means, personnel and other resources exclusively for the benefit or exclusively on behalf of the Partnership or on activities in which the Partnership is participating or will participate, but rather shall only be required to devote so much of such resources as may be necessary to promote in good faith the business of the Partnership and to exercise its responsibilities in a fiduciary manner. Neither the General Partner nor any Affiliate of the General Partner shall have any obligation to allow the Partnership or the Limited Partners to invest in any property of the General Partner or any Affiliate of the General Partner.

5.7. Indemnification and Liability of General Partner.

- (a) The Partnership shall indemnify each of the General Partner and its Affiliates against any claim or liability incurred or imposed upon such General Partner or such Affiliates relative to the Partnership, provided that such General Partner or Affiliate has determined, in good faith, that the course of conduct which caused the loss or liability was in the best interests of the Partnership, and such General Partner or Affiliate was not guilty of gross negligence or willful or wanton misconduct. Neither the General Partner nor any of its Affiliates shall be liable to the Partnership or any Partner by reason of any act or omission of such General Partner or Affiliate, provided that the General Partner has determined, in good faith, that the course of conduct which caused the loss or liability was in the best interests of the Partnership, and such General Partner or Affiliate was not guilty of gross negligence or willful or wanton misconduct. Solely for the purposes of this Section 5.7 but for all such purposes, the term "Affiliate" shall mean only an Affiliate that has performed an act on behalf of the Partnership or the General Partner and for which the General Partner would have been entitled to indemnification had such act been performed by the General Partner.
- (b) Notwithstanding anything to the contrary contained in Section 5.7(a) above, neither the General Partner nor any of its Affiliates shall be indemnified from any liability, loss or damage incurred by them arising due to an alleged violation of federal or state securities laws unless (i) there have been a successfully adjudication on the merits of each count involving alleged securities law violations as to the particular indemnified party, (ii) such claims have been dismissed with prejudice on the merits by a court of competent jurisdiction as to the particular indemnified party, or (iii) a court of competent jurisdiction approves a settlement of the claims against the particular indemnified party and finds that indemnification of the settlement and related costs should be made. Prior to seeking a court approval for indemnification, the General Partner shall undertake to cause the party seeking indemnification to apprise the court of the position of the Securities and Exchange Commission with respect to indemnification for securities violations.
- (c) The Partnership shall not incur the cost of the portion of any insurance which insures any party against any liability as to which such party is prohibited from being indemnified as set forth above.
- (d) Any indemnification pursuant to this Section, or otherwise, shall be recoverable only from the assets of the Partnership and not from any of the Limited Partners.
- (e) Notwithstanding anything to the contrary contained elsewhere in this Agreement, neither the General Partner nor any of its Affiliates shall be liable, responsible or accountable in damages or otherwise to any other Partner, the Partnership, its receiver or trustee for, and such Person shall be indemnified hereunder with respect to, any claims to the extent relating to any of the events or outcomes set forth in the Private Offering Memorandum as possible results, outcomes or risks associated with the business of the Partnership or the investment in Units.
- 5.8. Partnership Expenses. All of the Partnership's expenses will be paid by the Partnership. Reimbursements to the General Partner or any Affiliate shall not exceed such Person's actual cost for the service, goods or materials used for or by the Partnership. Notwithstanding the foregoing, the Partnership will reimburse the General Partner or its Affiliates for organization and offering expenses in connection with the Offering, including legal and accounting fees, printing costs, filing fees and distribution costs, not to exceed 1.0% of the gross proceeds of the Offering.

Table of Contents

- 5.9. <u>Compensation</u>. Compensatory arrangements between the Partnership and the General Partner and its Affiliates shall be subject to the following:
- (a) <u>Asset Management Fees</u>. The General Partner or its Affiliates may provide asset management services in exchange for fees equal to 1% of the net equity under management.
- (b) <u>Acquisition Fees</u>. The General Partner, its Affiliates or unrelated third parties may provide acquisition services for Projects acquired by the Partnership in exchange for acquisition fees not to exceed, for any single acquisition, 3% of the contract purchase price up to \$20,000,000, and 2% of the contract purchase price in excess of \$20,000,000. The Partnership will not pay acquisition fees for a Project with respect to which the Partnership has paid a development or redevelopment fee.
- (c) <u>Development and Redevelopment Fees</u>. The General Partner, its Affiliates or unrelated third parties may provide development and redevelopment services for Projects developed and redeveloped by the Partnership in exchange for development and redevelopment fees not to exceed 6% of the total costs up to \$10,000,000, 5% of the total costs between \$10,000,000 and \$20,000,000 and 4% of the total costs in excess of \$20,000,000. The Partnership will not pay a development or redevelopment fee for a Project where the Partnership has paid an acquisition fee.
- (d) <u>Property Leasing Fees</u>. The General Partner, its Affiliates or unrelated third parties may provide leasing services in connection with Projects in exchange for fees not to exceed 4% of base rent on a lease renewal and not to exceed 6% of base rent on an initial lease; provided, however, that on leases of over 10,000 square feet, leasing fees may be paid on the basis of square footage, not to exceed \$6.00 per square foot.
- (e) <u>Property Management Fees</u>. The General Partner, its Affiliates or unrelated third parties may provide property management services in connection with the Projects in exchange for fees not to exceed 4% of gross revenue on multi-tenant or multi-pad properties.
- (f) <u>Real Estate Brokerage Commissions</u>. The General Partner, its Affiliates or unrelated third parties may provide real estate brokerage services in connection with Projects in exchange for commissions not to exceed 4% of the sales price on individually brokered transactions and not to exceed 6% of the sales price on co-brokered transactions. The General Partner and its affiliates will not be paid real estate brokerage commissions on the sale of a property if such property has not generated an annual return of at least 8.5% per annum on the equity contributed to such property.
- (g) <u>Financing</u>. The General Partner or an Affiliate may, but is not obligated to, provide Project or investment financing to the Partnership at rates not in excess of then prevailing market rates available from unrelated institutional lenders.
- (h) <u>Reimbursements</u>. The General Partner and/or its Affiliates shall be reimbursed by the Partnership for their actual costs in performing acquisition, development, management and administrative functions for the Partnership.
- (i) Other. The General Partner and/or its Affiliates may receive additional compensation for any additional services performed on behalf of the Partnership so long as such services are provided on terms and conditions no less favorable to the Partnership than can be obtained from independent third parties for comparable services in the same location. Such services may include, but are not limited to, leasing coordination fees, construction management fees, including in connection with renovation and remodeling, and tax appeal fees.
- 5.10. <u>Liquidation of the Partnership and Redemption of the General Partner Investment</u>. The General Partner shall in good faith actively (i) cause all real property owned by the Partnership and not in the development or redevelopment stage to be marketed for sale on or before the last day of the Operating Period and (ii) cause all properties which are in the development or redevelopment stage at the end of the Operating Period to be marketed for sale upon completion. If the General Partner has not taken all commercially reasonable efforts to diligently pursue the marketing for sale of the Partnership's properties when required by the foregoing, the Partnership shall redeem the General Partner Investment for \$1.00 without any action on the part of the General Partner.
- 5.11. <u>Joint Ventures</u>. Subject to the provisions below, the General Partner may choose to acquire properties in joint ventures or other co-ownership arrangements with third party developers and real estate investors, including the General Partner, its Affiliates and entities owned or managed by Affiliates, with the purpose of acquiring and/or improving properties ("Joint Ventures"). The General Partner and/or its Affiliates may provide services to the Joint Venture, including, but not limited to, acquisition, development, management, leasing and/or real estate disposition services.
 - (a) Notwithstanding the foregoing, the Partnership shall not enter into a Joint Venture unless:

- 1. The Partnership has the right either to approve significant decisions of the Joint Venture or to control operations of the Joint Venture, subject to the right of the Joint Venture partner to approve sales or refinancing;
- 2. The total compensation paid by the Partnership and the Joint Venture to the General Partner and its Affiliates in connection with a Joint Venture would be permissible under this Agreement if the Partnership owned 100% of the Joint Venture;
- 3. There is no duplication of Joint Venture costs and expenses and the costs and expenses of the Partnership relating to the Joint Venture business, including organization and syndication expenses, acquisition and development costs; and
- 4. Any purchase, sale or financing transactions between the Joint Venture partner and the General Partner or its Affiliates are on terms which are commercially reasonable and comparable to those relating to transactions between unrelated parties.
- (b) If the Partnership enters into a Joint Venture with the General Partner, an Affiliate or an entity owned or managed by an Affiliate, the General Partner or the managing member, largest shareholder, general partner or other controlling or majority owner of the Affiliate or such other entity (a "Controlling Person") may contribute capital to the Joint Venture on the same terms and conditions as the Partnership. Allocable profits in a Joint Venture shall be calculated based on the sum of net sale proceeds from the sale of a property (after repayment of debt) plus reserves less capital contributions of each joint venturer plus actual origination and carrying costs of the additional financing incurred in connection with such property. Distributions will be pro rata to the joint venturers based on their aggregate capital contributions.
- (c) The General Partner or an Affiliate may form another partnership or other entity with essentially the same investment objectives as the Partnership and such entity may acquire properties through joint venture or other arrangements. Such other entities may have as investors Controlling Persons (as defined above) or other former and current investors in programs sponsored by Affiliates. The terms and conditions upon which persons become investors in such other entities may differ from those of the Partnership. The Partnership will not acquire properties jointly with other entities unless the ownership is in the form of a partnership, limited liability company, joint venture or other co-ownership arrangement.

5.12. Merger, Exchange and Conversion.

- (a) The Partnership may (1) adopt a plan of merger and may merger with one or more domestic or foreign limited partnerships or other entities, resulting in there being one or more surviving entities, (2) adopt a plan of exchange by which a domestic or foreign limited partnership or other entity is to acquire all of the outstanding Partnership interests of the Partnership in exchange for cash, securities or other property of the acquitting domestic or foreign limited partnership or other entity or (3) adopt a plan of consideration and convert to a foreign limited partnership or other entity. Any such plan of merger, plan of exchange or plan of conversion shall in all events comply with the applicable requirements of the Act and this Agreement.
- (b) Any such merger shall be conditioned upon the merger being permitted by the laws under which each other entity that is a party to the merger is incorporated or organized or by the constituent documents of such other entity that are not inconsistent with such laws. Any such plan of exchange shall be conditioned upon the issuance of shares or other interests of the acquiring foreign limited partnership or other entity being permitted by the laws under which such foreign limited partnership or other entity is incorporated or organized or is not inconsistent with such laws. Any such conversion shall be conditioned upon such conversion being permitted by, or not inconsistent with, the laws of the jurisdiction in which the converted entity is to be incorporated, formed or organized, and the incorporation, formation or organization of the converted entity is effected in compliance with such laws.
- (c) The Partnership may adopt a plan of merger, plan of exchange or plan of conversion if the General Partner acts upon and the Limited Partners (if required by Section 5.12(d) below) approve the plan of merger, plan of exchange or plan of conversion in the manner prescribed in Section 5.12(d) below.
- (d) Except as provided by Section 5.12(e) and 5.12(f), in order for a plan of merger, plan of exchange or plan of conversion to be approved:
- 1. the General Partner shall adopt a resolution recommending that the plan of merger, plan of exchange or plan of conversion, as the case may be, be approved by the Limited Partners, unless the General Partner determines that for any reason it should not make that recommendation, in which case the General Partner shall adopt a resolution directing that the plan of merger, plan of exchange or plan of conversion, as the case may be, be submitted to the Limited Partners for approval without recommendation and, in connection with the submission, communicate the basis for its determination that the plan of merger, plan of exchange or plan of conversion be submitted to the Limited Partners without any recommendation;
- 2. the General Partner shall then submit the plan of merger, plan of exchange or plan of conversion for approval by the Limited Partners; and
- 3. the Limited Partners entitled to vote on the plan of merger, plan of exchange or plan of conversion must approve the plan by a Majority Vote (unless the General Partner requires a greater vote).

The General Partner may condition its submission to the Limited Partners of a plan of merger, plan of exchange or plan of conversion, and the effectiveness of such plan, on any basis, including without limitation, that a specified percentage in excess of a Majority Vote be required for the approval of the plan of merger, plan of exchange or plan of conversion. Unless the General Partner requires a greater vote, a Majority Vote shall be required for approval of a plan of merger, plan of exchange or plan of conversion.

The General Partner shall notify each Limited Partner, whether or not entitled to vote, of the meeting of the Limited Partners at which the plan of merger, plan of exchange or plan of conversion is to be submitted for approval in accordance with Section 5.12(d) and applicable law. The notice shall be given at least ten (10) days before the meeting and shall state that the purpose, or one of the purposes, of the meeting is to consider the plan of merger, plan of exchange or plan of conversion and shall contain or be accompanied by a copy or summary of the plan of merger, plan of exchange or plan of conversion. Any such approval may be by written consent of the Limited Partners as would be required to approve the plan of merger, plan of exchange or plan of conversion at any meeting where all of the Limited Partners are present.

- (e) Unless applicable law otherwise requires (in which case the approval of the Limited Partners shall continue to be required and the provisions of Section 5.12(d) shall continue to apply), (1) approval by the Limited Partners of a plan of exchange shall not be required, and the provisions of Section 5.12(d) do not apply, if the Partnership is the acquiring entity in the plan of exchange and (2) approval by the Limited Partners on a plan of merger, plan of exchange or plan of conversion shall not be required and the provisions of Section 5.12(d) do not apply if either:
 - 1. (A) A limited partnership is the sole surviving or resulting entity;

- (B) the partnership agreement of the surviving or resulting limited partnership will not differ from this Agreement before the merger or conversion in any manner other than as to the application of applicable law or other insignificant conforming differences;
- (C) Limited Partners who held interests in the Partnership immediately before the effective date of the merger or conversion will hold the same interests in the same proportions, immediately after the effective date of the merger or conversion; and
- (D) the General Partner adopts a resolution approving the plan of merger or plan of conversion; or
- 2. T he transaction involves the conversion to corporate, trust or association form of only the Partnership and, as a consequence of the transaction, there will be no significant adverse change in any of the following rights or terms, as compared to such rights and terms in effect for the Partnership prior to such transaction:
 - A. voting rights of holders of the class of securities to be held by Limited Partners;
 - B. the term of existence of the surviving or resulting entity;
 - C. compensation to the sponsor of the surviving or resulting entity; or
 - D. the investment objectives of the surviving or resulting entity.
- (f) After a plan of merger, plan of exchange or plan of conversion is approved, and at any time before the merger, exchange or conversion has become effective, the plan of merger, plan of exchange or plan of conversion may be abandoned (subject to any contractual rights by any of the entities that are a party thereto), without action by the Limited Partners, in accordance with the procedures set forth in the plan of merger, plan of exchange or plan of conversion or, if no such procedures are set forth in such plan, in the manner determined by the General Partner.

DISTRIBUTIONS

6.1. Share of Distributions.

- (a) Partnership distributions of Net Cash during the Operating Period are targeted to be made to the Partners as follows: Distributions of Net Cash shall be made 1% to the General Partner and 99% to the Limited Partners. Distributions shall be paid monthly to each Partner in a an amount determined by the General Partner, in its sole discretion, to the extent Net Cash is available, in a target amount equal to the Applicable Percentage per annum uncompounded on such Partner's Unreturned Deemed Capital Contribution.
 - (b) Partnership distributions of Net Cash after the Operating Period shall be made to the Partners as follows:
 - 1. First, distributions will be made 100% to the Limited Partners (in proportion to their Unreturned Invested Capital) until such time as the Limited Partners have received cumulative distributions equal to 100% of their Unreturned Invested Capital;
 - 2. Second, distributions will be made 100% to the General Partner until such time as the General Partner has received cumulative distributions equal to 100% of its Capital Contributions;
 - 3. Third, distributions will be made 1% to the General Partner and 99% to the Limited Partners (in proportion to the number of Units held by each Limited Partner as compared to the aggregate Units held by all Limited Partners) until such time as the Limited Partners have received cumulative distributions from all sources (including monthly cash distributions made during the Operating Period) equal to an 8.5% Annual Return;
 - 4. Fourth, distributions will be made 100% to the General Partner until such time as the General Partner has received cumulative distributions (other than with respect to its Units) from all sources (including monthly cash distributions made during the Operating Period) in excess of cumulative distributions to date which constitute a return of capital contributed pursuant to Section 4.1 ("GP Excess Distributions") equal to 40% of all LP Excess Distributions made to Limited Partners pursuant to the provisions of this Section 6.1; and
 - 5. Thereafter, distributions will be made 60% to the Limited Partners (in proportion to the number of Units held by each Limited Partner as compared to the aggregate Units held by all Limited Partners) and 40% to the General Partner.
- 6.2. <u>Source of Funds.</u> In making distributions pursuant to Sections 6.1(a) and (b) above, no distinction shall be made by reason of the source of such funds.

6.3. Liquidation or Dissolution.

- (a) In the event the Partnership is liquidated or dissolved, the assets of the Partnership shall be distributed to the Partners in accordance with Section 8.4(d), but only after the payment of Partnership debts and the establishment of reasonable reserves for anticipated contingencies, if any.
- (b) No Partner with a negative balance in such Partner's Capital Account Balance, at any time, including following the distribution of liquidation proceeds, shall be required to restore such balance to the Partnership except as may be required under the Act.
- (c) Any Limited Partner who has a separate written obligation approved by the General Partner to make additional Capital Contributions to the Partnership which has not been paid at the date of liquidation shall pay such amount to the Partnership within ninety (90) days of the liquidation of the Partnership, or such obligation may be offset against any distributions to be made to such Limited Partner.
- 6.4. Amount and Time of Distributions. Distributions pursuant to this Article VI will be made at such times and in such amounts as the General Partner may, in its sole discretion, determine. The General Partner shall endeavor to distribute each Partnership Fiscal Year to each Limited Partner an amount at least equal to the Applicable Percentage per annum uncompounded on such Limited Partner's Unreturned Deemed Capital Contribution. The Partnership may be restricted from making distributions under the Act or the terms of notes, mortgages or other debt obligations which it may obtain from lenders in connection with Partnership loans. Distributions may also be restricted or suspended whenever the General Partner, in its sole discretion, determines that such action is in the best interests of the Partnership. All distributions are subject to the payment of Partnership expenses, including maintenance of reasonable operating reserves. Notwithstanding anything herein

to the contrary, in each Partnership Fiscal Year, the General Partner may, in its sole discretion, distribute to the Partners such amounts as the General Partner, in its sole discretion, determines are necessary to enable the Partners to pay the Federal income taxes on their distributive shares of the Partnership's taxable income (including separately stated items). The General Partner shall not incur any liability as a result of its determination to distribute Net Cash, even though such distribution may result in the Partnership's retaining insufficient funds for the operation of its business, provided its determination was made in good faith and not as a result of its gross negligence or willful misconduct.

PROFITS AND LOSSES

- 7.1. Capital Account. A separate capital account shall be maintained for each Partner (a "Capital Account"). Each Partner's Capital Account shall be increased by (1) the amount of money contributed by it to the Partnership, (2) the fair market value of any property contributed by it to the Partnership (net of any liabilities securing such contributed property that the Partnership is considered to assume or take subject to under Section 752 of the Code), and (3) allocations to it of income or gain (or items thereof), including income and gain exempt from tax, and income and gain described in Regulations Section 1.704-1(b)(2)(iv)(g), but excluding income and gain described in Regulations Section 1.704-1(b)(4)(i); and shall be decreased by (4) the amount of money distributed to it by the Partnership, (5) the fair market value of property distributed to it by the Partnership (net of liabilities securing such distributed property that such Partner is considered to assume or take subject to under Section 752 of the Code), (6) allocations to it of expenditures of the Partnership of the type described in Code Section 705(a)(2)(B), and (7) allocations of deduction and loss, including items of loss and deduction described in Regulations Section 1.704-1(b)(2)(iv)(g), but excluding items described in (6) above and loss or deduction described in Regulations Sections 1.704-1(b)(4)(i) or 1.704-1(b)(4)(iii). For purposes of this Agreement, a Partner who has more than one interest in the Partnership shall have a single Capital Account that reflects all such interests, regardless of the class of interests owned by such Partner, and regardless of the time or manner in which such interests were acquired. All items of income exempt from federal income tax, and expenditures not deductible in computing federal income tax shall be allocated among the Partners in accordance with their allocable share of income, gain, deduction or losses (as the case may be). If the book values of Partnership assets are adjusted, the Capital Accounts of all Partners shall be adjusted simultaneously to reflect the aggregate net adjustment as if the Partnership recognized gain or loss equal to the amount of such aggregate net adjustment. Subject to Section 8.4, if any assets of the Partnership are to be distributed in kind, such assets shall be distributed on the basis of their fair market value as determined by independent appraisal after the Partners' Capital Accounts have been adjusted to reflect the manner in which any unrealized gain and loss with respect to such assets (that have not been reflected in their Capital Accounts previously) would be allocated among the Partners if there were a taxable disposition of the property for its fair market value. It is the intent of the Partnership that the Capital Accounts of all Partners be determined and maintained in accordance with the principles of Regulations Section 1.704-1 at all times throughout the full term of the Partnership, and this Section 7.1 shall be so interpreted and applied.
- 7.2. <u>Deficit Capital Account Balances</u>. Upon liquidation of the Partnership or the liquidation of a Partner's interest in the Partnership, no Partner with a deficit balance in its Capital Account shall have the obligation to restore such deficit balance by making an additional contribution to the capital of the Partnership, except as may be required under the Act.

7.3. Allocations.

- (a) Allocations of Net Profit and Net Loss. After giving effect to the special allocations set forth in Sections 7.3(b), (c), (d), (e), (f), (g), (h), and (j), Net Profit and Net Loss for any taxable year or other applicable period shall be allocated as set forth below.
 - (i) Net Profit shall be allocated:
 - (A) First, 99% to the Limited Partners (in proportion to the number of Units held by each Limited Partner as compared to the aggregate Units held by all Limited Partners) and 1% to the General Partner, to the extent of the LP Target Distribution Amount;
- (B) Second, 100% to the General Partner until its Capital Account balance equals 40% of the Partners' aggregate Capital Account balance, taking into account any proposed distributions and allocations for the current taxable year or other applicable period; and
- (C) Thereafter, 60% to the Limited Partners (in proportion to the number of Units held by each Limited Partner as compared to the aggregate Units held by all Limited Partners), and 40% to the General Partner.
 - (ii) Net Loss shall be allocated:
 - (A) First, 60% to the Limited Partners (in proportion to the number of Units held by each Limited Partner as compared to the aggregate Units held by all Limited Partners) and 40% to the General Partner, until the cumulative amount of Net Loss allocated pursuant to this Section 7.3(a)(ii)(A) for the current taxable year or other applicable period and all previous taxable years or other applicable periods is equal to the cumulative amount of Net Profit allocated to each of the Partners pursuant to Section 7.3(a)(i)(C) for all previous taxable years or other applicable periods;
 - (B) Second, to the General Partner, until the aggregate amount of Net Loss allocated pursuant to this Section 7.3(a)(ii)(B) for the current taxable year or other applicable period and all previous taxable years or other applicable periods is equal

to the cumulative amount of Net Profit allocated to the General Partner pursuant to Section 7.3(a)(i)(B) for all previous taxable years or other applicable periods;

- (C) Thereafter, 99% to the Limited Partners (in proportion to the number of Units held by each Limited Partner as compared to the aggregate Units held by all Limited Partners) and 1% to the General Partner.
- (iii) Notwithstanding anything to the contrary in Section 7.3(a), no Partner shall be allocated any Net Loss for any taxable year or other applicable period to the extent that such allocation would result in such Partners having an Adjusted Capital Account Deficit as of the end of such taxable year or other applicable period and the other Partners having a positive Capital Account balance at the end of such taxable year or other applicable period. Any Net Loss not allocated to a Partner pursuant to the preceding sentence shall be allocated to the other Partners until their Capital Account balance has been reduced to zero, provided that, notwithstanding anything to the contrary in Section 7.3(a), any Net Profit for any subsequent taxable year or other applicable period that would have been allocated to a Partner to which Net Loss would have been allocated but for the effect of the first sentence of this Section 7.3(a)(iii) shall be allocated to the other Partners to the extent of the aggregate amount of Net Loss allocated to such other Partners pursuant to the first sentence of this Section 7.3(a)(iii).
- (b) Partnership Minimum Gain Chargeback. Notwithstanding any other provision of this Section 7.3, if there is a net decrease in Partnership Minimum Gain during any taxable year or other period for which allocations are made, prior to any other allocation under this Agreement, each Partner will be specially allocated items of Partnership income and gain for that period (and, if necessary, subsequent periods) in proportion to, and to the extent of, an amount equal to such Partner's share of the net decrease in Partnership Minimum Gain during such year determined in accordance with Regulations Section 1.704-2(g)(2). The items to be allocated will be determined in accordance with Regulations Section 1.704-2(g). This Section 7.3(b) is intended to comply with the partnership minimum gain chargeback requirements of the Regulations, will be interpreted consistently with the Regulations and will be subject to all exceptions provided therein.
- (c) Partner Nonrecourse Debt Minimum Gain Chargeback. Notwithstanding any other provision of this Section 7.3 (other than Section 7.3(b) which shall be applied first), if there is a net decrease in Partner Nonrecourse Debt Minimum Gain with respect to a Partner Nonrecourse Debt during any taxable year or other period for which allocations are made, any Partner with a share of such Partner Nonrecourse Debt Minimum Gain (determined under Regulations Section 1.704-2(i)(5)) as of the beginning of the year will be specially allocated items of Partnership income and gain for that period (and, if necessary, subsequent periods) in an amount equal to such Partner's share of the net decrease in the Partner Nonrecourse Debt Minimum Gain during such year determined in accordance with Regulations Section 1.704-2(g)(2). The items to be so allocated will be determined in accordance with Regulations Section 7.3(c) is intended to comply with the partner nonrecourse debt minimum gain chargeback requirements of the Regulations, will be interpreted consistently with the Regulations and will be subject to all exceptions provided therein.
- (d) <u>Qualified Income Offset</u>. A Partner who unexpectedly receives any adjustment, allocation or distribution described in Regulations Sections 1.704-1(b)(2)(ii)(d)(4), (5) or (6) will be specially allocated items of Partnership income and gain in an amount and manner sufficient to eliminate, to the extent required by the Regulations, the Adjusted Capital Account Deficit of the Partner as quickly as possible.
- (e) <u>Nonrecourse Deductions</u>. Nonrecourse Deductions for any taxable year or other period for which allocations are made will be allocated 1% to the General Partner and 99% to the Limited Partners, pro rata among them in proportion to the number of Units held by each Limited Partner as compared to the aggregate Units held by all Limited Partners.
- (f) <u>Partner Nonrecourse Deductions</u>. Notwithstanding anything to the contrary in this Agreement, any Partner Nonrecourse Deductions for any taxable year or other period for which allocations are made will be allocated to the Partner who bears the economic risk of loss with respect to the Partner Nonrecourse Debt to which the Partner Nonrecourse Deductions are attributable in accordance with Regulations Section 1.704-2(i).
- (g) <u>Code Section 754 Adjustments</u>. To the extent an adjustment to the adjusted tax basis of any Partnership asset under Code Sections 734(b) or 743(b) is required to be taken into account in determining Capital Accounts under Regulations Section 1.704-1(b)(2)(iv)(m), the amount of the adjustment to the Capital Accounts will be treated as an item of gain (if the adjustment increases the basis of the asset) or loss (if the adjustment decreases the basis), and the gain or loss will be specially allocated to the Partners in a manner consistent with the manner in which their Capital Accounts are required to be adjusted under Regulations Section 1.704-1(b)(2)(iv)(m).
- (h) <u>Interest in Partnership</u>. Notwithstanding any other provision of this Agreement except Section 7.12, no allocation of Net Profit, Net Loss or item of income, gain, loss or deduction will be made to a Partner if the allocation would not have "economic effect" under Regulations Section 1.704-1(b)(2)(ii) or otherwise would not be in accordance with the Partner's interest in the Partnership within the meaning of Regulations Section 1.704-1(b)(3) or 1.704-1(b)(4)(iv). The General Partner will have the authority to reallocate any item in accordance with this Section 7.3(h). Notwithstanding the authority granted to make such allocations as may be appropriate to accomplish the purposes of this Section 7.3(h), no allocation under this Section 7.3(h) will affect or otherwise alter the amount of any distribution to which a Partner would otherwise be entitled pursuant to any provision of this Agreement.

- (i) [Intentionally Left Blank]
- (j) <u>Curative Allocations</u>. The allocations set forth in subsection (b), (c), (d), (e), (f) and (g) (the "Regulatory Allocations") are intended to comply with certain requirements of Regulations Sections 1.704-1(b) and 1.704-2. The Regulatory Allocations may effect results that would not be consistent with the manner in which the Partners intend to divide Partnership distributions. Accordingly, the General Partner is authorized to divide other allocations of Net Profits, Net Losses, and other items among the Partners so as to prevent the Regulatory Allocations from distorting the manner in which Partnership distributions would be divided among the Partners under Section 8.4. In general, the reallocation will be accomplished by specially allocating other Net Profits, Net Losses and items of income, gain, loss and deduction, to the extent they exist, among the Partners so that the net amount of the Regulatory Allocations and the special allocations to each Partner is zero. The General Partner will have discretion to accomplish this result in any reasonable manner that is consistent with Code Section 704 and the related Regulations.
- 7.4. <u>Taxable Gain on Exchange</u>. Any taxable gain incurred by the Partnership as a result of receiving taxable "boot" in an otherwise tax-deferred exchange of a Partnership property shall be allocated to the Partners receiving said funds and other consideration as said allocation is provided in Code Section 704(a) and 704(b), but only to the extent that the total gain so allocated does not exceed the amount of gain that would be allocated to the Partner had the property been sold outright for all cash by the Partnership for the net amount credited to the Partnership in the exchange. Any such excess shall be allocated to the remaining Partners in the same ratio that total gain would have been allocated had the property been sold outright for all cash by the Partnership for the net amount credited to the Partnership in the exchange.
- 7.5. Depreciation Recapture. To the extent possible, if taxable gain to be allocated includes income resulting from the sale or disposition of Partnership property which is treated as ordinary income (or un-recaptured Code Section 1250 gain) for income tax purposes, such gain so treated as ordinary income (or un-recaptured Code Section 1250 gain) shall be allocated to and reported by each Partner in proportion to allocations received by that Partner of the items that gave rise to such ordinary income (or Code Section 1250 gain) recapture, and the Partnership shall keep records of such allocations.
- 7.6. Allocation between Assignor and Assignee. The initial Capital Account Balance for a transferee of Units pursuant to this Agreement shall be that of the transferor on the date the transfer is effective hereunder. The portion of the income, gain, losses, credits and deductions of the Partnership for any Partnership Fiscal Year during which a Partner transfers its interest in the Partnership pursuant to this Agreement is allocable to the transferor and the transferee on the basis of the number of days during the relevant fiscal year that each is the owner thereof, using a reasonable daily proration method.
- 7.7. Express Consent to Allocations. The allocations which are to be made pursuant to this Agreement, including allocations of distributions, are expressly consented to by each Partner and each Partner hereby agrees to file its tax returns consistent therewith.

Table of Contents

- 7.8. Adjustments to Gain and Loss. For purposes of computing gain or loss on the disposition of Partnership property, or for purposes of determining the cost recovery, depreciation or amortization deduction with respect to any asset, the Partnership shall use such property's book value, as determined in accordance with Regulations Section 1.704-1(b). Consequently, such property's book value shall be equal to its adjusted basis for federal income tax purposes, except as follows:
 - (a) The initial book value of the property shall be the fair market value net of encumbrances on such property.
- (b) The book values of all Partnership properties may be adjusted to equal their respective gross fair market values, as determined by the General Partner, as of the following times: (i) the acquisition of an additional interest in the Partnership by any new or existing Partner in exchange for more than a capital contribution of \$1,000; (ii) the distribution by the Partnership to a Partner of more than a de minimis amount of Partnership property other than money; (iii) the termination of the Partnership for federal income tax purposes pursuant to Code Section 708(b)(1)(B); and (iv) the liquidation of any Partner's Interest or portion thereof (other than a de minimis amount) in the Partnership.
- (c) If the book value of property has been determined pursuant to subsection 7.8(a) or 7.8(b), such book value shall thereafter be adjusted by depreciation or amortization, if any, taken into account with respect to such property for purposes of computing income, gain or loss.
- 7.9. <u>Tax Allocations: Code Section 704(c)</u>. In accordance with Code Section 704(c) and the Regulations, income, gain, loss and deductions with respect to any property contributed to the capital of the Partnership shall, solely for tax purposes, be allocated among the Partners so as to take into account any variation between the adjusted basis of such property to the Partnership for federal income tax purposes and its initial book value (computed in accordance with Section 7.8).

If the book value of any Partnership property is adjusted pursuant to Section 7.8, subsequent allocations of income, gain, loss and deductions with respect to such asset shall take into account any variation between the adjusted basis of such property for federal income tax purposes and its book value in the same manner as under Code Section 704(c) and the Regulations.

Any elections or other decisions relating to such allocations shall be made by the General Partner in any manner that reasonably reflects the purpose and intention of this Agreement. Allocations pursuant to this Section 7.9 are solely for purposes of federal, state and local taxes and shall not affect, or in any way be taken into account in computing, any Partner's Capital Account Balance.

7.10. General Provisions Relating to Distributions, Net Profit and Net Loss Allocations.

- (a) For each taxable year of the Partnership, all Net Profit and Net Loss of the Partnership shall be allocated at and as of the end of the taxable year. The General Partner shall distribute to each Partner a report of the allocations of Net Profit and Net Loss within ninety (90) days after the end of such taxable year (or later if due to a delay beyond the control of the General Partner).
- (b) Net Profit and Net Loss shall be allocated, and all distributions shall be distributed, as the case may be, to the Persons shown on the records of the Partnership to have been Partners as of the last day of the taxable year for which the allocation or distribution is to be made or if such distribution is to be made in connection with the dissolution of the Partnership, as of the date such distribution is made pursuant to Section 8.4(d). Notwithstanding the foregoing, if a Partner transfers all or any portion of his or her Partnership Interest to any Person who, during such taxable year, is admitted as a Substitute Limited Partner, the Net Profit and Net Loss shall be allocated between the transferor and the transferee on the basis of the number of days of the taxable year in which each was a Partner or assignee of record, as the case may be.
- (c) The methods set forth hereinabove by which Net Profit, Net Loss, and the distributions are allocated, apportioned and paid are hereby expressly consented to by each Partner as an express material condition to becoming a Partner.
- 7.11. Adjustments to Allocations. Anything herein to the contrary notwithstanding, except and subject to Section 7.12, in the event the General Partner in its sole discretion, based on advice from the Partnership's tax advisor, determines that for any Partnership tax year or other applicable period an allocation of Net Profit, Net Loss, gain, loss, deduction or credit is likely not to be in compliance with the Code or Regulations promulgated thereunder, the General Partner may, without limiting the authority of the General Partner otherwise under this Article VII, amend this Agreement, and each Limited Partner hereby consents to any such amendment, to reallocate such item so as to comply with the Code or Regulations in such manner as it determines, in its sole judgment, to least change the original allocation.
- 7.12. Fractions Rule. The Partners intend that the foregoing allocation provisions of this Article VII shall be made in such a manner as to have "substantial economic effect" pursuant to Regulations Section 1.704-1(b), and enable each Limited Partner which is a qualified organization under Section 514 of the Code ("Qualified Organization") to meet the requirements of Section 514(c)(9)(E) of the Code. To the extent such foregoing allocation provisions do not have "substantial economic effect," or cause a Limited Partner which is a Qualified

Organization to be unable to meet the requirements of Section 514(c)(9)(E) of the Code, such provisions shall be amended by the General Partner if and to the extent necessary to produce such result; provided, however, that no such adjustment that has a material adverse effect upon any Partner shall be made without the General Partner having received from counsel an opinion that such adjustments are necessary for a Limited Partner which is a Qualified Organization to comply with the requirements of Section 514(c)(9)(E) of the Code. This Section 7.12 shall control notwithstanding any other provision of this Agreement or the reallocation or adjustment of taxable income, taxable loss or items thereof by the Internal Revenue Service or any other taxing authority.

ARTICLE VIII

TERMINATION AND DISSOLUTION OF THE PARTNERSHIP

- 8.1. No Termination by Admission or Incapacity of Limited Partners. Neither the admission to the Partnership of any additional Limited Partner, nor the transfer of an Interest, nor any Event of Dissolution with respect to any of the Limited Partners will affect the Partnership's existence in any manner whatsoever. If an Event of Dissolution shall occur with respect to a Limited Partner, his Representative shall have the same rights for the purpose of settling his estate or business and shall be subject to the same limitations, conditions, and liabilities as applied to the Limited Partner; provided, however, that upon the death, dissolution or incapacity of a Limited Partner, the successor-in-interest to such Limited Partner shall have the right to become a Substitute Limited Partner as provided in Section 9.3 of this Agreement.
- 8.2. <u>Termination of the Partnership</u>. The Partnership shall be dissolved upon the happening of any of the following events, whichever shall first occur:
- (a) an Event of Dissolution with respect to any General Partner, unless the Partnership is continued in accordance with the provisions of Section 8.3 of this Agreement;
- (b) the sale, exchange, condemnation or other disposition of all the interests in real estate (including, without limitation, purchase money security interests and interest in joint ventures or other entities owning interests in real estate) unless the General Partner has determined to reinvest the proceeds consistent with the provisions of this Agreement; or
 - (c) upon the expiration of the term provided for in Section 1.4 of this Agreement.

8.3. Continuation of Partnership upon Certain Events.

- (a) Upon the occurrence of any event or events provided in Section 8.2(a) of this Agreement with respect to less than all the General Partner(s) (if there is then more than one General Partner), the remaining General Partner(s) shall immediately send Notice of such event to the Limited Partners, and such remaining General Partner(s) may then elect, within 30 days after the occurrence of such event, to continue the business of the Partnership in accordance with the terms of this Agreement; provided, however, that counsel to the Partnership determines that such continuation would not result in the Partnership being classified for federal income tax purposes as an association taxable as a corporation and not as a partnership. If the remaining General Partner does not so elect, the Limited Partners may Consent, within sixty (60) days after receipt of Notice of such event, to reconstitute the Partnership and continue its business in accordance with the terms of this Agreement and with the new General Partner or the remaining General Partner selected by the Limited Partners.
- (b) Upon the occurrence of any event or events provided in Section 8.2(a) of this Agreement with respect to a sole General Partner, the Limited Partners shall have the right to continue the business of the Partnership in accordance with the terms of this Agreement upon the selection, by a Majority in Interest of the Limited Partners within ninety (90) days of such occurrence, of a substitute General Partner and upon such substitute General Partner executing this Agreement and an amendment to the Certificate and agreeing to be bound by all of the terms and provisions hereof; provided, however, that counsel to the Partnership determines that such continuation would not result in the Partnership being classified for federal income tax purposes as an association taxable as a corporation and not as a partnership.

8.4. <u>Dissolution and Liquidation</u>.

- (a) Upon the termination of the Partnership, and absent any continuation of the Partnership pursuant to Section 8.3, the Partnership's affairs shall be wound up and the Partnership shall be dissolved as soon as practicable thereafter by the General Partner or its designee, or, if there is no General Partner then remaining, by such other person designated by a Majority in Interest of the Limited Partners (the "Liquidator"). In winding up the affairs of the Partnership, the Liquidator shall proceed to liquidate the assets of the Partnership in such manner as it shall determine, allowing a reasonable time thereafter to minimize losses attendant upon a liquidation.
- (b) Upon the liquidation of the Partnership's assets, the proceeds, if any, from such liquidation shall be applied and distributed first to the payment of the expenses or other charges incurred in connection with such liquidation; second, to the payment of all debts and liabilities of the Partnership including, without limitation, compensation owing to the General Partner or an Affiliate pursuant to Section 5.9; third, to the establishment of such reserves which the Liquidator shall deem reasonably necessary to provide for contingent and unforeseen liabilities or obligations of the Partnership; and fourth, to the Partners in the manner and order provided in subsection (d) below.
- (c) The General Partner shall not be personally liable for the repayment of the Capital Contributions made by the Limited Partners or any portion thereof. Any repayment will be made solely from the assets of the Partnership available for such repayment. The

General Partner shall not be liable to the Partnership on account of a Capital Account Deficit of any Partner; provided, however, that the General Partner shall not be relieved of any obligation to pay liabilities to third parties incurred in its capacity as General Partner.

- (d) Distributions in connection with the dissolution of the Partnership shall be made in accordance with the Partners' respective positive Capital Accounts after they have been adjusted to reflect the allocations of Net Profits, Net Losses, income, gain, losses and deductions as set forth in Section 7.3. In the event that all or any part of the Partnership's property is itself distributed to the Partners on dissolution and termination of the Partnership without a sale of such property having occurred, an independent appraisal of the fair market value of each such asset as of a date reasonably close to the date of the Partnership's liquidation shall be obtained by the Liquidator at the expense of the Partnership. Any unrealized appreciation or depreciation with respect to such assets shall be allocated among the Partners in accordance with Section 7.3, assuming the assets were sold for their appraised value. Distribution of any such asset in kind to a Partner shall be considered a distribution of an amount equal to the asset's appraised fair market value.
- (e) Final Capital Account Balances. The parties intend that the allocation provisions of Section 7.3 shall produce final Capital Account balances of the Partners that will permit liquidating distributions under Section 8.4(d) to be made in a manner identical to the order of priorities set forth in Section 6.1(b). To the extent that the allocation provisions of Section 7.3 would fail to produce such final Capital Account balances, (a) such provisions shall be amended by the General Partner if and to the extent necessary to produce such result, and (b) profits and losses of the Partnership for prior open years (or items of gross income, gain, loss and deduction of the Partnership for such years) shall be reallocated by the General Partner among the Partners to the extent it is not possible to achieve such result with allocations of items of income (including gross income and gain), deduction and loss for the current year and future years. This Section 8.4(e) shall control notwithstanding any other provision of this Agreement, other than Section 7.12 and the reallocation or adjustment of taxable income, taxable loss or items thereof by the Internal Revenue Service or any other taxing authority.

ARTICLE IX

WITHDRAWAL AND TRANSFER BY PARTNERS

9.1. <u>Voluntary Withdrawal or Assignment by the General Partner</u>. The General Partner shall not resign, retire or withdraw as a general partner from the Partnership, or at any time assign, transfer, or otherwise dispose of all or any part of its Partnership Interest, unless: (i) any remaining General Partner and a Majority in Interest of the Limited Partners shall have Consented; (ii) such General Partner shall have provided an additional or successor General Partner satisfactory to a Majority in Interest of the Limited Partners; and (iii) the Partnership shall have received advice of its counsel to the effect that such resignation, withdrawal, assignment or transfer would not subject the Partnership to federal income taxation as an association taxable as a corporation and not as a partnership, and would not cause a termination of the Partnership for federal income tax purposes.

9.2. <u>Involuntary Withdrawal and Transfer of a General Partner's Interest.</u>

- (a) The General Partner may be removed, with or without cause, by an affirmative vote or written consent of the Limited Partners owning at least a majority of the Units then outstanding (not counting any Units then owned by the General Partner or its Affiliates). Limited Partners owning at least a majority of the Units then outstanding (not counting any Units then owned by the General Partner or its Affiliates) shall provide Notice to the General Partner of such removal which such Notice shall include (1) the Limited Partners voting for removal and the Limited Partners voting against removal, (2) the reason for such removal and (3) whether or not such removal was for cause. For the purposes of the foregoing, "cause" shall mean (i) the bankruptcy of the General Partner pursuant to an Event of Bankruptcy, (ii) the incapacity of the General Partner, (iii) the commission by the General Partner of an intentional material breach of this Agreement, or (iv) conduct of the General Partner constituting gross negligence or willful or wanton misconduct harmful to the Partnership.
- (b) In the event that the Limited Partners vote to remove the General Partner, the General Partner shall have an additional thirty (30) business days from the date of the notice of removal from the Limited Partners in which to cure or otherwise resolve the matter(s) resulting in the notice to remove to the reasonable satisfaction of the Limited Partners holding a majority of the Units then outstanding (not counting any Units then owned by the General Partner). The Limited Partners shall not unreasonably withhold approval of the cure accomplished by the General Partner. In the event that the General Partner is removed, it hereby agrees to convert its General Partner's Capital Contribution into that of a Limited Partner having no voting rights but whose Interest is unchanged in all other respects. The removed General Partner shall be removed from and indemnified against Partnership liabilities, including without limitation liabilities the General Partner has guaranteed. Further, if removed other than for cause, the General Partner's Interest must be purchased by the Partnership and/or any successor General Partner for fair market value, which will be determined by the median of three independent appraisals of the General Partner Interest.
- (c) In the event of the dissolution and liquidation or removal of the General Partner and the Partnership being continued in accordance with Section 8.3, the successor of the said General Partner shall not be entitled to any distributions due the former General Partner under this Agreement. The Limited Partners hereby agree to such a transfer and specifically authorize that the power of attorney provided for in their Subscription Agreements with the Partnership be utilized, if need be, on their behalf to effectuate fully the legal conversion or transfer, as the case may be, of the dissolved or removed General Partner's Interest hereunder.

9.3. <u>Transfer by Limited Partners</u>.

- (a) Subject to Section 9.4 hereof, no Limited Partner may at any time sell, transfer, or assign all or any portion of its Interest in the Partnership, unless:
- 1. such Limited Partner and the purchaser, transferee or assignee execute, acknowledge and deliver to the General Partner such instruments of transfer and assignment with respect to such transaction as may be reasonably requested by the General Partner to assure the payment of any unpaid amounts due from the Limited Partner hereunder;
- 2. such Limited Partner obtains the prior written Consent of the General Partner, the granting of which shall be within the sole discretion of the General Partner and will not be given unless (i) the General Partner determines, in its sole discretion, that the Partnership (A) would be able to satisfy any of the secondary market safe harbors contained in Regulations Section 1.7704-1 (or any other applicable safe harbor from publicly traded partnership status which may be adopted by the Internal Revenue Service) for the Partnership's taxable year in which such transfer otherwise would be effective and (B) would not be treated as an association taxable as a corporation as a result of such transfer or assignment, or (ii) the Partnership has received an opinion of counsel satisfactory to the General Partner or a favorable Internal Revenue Service ruling that any such transfer will not result in the Partnership's being classified as a publicly traded partnership for federal income tax purposes (the General Partner may waive the requirement of the legal opinion in its sole discretion); and

3. such Limited Partner pays the Partnership the actual costs reasonably incurred by it in effecting the transfer or assignment, which shall not be less than \$150 per transaction.

In connection with Section 9.3(a)(2), the Limited Partners agree to provide all information with respect to a proposed transfer that the General Partner deems necessary or desirable in order to make such determination, including but not limited to, information as to whether the transfer occurred on a secondary market (or the substantial equivalent thereof).

- (b) Notwithstanding anything contained in this Agreement to the contrary, no purchaser, transferee or assignee of an Interest in the Partnership shall have any right to become a Substitute Limited Partner unless the General Partner consents in writing to such substitution. Further, notwithstanding anything contained in this Agreement to the contrary, no sale, transfer, assignment or substitution by a Limited Partner shall be effective as against the Partnership (and any such purported sale, transfer, assignment or substitution shall be void ab initio) if it (i) would result in the Partnership being treated as an association taxable as a corporation (and any such purported sale, transfer, assignment or substitution shall be void ab initio) or would otherwise cause the Partnership to fail to satisfy any of the applicable safe harbors from publicly traded partnership status contained in Regulations Section 1.7704-1 (or any other applicable safe harbor from publicly traded partnership status which may be adopted by the Internal Revenue Service), (ii) would violate any applicable federal or state securities, real estate syndication or comparable laws or (iii) would cause the Units to fail to fall within the insignificant investment exclusion from the definition of plan assets contained in Section 2510.3-101(f) of Title 29 of the Code of Federal Regulations.
- (c) No sale, transfer, assignment or substitution by a Limited Partner, which has otherwise been consented to by the General Partner, shall be effective as against the Partnership until the purchaser, transferee, assignee or Substitute Limited Partner, and all the Partners perform all such acts which the General Partner deems necessary or appropriate to constitute such purchaser, transferee or assignee as a Substitute Limited Partner and to preserve the limited liability status of the Limited Partners in the Partnership after the completion of such sale, transfer, assignment or substitution under the laws of each jurisdiction in which the Partnership is doing business. The Limited Partners agree upon request of the General Partner to execute such certificates or other documents and perform such other acts as may be reasonably requested by the General Partner in this regard.
- (d) Any sale, transfer or assignment of an Interest in the Partnership or substitution of a Limited Partner made in compliance with this Section 9.3 shall be effective as of the day of the month in which the execution of such instruments, certificates, or other documents and the performance of such other acts by the Partner is completed as provided in subsections (a) and (c) of this Section 9.3, or in which any required written Consent thereto is given by the General Partner pursuant to subsections (a) and (b) of this Section 9.3, whichever is later.
- (e) The Net Profits and Net Losses attributable to an assigned Interest in the Partnership shall be allocated among the assignor and assignee of such Interest as of the effective date of the assignment thereof, in accordance with Section 7.6 hereof.
- (f) Upon the death, dissolution or incapacity of a Limited Partner, the successor-in-interest to the Limited Partner shall have the right to become a Substitute Limited Partner upon written notice to the Partnership within ninety (90) days after the appointment of such Limited Partner's Representative, but not later than one hundred eighty (180) days after the death or certified incapacity of the Limited Partner, and upon such successor executing this Agreement or an amendment hereto, the certificate and such other documents as the General Partner may request. If such right is not exercised, the Representative of the deceased or incapacitated Limited Partner shall have only the rights to distributions and allocations as the Limited Partner would have had.
- (g) Notwithstanding anything contained in this Agreement to the contrary, the General Partner may require, as a condition to any transfer or assignment, a prior written notice of assignment of Units in form and substance satisfactory to legal counsel, signed by both the transferor and transferee, and a determination by counsel in form and substance satisfactory to the Partnership that (i) registration will not be required under the Securities Act of 1933, as amended; (ii) such assignment or transfer will not violate any applicable federal or state securities, real estate syndication or comparable laws; (iii) such assignment or transfer will not cause a termination of the Partnership for federal income tax purposes; (iv) such assignment or transfer will not be effectuated through an "established securities market" or a "secondary market (or the substantial equivalent thereof)" within the meaning of Section 7704(b) of the Code; (v) such assignment or transfer will not result in the Partnership being treated as a publicly traded partnership within the meaning of Section 7704 of the Code; and (vi) such assignment or transfer will not violate any loan documents or other agreements to which the Partnership is a party or by which the Partnership is bound.
- 9.4. Put Right. At any time after November 15, 2009 and prior to November 15, 2013, any Partner who has held Units for at least three years (the "Put Partner") shall have the right to request that the Partnership redeem all of such Units. Such request shall be made in writing, state a requested date for the redemption (the "Requested Redemption Date") and be delivered to the General Partner at least 60 calendar days in advance of the Requested Redemption Date. The General Partner shall determine whether the Partnership has sufficient funds to grant the request, which determination shall be made prior to the Requested Redemption Date in the sole discretion of the General Partner. If the General Partner determines that sufficient funds are available, the request shall be granted, and the Partnership shall transfer and deliver to the Put Partner no sooner than the Requested Redemption Date, but no later than 60 calendar days thereafter, 92% of the Unreturned Invested Capital of the Put Partner with respect to the redeemed Units determined as of the Requested Redemption Date; provided that the sum of the percentage interests in Partnership capital or profits transferred during the taxable year of the Partnership does not exceed 9% of the total

interests in partnership capital or profits as determined in the sole discretion of the General Partner. Notwithstanding the foregoing, at no time during any 12-month period may the number of Units redeemed by the Partnership exceed 2% of the number of Units outstanding at the beginning of such 12-month period unless such redemption is otherwise deemed to be a disregarded transfer for purposes of determining whether the Partnership is a publicly traded partnership pursuant to Regulations Section 1.7704-1 as determined in the sole discretion of the General Partner. If the General Partner determines that sufficient funds are not available, or if the requested redemption would cause the number of Units redeemed by the Partnership to exceed 2% of the number of Units outstanding at the beginning of such 12-month period, the Partnership shall either (i) decline to perform the requested redemption or (ii) perform the requested redemption solely to the extent such redemption does not violate the provisions of Section 9.3 or this Section 9.4, to be decided in the sole discretion of the General Partner.

Each Put Partner covenants and agrees with the Partnership and the General Partner that all Units delivered in connection with the exercise of the put right under this Section 9.4 shall be delivered to the Partnership or the General Partner, respectively, free and clear of all liens, encumbrances, liabilities, claims or charges of any kind and, notwithstanding anything contained herein to the contrary, neither the Partnership nor the General Partner shall be under any obligation to acquire any Put Partner's Units, (1) to the extent that any such Units are subject to any liens, encumbrances, liabilities, claims or charges of any kind or (2) in the event that any such Put Partner shall fail to give the General Partner adequate assurances that such Units are not subject to any such liens, encumbrances, liabilities, claims or charges of any kind or shall fail to agree to fully indemnify the General Partner from any such liens, encumbrances, liabilities, claims or charges of any kind as well as any costs and expenses relating to the Put Partner's Units or the exercise of the put right. Each Put Partner further agrees that, in the event any state or local transfer tax is payable as a result of the transfer of its Units to the Partnership or General Partner, respectively, each such Put Partner shall assume and pay such transfer tax.

ARTICLE X

BOOKS, RECORDS AND ACCOUNTING

10.1. Books and Records.

- (a) The Partnership shall keep and maintain the following records in its principal office or make them available in that office within five business days after the date of receipt of a written request under subsection (d) below:
 - 1. a current list that states:
- A. the name and mailing address of each partner, separately identifying in alphabetical order the General Partner and the Limited Partners:
 - B. the last known street address of the business or residence of the General Partner; and
 - C. the percentage or other interest in the Partnership owned by each Partner;
- 2. copies of the Partnership's federal, state, and local information or income tax returns for each of the Partnership's six most recent tax years;
 - 3. copies of the Agreement and the Certificate, and all amendments thereof;
 - 4. unless contained in this Agreement, a written statement of:
- A. the amount of the cash contribution and a description and statement of the agreed value of any other contribution made by each Partner, and the amount of the cash contribution and a description and statement of the agreed value of any other contribution that the Partner has agreed to make in the future as an additional contribution;
 - B. events requiring the Partnership to be dissolved and its affairs wound up;
 - C. the date on which each Partner in the Partnership became a Partner; and
 - D. books and records of account of the Limited Partners.
- (b) The Partnership shall maintain its records in written form or in another form capable of conversion into written form within a reasonable time.
- (c) The Partnership shall keep in its registered office in Delaware, and make available to Partners on reasonable request, the street address of its principal United States office in which the records required by this section are maintained or will be available.
- (d) A Partner or an assignee of a Partnership Interest, on written request stating the purpose, may examine and copy, in person or by the Partner's or assignee's representative, at any reasonable time, for any proper purpose, and at the Partner's expense, records required to be kept under this section and other information regarding the business, affairs, and financial condition of the Partnership as is just and reasonable for the person to examine and copy.
- (e) On the written request by any Partner or an assignee of a Partnership Interest made to the person and address designated in this Agreement or, if there is no designation, to the General Partner at the Partnership's principal United States office, the Partnership shall provide to the requesting Partner or assignee without charge, true copies of:
 - 1. this Agreement and all amendments or restatements;
 - 2. the Certificate, and all amendments or restatements; and
 - 3. any of the tax returns described in Subdivision (2) of Subsection (a) above.
 - 10.2. Accounting Method; Audits and Reports.

- (a) The Partnership shall adopt the accrual method of accounting unless the General Partner determines, in the interest of the Partnership, that a different method should be adopted.
- (b) At all times during the existence of the Partnership, the General Partner shall keep or cause to be kept full and true books of account in which shall be entered fully and accurately each transaction of the Partnership. Unless otherwise required by applicable law, the General Partner shall deliver to the Limited Partners within one hundred twenty (120) days after the end of each Partnership Fiscal Year unaudited annual financial statements of the Partnership. Such financial statements may be prepared by the General Partner or its Affiliates. Such financial statements will be audited for a Partnership Fiscal Year by an independent certified accountant if the affirmative vote or written consent of the Limited Partners owning at least 30% of the then outstanding Units (not counting any Units then owned by the General Partner) is obtained at least 90 days prior to the end of such Partnership Fiscal Year. If such financial statements are audited by an independent certified public accountant, the Partnership will bear 100% of the cost of such services. In addition, the General Partner shall deliver within ninety (90) days after the end of each Partnership Fiscal Year so-called "Information Returns" showing the actual Profits or losses and the allocation thereof to each Partner for the Partnership's preceding Partnership Fiscal Year.
- (c) The General Partner shall deliver to the Limited Partners periodically, but not less than annually, operating reports and financial or other pertinent information as is determined appropriate by the General Partner.
- 10.3. <u>Bank Accounts</u>. The General Partner shall maintain on behalf of the Partnership a bank account or accounts with such depositories as it shall determine, in which all monies received by or on behalf of the Partnership shall be deposited. All withdrawals from such accounts shall be made upon the signature of such person or persons as the General Partner may from time to time designate.

ARTICLE XI

SPECIAL TAX MATTERS

- 11.1. <u>Tax Elections</u>. All elections required or permitted to be made by the Partnership under the Code shall be made by the General Partner in the manner as will, in its opinion, be most advantageous to the Partners. No Partner shall take or refuse to take any action which would cause the Partnership to lose the benefits of any tax election made or to be made. The General Partner shall designate the "Tax Matters Partner" and the "Notice Partner" for the Partnership within the meaning of the Code. For all purposes with respect to the Code, the General Partner appoints itself the "Tax Matters Partner" and the "Notice Partner," as applicable, to serve in such capacity until its resignation or replacement is appointed.
- 11.2. <u>Tax Audit Expenses</u>. In the event the Partnership is audited by the Internal Revenue Service, the Partnership shall pay all of the costs incurred in defending the Partnership's tax position, including the cost of third party professionals engaged to assist in preparing for and representing the Partnership in such an audit. In the event the Partnership does not have funds available to pay such costs, the General Partner may advance such sums or may loan sufficient funds to the Partnership to cover such costs, in accordance with Section 5.4 herein. The General Partner shall be solely responsible for determining when and to what extent to compromise, defend, or otherwise pursue the Partnership's position with respect to the treatment of any item of Partnership income, expense, deduction, or allocation with respect to such an audit.

ARTICLE XII

AMENDMENTS AND MEETINGS

12.1. General.

- (a) The General Partner may, at any time, amend this Agreement in a way that does not adversely impact the rights of the Limited Partners; provided, however, that the General Partner shall provide written notice to the Limited Partners of such amendment no later than fifteen (15) days after the effective date of such amendment.
- (b) The General Partner may, at any time, propose an amendment to this Agreement that adversely impacts the rights of the Limited Partners by providing notice thereof to all Partners in writing, together with a statement of the purpose(s) of the amendment and such other matters as the General Partner deems material to the consideration of such amendment. An amendment to this Agreement proposed by the General Partner and adversely impacting the rights of the Limited Partners shall require the Consent of a Majority in Interest of the Limited Partners and such approval may be obtained at either a special meeting of Partners called for the purposes of considering such amendment or pursuant to the terms of Section 12.2. A special meeting called for the consideration of an amendment to this Agreement proposed by the General Partner shall be called by the General Partner by providing written notice to all Partners not less than ten (10) not more than sixty (60) days before the meeting. Included in the notice shall be a description of the proposed amendment.
- (c) At any time, Limited Partners holding not less than 25% of the issued and outstanding Units may propose an amendment to this Agreement, or a meeting of Limited Partners to consider any other proposal for which the Limited Partners may vote hereunder, including the sale of all or substantially all of the assets of the Partnership. Upon the request in writing to the General Partner of any person entitled to call a meeting, the General Partner shall promptly call a special meeting of all Partners to consider the proposal by providing written notice of the meeting to all Partners not less than ten (10) nor more than sixty (60) days before the meeting. Included in the notice shall be a description of the action proposed. The notice shall provide that Limited Partners may vote in person or by proxy. Any amendment to the Agreement properly proposed by Limited Partners shall require the Consent of a Majority in Interest of the Limited Partners; provided, however, that no such amendment affecting the rights or obligations of the General Partner shall be approved without the approval of the General Partner.
- 12.2. <u>Action Without a Meeting</u>. As an alternative to voting at meetings of the Partnership pursuant to this and other Articles of this Agreement, the Limited Partners may Consent to and approve by written action any matter that the Limited Partners may Consent to and approve by vote at a meeting. In order to Consent to and approve the matter, the same percentage of Limited Partners, by interest, must sign the written action as is required by vote at a meeting; provided, however, that written notice of the action so taken is given to all Partners at least fifteen (15) days after such required Consent is obtained.

12.3. Amendments on Admission or Withdrawal of Partners and Clarification Amendments.

- (a) Notwithstanding the provisions of Section 12.1 hereof, this Agreement may be amended without the Consent of the Limited Partners to reflect the addition, removal or substitution of a Limited Partner. Any amendment to this Agreement pursuant to this Section 12.3(a) shall be signed by the General Partner, and by the Persons to be added or substituted, or their attorneys-in-fact, and, if Limited Partners are to be removed or substituted, by the removed or assigning Limited Partners or their attorneys-in-fact.
- (b) Notwithstanding the provisions of Section 12.1 hereof, this Agreement and the Certificate may be amended without the Consent of the Limited Partners to reflect the designation of an additional or successor General Partner. Any amendment to this Agreement pursuant to this Section 12.3(b) shall be signed by the remaining General Partner(s), and by such additional or successor General Partner.
- (c) Notwithstanding the provisions of Section 12.1 hereof, this Agreement and the Certificate may be amended without the Consent of the Limited Partners to reflect the withdrawal or removal of a General Partner. Any amendment to this Agreement pursuant to this Section 12.3(c) shall be signed by the withdrawing or removed General Partners and the remaining or successor General Partner.
- (d) Notwithstanding the provisions of Section 12.1 hereof, this Agreement may be amended by the General Partner without the Consent of the Limited Partners to cure any ambiguity, to correct or supplement any provision herein which may be inconsistent with any other provisions herein, or to make any other provisions with respect to matters or questions arising under this Agreement which will not be inconsistent with the provisions of this Agreement. Any amendment to this Agreement pursuant to this Section 12.3(d) shall be signed by the General Partner.
- (e) Notwithstanding the provisions of Section 12.1 hereof, this Agreement may be amended by the General Partner without the Consent of the Limited Partners to add to the representations, duties or obligations of the General Partner or surrender any right or power

	~17~		

ARTICLE XIII

MISCELLANEOUS

- 13.1. <u>Notices</u>. All Notices, demands, or other communications hereunder shall be in writing and shall be deemed to have been given when the same are (a) deposited in the United States mail and sent by certified or registered mail, postage prepaid, return receipt requested, or (b) delivered, in each case to the parties at the addresses set forth below or at such other addresses as such parties may designate by written notice to the Partnership:
 - (a) if to the Partnership or the General Partner, at the principal office of the Partnership; and
- (b) if to the Limited Partners, at the addresses set forth in their respective Subscription Agreements or any written notice of change of address received by the General Partner.
- 13.2. <u>Further Assurances</u>. The Partners will execute and deliver such further instruments and do such further acts or forbear from such action and things as may be required to carry out the intent and purpose of this Agreement.
- 13.3. Representations of the Partners. Each Partner represents and warrants to the Partnership and every other Partner that (a) he is fully aware of, and is capable of bearing, the risks relating to an investment in the Partnership, (b) he understands his Interest in the Partnership has not been registered under the Securities Act of 1933, as amended, or the securities law of any jurisdiction in reliance upon exemptions contained in those laws, and (c) he has acquired his Interest in the Partnership for his own account, with the intention of holding the Interest for investment and without any intention of participating directly or indirectly in any redistribution or resale of any portion of the Interest in violation of the Securities Act of 1933, as amended, or any applicable law.
- 13.4. <u>Survival of Representations and Agreements</u>. All representations and agreements herein shall survive until the dissolution and final liquidation of the Partnership, except to the extent that a representation or agreement expressly provides otherwise.
- 13.5. <u>Agreement in Counterparts</u>. This Agreement may be executed in counterparts and all so executed shall constitute one agreement binding on all parties hereto notwithstanding that all the parties hereto are not signatories to the original or to the same counterpart.
- 13.6. <u>Captions and Context</u>. Captions contained in this Agreement are inserted only as a matter of convenience and in no way define, limit, extend or describe the scope of this Agreement or the intent of any provisions hereof. The use of a masculine pronoun shall be deemed interchangeable with a feminine or neuter pronoun as the content would require; likewise singular nouns shall be interchangeable with plural as the content requires.
- 13.7. <u>Construction</u>. The title of the Articles and Sections in this Agreement have been inserted as a matter of convenience of reference only and do not affect the meaning or construction of any of the provisions in this Agreement.
- 13.8. <u>Governing Law; Successors.</u> This Agreement and the rights and obligations of the Partners shall be governed by and construed in accordance with the internal laws of the State of Delaware. Except as otherwise expressly provided in this Agreement, all provisions of this Agreement shall be binding upon, inure to the benefit of, and be enforceable by or against, the heirs, successors, legal representatives and assigns of the parties hereto.
- 13.9. <u>Binding Effect</u>. Subject to Section 13.11, this Agreement will inure to the benefit of and will be binding upon the Partners, their legal representatives, transferees, heirs, administrators, successors and assigns.
- 13.10. <u>Validity of Agreement</u>. The invalidity of any portion of this Agreement shall not affect the validity of the remainder hereof. Should any portion of this Agreement be declared invalid and unenforceable, then such provisions shall automatically be struck from this Agreement and replaced with a provision as close in meaning to the original provision as possible while remaining valid and enforceable.
- 13.11. <u>Creditors Not Benefited</u>. Nothing in this Agreement (including specifically Sections 4.4, 5.4, 6.3 and 11.2) is intended to benefit any creditor of (i) the Partnership or (ii) a Partner. No creditor of the Partnership or a Partner will be entitled to require the General Partner to solicit or accept any loan or additional capital contribution for the Partnership or to enforce any right which the Partnership or any Partner may have against a Partner, whether arising under this Agreement or otherwise.

[Signatures on next page]

Table of Contents

IN WITNESS WHEREOF, the parties hereto have executed and delivered this Agreement to be effective as of the 15th day of November, 2006.

General Partner:

AmREIT Monthly Income & Growth IV Corporation

By: /s/ Chad C. Braun

Name: Chad C. Braun

Title: Vice President, Chief Financial Officer, Secretary and Treasurer

Initial Limited

Partner:

AmREIT

By: /s/ H. Kerr Taylor

Name: H. Kerr Taylor

Title: Chairman, Chief Executive Officer and President

AMREIT MONTHLY INCOME & GROWTH FUND IV, L.P.

AMENDMENT NO. 1 TO AGREEMENT OF LIMITED PARTNERSHIP

This AMENDMENT NO. 1 TO AGREEMENT OF LIMITED PARTNERSHIP (this "<u>Amendment</u>") dated as of December 7, 2006, is entered into by and between AmREIT Monthly Income & Growth IV Corporation (the "<u>General Partner</u>" and, in its capacity as a holder of Units, a "<u>Limited Partner</u>") and AmREIT (the "<u>Initial Limited Partner</u>" and, in its capacity as a holder of Units, a "<u>Limited Partner</u>").

WITNESSETH:

WHEREAS, the General Partner and the Initial Limited Partner entered into that certain Agreement of Limited Partnership on November 15, 2006 (the "<u>Agreement</u>"), relating to the management of AmREIT Monthly Income & Growth Fund IV, L.P. as more specifically provided therein;

WHEREAS, the General Partner has purchased Units, thereby becoming a Limited Partner; and

WHEREAS, pursuant to and in accordance with Article XII of the Agreement, the parties wish to amend the Agreement as set forth in this Amendment.

NOW THEREFORE, in consideration of the rights and obligations contained herein, and for other good and valuable consideration, the adequacy of which is hereby acknowledged, the parties agree as follows:

1. Defined Terms; References.

Capitalized terms used but not defined herein shall have the meanings assigned to such terms in the Agreement. Each reference to "hereof," "hereunder," "herein" and "hereby" and each other similar reference and each reference to "this Agreement" and each other similar reference contained in the Agreement shall, after the date hereof, refer to the Agreement as amended hereby.

2. Waiver.

For the purposes of this Amendment, the Limited Partners irrevocably waive the requirements set forth in Section 12.1(b) of the Agreement, including without limitation, the requirement of notice and statement of purpose of the amendment to be delivered to the Limited Partners by the General Partner.

3. Amendments.

2.1 The following defined term shall be added to Article II immediately following the defined term "Capital Contribution":

"'Casa Linda Loan' shall mean a loan to the Partnership to facilitate the Partnership's acquisition of equity interests in a joint venture formed to acquire Casa Linda Plaza located at North Bruckner Blvd and Garland Road, Dallas, Texas 75218."

2.2 Section 5.3(a)5. of the Agreement is hereby amended and replaced in its entirety with the following:

"other than the Casa Linda Loan, incur indebtedness on behalf of the Partnership in an amount greater than 75% of the Partnership's cash and cash equivalents plus the aggregate market value of the Projects, which market values shall be determined by the General Partner based on a capitalization rate approval applied to the net operating income of the Projects;"

3. <u>Titles and Headings</u>.

The headings in this Amendment are for reference purposes only, and shall not in any way affect the meaning or interpretation of this Amendment.

4. <u>Severability</u>.

The invalidity of any portion of this Amendment shall not affect the validity, force or effect of the remaining portions hereof. If it is ever held that any restriction hereunder is too broad to permit enforcement of such restriction to its fullest extent, such restriction shall be enforced to the maximum extent permitted by law.

5. <u>Counterparts and Recognition of Facsimile Signatures.</u>

This Amendment may be executed in one or more counterparts, each of which shall be deemed an original agreement, but all of which together shall constitute one and the same instrument. Additionally, the parties acknowledge and agree that a facsimile signature to this Agreement will be recognized and accepted as an original signature.

6. <u>Governing Law.</u>

The parties agree that this Amendment shall be governed by the provisions of Section 13.8 of the Agreement.

IN WITNESS WHEREOF, the parties hereto have executed and delivered this Amendment No. 1 to Agreement of Limited Partnership as of the date first written above.

AMREIT MONTHLY INCOME & GROWTH IV CORPORATION

By: /s/ Chad C. Braun

Chad C. Braun

Vice President, Chief Financial Officer, Secretary and Treasurer

AmREIT

By: /s/ H. Kerr Taylor

H. Kerr Taylor

Chairman, Chief Executive Officer and President

PROMISSORY NOTE DATE DECEMBER 8, 2006

Exhibit 10.10

MS Loan No. 06-29049

PROMISSORY NOTE

(Fixed — Defeasance or Yield Maintenance)

\$37,950,000.00

Dallas, Texas December 8, 2006

FOR VALUE RECEIVED, AMREIT CASA LINDA, LP, a Texas limited partnership, as maker, having its principal place of business at 8 Greenway Plaza, Suite 1000, Houston, Texas 77046 ("Borrower"), hereby unconditionally promises to pay to the order of MORGAN STANLEY MORTGAGE CAPITAL INC., a New York corporation, as payee, having an address at 1221 Avenue of the Americas, 27th Floor, New York, New York 10020 ("Lender"), or at such other place as the holder hereof may from time to time designate in writing, the principal sum of Thirty-Seven Million Nine Hundred Fifty Thousand and No/100 Dollars (\$37,950,000.00), in lawful money of the United States of America with interest thereon to be computed from the date of this Note at the Applicable Interest Rate (defined below) in accordance with the terms of this Note.

Article 1: Payment Terms

Borrower agrees to pay sums under this Note in installments as follows:

- (a) on the date hereof, a payment of interest only with respect to the period commencing on the date hereof and ending on, and including, the last day of the month in which this Note is executed;
- (b) beginning on the first day of February, 2007, and on the first day of each calendar month thereafter, up to and including the first day of December, 2013 (each, a "Payment Date"), payments of interest only based on the Applicable Interest Rate accrued on the outstanding principal balance; and
- (c) the balance of the principal sum and all interest thereon on the first day of January 1, 2014 (the "Maturity Date").

Article 2: Interest

The interest rate on this Note is five and forty-eight hundredths percent (5.48%) per annum (the "Applicable Interest Rate"). Interest on the principal sum of this Note shall be calculated by multiplying (a) the actual number of days elapsed in the period for which the calculation is being made by (b) a daily rate based on a three hundred sixty (360) day year (that is, the Applicable Interest Rate or the Default Rate, as then applicable, divided by 360) by (c) the outstanding principal balance.

Article 3: Default and Acceleration

Borrower covenants and agrees that if

- (a) any payment required hereunder (other than the payment due on the Maturity Date) is not paid prior to the fifth (5th) day after the same is due, or
- (b) the entire Debt (defined below) is not paid on or before the Maturity Date or
- (c) any other Event of Default (as defined in the Security Instrument (defined below)) shall continue to exist after giving effect to all applicable grace periods, then at the option of Lender (i) the whole of the principal sum of this Note, (ii) interest, default interest, late charges and other sums, as provided in this Note, the Security Instrument or the Other Security Documents (as defined in the Security Instrument), (iii) all other monies agreed or provided to be paid by Borrower in this Note, the Security Instrument or the Other Security Documents, (iv) all sums advanced pursuant to the Security Instrument to protect and preserve the Property (defined below) and the lien and the security interest created thereby, and (v) all sums advanced and costs and expenses incurred by Lender in connection with the Debt or any part thereof, any renewal, extension, or change of or substitution for the Debt or any part thereof, or the acquisition or perfection of the security therefor, whether made or incurred

at the request of Borrower or Lender (all the sums referred to in (i) through (v) above shall collectively be referred to as the "Debt") shall without notice become immediately due and payable. Whenever any payment to be made under this Note, the Security Instrument or under any Other Security Document shall be stated to be due on a day which is not a Business Day (hereinafter defined), the due date thereof shall be the Business Day immediately preceding such day. For purposes hereof, the term "Business Day" shall mean any day other than a Saturday, Sunday or any other day on which banks in New York, New York are not open for business.

Article 4: Default Interest

Borrower agrees that upon the occurrence of an Event of Default, Lender shall be entitled to receive and Borrower shall pay interest on the entire unpaid principal sum at a per annum rate equal to the lesser of (a) five percent (5%) plus the Applicable Interest Rate or (b) the maximum interest rate which Borrower may by law pay (the "Default Rate"). The Default Rate shall be computed from the occurrence of the default giving rise to such Event of Default (without regard to any notice or grace period) until the earlier of the date upon which the Event of Default is cured or the date upon which the Debt is paid in full. Interest calculated at the Default Rate shall be deemed part of the Debt, and shall be deemed secured by the Security Instrument. This clause, however, shall not be construed as an agreement or privilege to extend the date of the payment of the Debt, nor as a waiver of any other right or remedy accruing to Lender by reason of the occurrence of any Event of Default.

Article 5: Late Charge

If any monthly installment payable under this Note is not paid prior to the fifth (5th) day after the applicable Payment Date, Borrower shall pay to Lender upon demand an amount equal to the lesser of five percent (5%) of such unpaid sum or the maximum amount permitted by applicable law to defray the expenses incurred by Lender in handling and processing the delinquent payment and to compensate Lender for the loss of the use of the delinquent payment and the amount shall be secured by the Security Instrument and the Other Security Documents.

Article 6: Prepayment; Defeasance or Yield Maintenance

- (a) The principal balance of this Note may not be prepaid in whole or in part except as expressly permitted pursuant hereto.
- (b) Subject to compliance with and satisfaction of the terms and conditions of this Article 6 and provided that no Event of Default exists, Borrower may elect to obtain a release (the "Release") of the Property from the lien of the Security Instrument on any Payment Date after the Lockout Period Expiration Date (defined below) by delivering to Lender, as security for the payment of all interest due and to become due pursuant to this Note through the Maturity Date, plus the principal balance of this Note scheduled to be outstanding on the Maturity Date, Defeasance Collateral (defined below) sufficient to generate Scheduled Defeasance Payments (defined below) (the Release and the delivery of the Defeasance Collateral, a "Defeasance").
- (c) As a condition precedent to a Defeasance, and prior to any Release, Borrower shall have complied with all of the following:
- (i) Borrower shall provide not less than sixty (60) days prior written notice to Lender specifying a Payment Date upon which it intends to effect a Defeasance hereunder (the "Defeasance Date").
- (ii) All accrued and unpaid interest on the principal balance of this Note to and including the Defeasance Date, the scheduled amortization payment due on such Defeasance Date, and all other sums then due under this Note, the Security Instrument and the Other Security Documents, shall be paid in full on or prior to the Defeasance Date.
- (iii) Borrower shall execute and deliver to Lender any and all certificates, opinions, documents or instruments reasonably required by Lender in connection with the Defeasance and Release, including, without limitation, a pledge and security agreement reasonably satisfactory to Lender creating a first priority lien on the Defeasance Collateral (a "Defeasance Security Agreement"). This Note shall thereafter be secured by the Defeasance Collateral delivered in connection with the Defeasance. After Defeasance, this Note cannot be prepaid in whole or in part or be the subject of any further Defeasance.
- (iv) Borrower shall have delivered to Lender an opinion of Borrower's counsel that would be satisfactory to a prudent lender stating (A) that the Defeasance Collateral and the proceeds thereof have been duly and validly assigned and delivered to Lender and that Lender has a valid, perfected, first priority lien and security interest in the Defeasance Collateral delivered by Borrower and the proceeds thereof, (B) that if the holder of this Note shall at the time of the Release be a REMIC (defined below), (1) the Defeasance Collateral has been validly assigned to the REMIC Trust which holds this Note (the "REMIC Trust"), (2) the Defeasance has been effected in accordance with the requirements of Treasury Regulation 1.860(g)-2(a)(8) (as such regulation may be amended or substituted from time to time) and will not be treated as an exchange pursuant to Section 1001 of the IRS Code and (3) the tax qualification and status of the REMIC Trust as a REMIC will not be adversely affected or impaired as a result of the Defeasance and (C) that the delivery of the Defeasance Collateral and the grant of a security interest therein to Lender shall not constitute an avoidable preference under Section 547 of the U.S. Bankruptcy Code or applicable state law. The term "REMIC" shall mean a "real estate mortgage investment conduit" within the meaning of Section 860D of the IRS Code. The term "IRS Code" shall mean the United States Internal Revenue Code of 1986, as amended, and the related Treasury Department regulations, including temporary regulations.
- (v) Borrower shall have delivered to Lender written confirmation from the Rating Agencies (defined in the Security Instrument) that such Defeasance will not result in a withdrawal, downgrade or qualification of the then current ratings by the applicable Rating Agencies of the Securities or Participations (each as defined in the Security Instrument). If required by the Rating Agencies, Borrower's expense, also deliver or cause to be delivered a non-consolidation opinion with respect to the Defeasance Obligor (as defined below), if any, in form and substance that would be satisfactory to a prudent lender.
- (vi) Borrower shall have delivered to Lender a certificate given by Borrower's independent certified public accountant certifying that the Defeasance Collateral shall generate the Scheduled Defeasance Payments.
- (d) In connection with any Defeasance hereunder, Borrower shall transfer and assign all obligations, rights and duties under and to this Note and the Defeasance Security Agreement together with the pledged Defeasance Collateral to a newly -created successor entity, which entity shall be a single purpose, bankruptcy remote entity and which entity shall be designated or established by Lender, at Lender's option (the "Defeasance Obligor"). Lender shall also have the right to purchase on behalf of Borrower, or cause to be purchased on behalf of Borrower, the pledged Defeasance Collateral. Such rights to designate or establish the Defeasance Obligor as provided above or to purchase or cause the purchase on behalf of Borrower of the pledged Defeasance Collateral as provided above may be exercised by Lender in its sole discretion and shall be retained by Lender named herein (and any successor or assign of such Lender named herein under a specific assignment of such retained rights separate and apart from a transfer or securitization of the Loan in whole or in part. Such Defeasance Obligor shall assume the obligations under the Note and any Defeasance Security Agreement and shall be bound by and obligated under Sections 3.1, 7.2, 7.4(a), 11.2, 11.7 and 14.2 and Articles 13 and 15 of the Security Instrument; provided, however, that all references therein to "Property" or "Personal Property" shall be deemed to refer only to the Defeasance Collateral delivered to Lender, and Borrower shall be relieved of its obligations under such documents and, except with respect to any provisions therein which by

their terms expressly survive payment of the Debt in full, the Other Security Documents. Borrower shall pay \$1,000 to any such Defeasance Obligor as consideration for assuming such obligations.

- (e) The following terms shall have the meaning set forth below:
- (i) The term "Defeasance Collateral" as used herein shall mean direct, non-callable and non-redeemable obligations of the United States of America for the payment of which its full faith and credit is pledged, each of which shall be duly endorsed by the holder thereof as directed by Lender or accompanied by a written instrument of transfer in form and substance wholly satisfactory to Lender (including, without limitation, such instruments as may be required by the depository institution holding such securities or by the issuer thereof, as the case may be, to effectuate book-entry transfers and pledges through the book-entry facilities of such institution) in order to perfect upon the delivery of the Defeasance Collateral a first priority security interest therein in favor of the Lender in conformity with all applicable state and federal laws governing the granting of such security interests. Borrower shall authorize and direct that the payments received from such obligations shall be made directly to Lender or Lender's designee and applied to satisfy the obligations of Borrower or, if applicable, the Defeasance Obligor, under this Note.
- (ii) The term "Scheduled Defeasance Payments" as used herein shall mean the scheduled payments of interest and principal in accordance with the terms of the Defeasance Collateral (without consideration of any reinvestment of interest therefrom), providing for payments prior, but as close as possible, to all successive Payment Dates after the Defeasance Date through and including the Maturity Date, and in amounts equal to or greater than the scheduled payments of interest and principal due under this Note, including the principal balance of this Note scheduled to be outstanding on the Maturity Date.
- (iii) The term "Lockout Period Expiration Date" shall mean the date which is the earlier of (A) the second anniversary, of the date that is the "startup day," within the meaning of Section 860G(a) (9) of the IRS Code, of a REMIC that holds this Note or (B) the five (5) year anniversary of the first day of the first full calendar month following the date of this Note.
- (f) Upon Borrower's compliance with all of the conditions to Defeasance and a Release set forth in this Article 6, Lender shall release the Property from the lien of the Security Instrument and the Other Security Documents. All costs and expenses of Lender, third parties and the Rating Agencies incurred in connection with the Defeasance and Release, including, without limitation, the cost of establishing the Defeasance Obligor and Lender's counsel's fees and expenses, shall be paid by Borrower simultaneously with the delivery of the Release documentation. Any revenue, documentary stamp or intangible taxes or any other tax or charge due in connection with the Defeasance shall be paid by Borrower simultaneously with the occurrence of any Defeasance.
- (g) If a Default Prepayment (defined below) occurs, Borrower shall pay to Lender the entire Debt, including, without limitation, an amount (the "Default Consideration") equal to the greater of (i) the amount (if any) which when added to the then outstanding principal amount of this Note will be sufficient to purchase Defeasance Collateral providing the required Scheduled. Defeasance Payments assuming Defeasance would be permitted hereunder, or (ii) one percent (1%) of the Default Prepayment. For purposes of this Note, the term "Default Prepayment" shall mean a prepayment of the principal amount of this Note made after the occurrence of any Event of Default or an acceleration of the Maturity Date under any circumstances, including, without limitation, a prepayment occurring in connection with reinstatement of the Security Instrument provided by statute under foreclosure proceedings or exercise of a power of sale, any statutory right of redemption exercised by Borrower or any other party having a statutory right to redeem or prevent foreclosure, any sale in foreclosure or under exercise of a power of sale or otherwise.
- (h)
- (i) Notwithstanding anything to the contrary set forth in this Article 6, at any time after the Lockout Period Expiration Date, Borrower may elect, in lieu of a Defeasance, to prepay the principal balance of this Note in whole, but not in part, upon not less than sixty (60) days prior written notice to Lender specifying the date on which prepayment is to be made (the "Prepayment Date"), which date must be a Payment Date, and upon payment of:
- (A) all accrued interest to and including the Prepayment Date;
- (B) all other sums due under this Note, the Security Instrument and all Other Security Documents; and
- (C) the Prepayment Consideration (as defined below).
- (ii) The Prepayment Consideration shall equal an amount equal to the greater of (A) one percent (1%) of the principal balance of this Note being prepaid or (B) the product of (1) the ratio of the amount of the principal balance of this Note being prepaid over the outstanding principal balance of this Note on the Prepayment Date (after subtracting the scheduled principal payment on such Prepayment Date), multiplied by (2) the present value as of the Prepayment Date of the remaining scheduled payments of principal and interest from the Prepayment Date through the Maturity Date (including any balloon payment) determined by discounting such payments at the Discount Rate (as hereinafter defined) less the amount of the outstanding principal balance of this Note on the Prepayment Date. The "Discount Rate" is the rate which, when compounded monthly, is equivalent to the Treasury Rate (as hereinafter defined), when compounded semi-annually. The "Treasury Rate" is the yield calculated by the linear interpolation of the yields, as reported in Federal Reserve Statistical Release H.15-Selected Interest Rates

under the heading U.S. government securities/Treasury constant maturities for the week ending prior to the Prepayment Date, of U.S. Treasury constant maturities with maturity dates (one longer and one shorter) most nearly approximating the Maturity Date. (In the event Release H.15 is no longer published, Lender shall select a comparable publication to determine the Treasury Rate.) Lender shall notify Borrower of the amount and the basis of determination of the required prepayment consideration.

- (iii) If any notice of prepayment is given under this subparagraph (h), the principal balance of this Note and the other sums required under this prepayment section shall be due and payable on the Prepayment Date. Lender shall not be obligated to accept any prepayment of the principal balance of this Note unless it is accompanied by all sums due in connection therewith.
- (i) Notwithstanding anything to the contrary herein, Borrower may prepay the principal balance of this Note in whole, without premium or penalty, on any Payment Date during the three (3) months prior to the Maturity Date. In addition, Borrower shall prepay without premium or penalty the principal balance of this Note in an amount equal to any insurance proceeds or condemnation awards which Lender elects to have applied to the Debt pursuant to Sections 3.6 and 3.7 of the Security Instrument or the amount required by Lender due to changes in tax and debt credit pursuant to Section 7.3 (a) or (b) of the Security Instrument (provided, however, that in the event any such prepayment pursuant to this sentence shall be made on a date other than a Payment Date, the amount so prepaid shall include all interest which would have accrued on such amount through the next Payment Date). In each instance of prepayment permitted under this subparagraph (h), Borrower shall be required to pay all other sums due hereunder, and no principal amount repaid may be reborrowed.

Article 7: Security

This Note is secured by that certain Deed of Trust and Security Agreement dated the date hereof in the principal sum of \$37,950,000.00 given by Borrower to (or for the benefit of) Lender covering the fee simple estate of Borrower in certain premises located in Dallas County, State of Texas, and other property, as more particularly described therein (collectively, the "Property") and intended to be duly recorded in said Comity (the "Security Instrument"), and by the Other Security Documents.

Article 8: Loan Charges

- (a) This Note, the Security Instrument and the Other Security Documents are subject to the express condition that at no time shall Borrower be obligated or required to pay interest on the principal balance due hereunder at a rate which could subject Lender to either civil or criminal liability as a result of being in excess of the maximum interest rate which Borrower is permitted by applicable law to contract or agree to pay. If by the terms of this Note, the Security Instrument and the Other Security Documents, Borrower is at any time required or obligated to pay interest on the principal balance due hereunder at a rate in excess of such maximum rate, the Applicable Interest Rate, the Default Rate or any other consideration that constitutes interest under applicable law, as the case may be, shall be deemed to be immediately reduced to such maximum rate and all previous payments in excess of the maximum rate shall be deemed to have been payments in reduction of principal and not on account of the interest due hereunder. All sums paid or agreed to be paid to Lender for the use, forbearance, or detention of the Debt, shall, to the extent permitted by applicable law, be amortized, prorated, allocated, and spread throughout the full stated term of the Note until payment in full so that the rate or amount of interest on account of the Debt does not exceed the Maximum Lawful Rate (as defined below) of interest from time to time in effect and applicable to the Debt for so long as the Debt is outstanding.
- (b) It is expressly stipulated and agreed to be the intent of Borrower and Lender at all times to comply strictly with the applicable Texas law governing the maximum rate or amount of interest payable on this Note (or applicable United Stated federal law to the extent that it permits Lender to contract for, charge, take, reserve or receive a greater amount of interest than under Texas law). If the applicable law is ever judicially interpreted so as to render usurious any amount (i) contracted for, charged, taken, reserved or received pursuant to this Note, any of the other Loan Documents or any other communications or writing by or between Borrower and Lender related to the transaction or transactions that are the subject matter of the Loan Documents, (ii) contracted for, charged or received by reason of Lender's exercise of the option to accelerate the maturity of this Note, or (iii) Borrower will have paid or Lender will have received by reason of any voluntary prepayment by Borrower of this Note, then it is Borrower's and Lender's express intent that all amounts charged in excess of the Maximum Lawful Rate shall be automatically cancelled, ab initio, and all amounts in excess of the Maximum Lawful Rate theretofore collected by Lender shall be credited on the principal balance of this Note (or, if this Note has been or would thereby be paid in full, refunded to Borrower), and the provisions of this Note and the other Loan Documents immediately be deemed reformed and the amounts thereafter collectible hereunder and thereunder reduced, without the necessity of the execution of any new document, so as to comply with the applicable law, but so as to permit the recovery of the fullest amount otherwise called for hereunder and thereunder; provided, however, if this Note has been paid in full before the end of the stated term of this Note, then Borrower and Lender agree that Lender shall, with reasonable promptness after Lender discovers or is advised by Borrower that interest was received in an amount in excess of the Maximum Lawful Rate, either refund such excess interest to Borrower and/or credit such excess interest against this Note then owing by borrower to Lender. In no event shall the provisions of Chapter 346 of the Texas Finance Code (which regulates certain revolving credit loan accounts and revolving triparty accounts) apply to this Note, Notwithstanding anything to the contrary contained herein or in any of the other Loan Documents, it is not the intention of Lender to accelerate the maturity of any interest that has not accrued at the time of such acceleration or to collect unearned interest at the time of such acceleration.
- (c) As used herein, the term "Maximum Lawful Rate" shall mean the maximum lawful rate of interest which may be contracted for, charged, taken, received or reserved by Lender in accordance with the applicable laws of the State of Texas (or applicable United States federal law to the extent that it permits Lender to contract for, charge, take, receive or reserve a greater amount of interest than under Texas law), taking into account all Charges (as herein defined) made in connection with the transaction evidenced by this Note and the other Loan Documents. As used herein, the term "Charges" shall mean all fees, charges and/or any other things of value, if any, contracted for, charged, received, taken or reserved by Lender in connection with the transactions relating to this Note and the other Loan Documents, which are treated as interest under applicable law.
- (d) To the extent that Lender is relying on Chapter 303 of the Texas Finance Code to determine the Maximum Lawful Rate payable on this Note, Lender will utilize the weekly ceiling from time to time in effect as provided in such Chapter 303, as amended. To the extent United States federal law permits Lender to contract for, charge, take, receive or reserve a greater amount of interest than under Texas law, Lender will rely on United States federal law instead of such Chapter 303 for the purpose of determining the Maximum Lawful Rate. Additionally, to the extent permitted by applicable law now or hereafter in effect, Lender may, at its option and from time to time, utilize any other method of establishing the Maximum Lawful Rate under such Chapter 303 or under other applicable law by giving notice, if required, to Borrower as provided by applicable law now or hereafter in effect.

Article 9: Waivers

Borrower and all others who may become liable for the payment of all or any part of the Debt do hereby severally waive presentment and demand for payment, notice of dishonor, protest and notice of protest and non-payment and all other notices of any kind, except for notices expressly provided for in this Note, the Security Instrument or the Other Security Documents. No release of any security for the Debt or extension of time for payment of this Note or any installment hereof, and no alteration, amendment or waiver of any provision of this Note, the Security Instrument or the Other Security Documents made by agreement between Lender or any other person or party shall release, modify, amend, waive, extend, change, discharge, terminate or affect the liability of Borrower, and any other person or entity who may become liable for the payment of all or any part of the Debt, under this Note, the Security Instrument or the Other Security Documents. No notice to or demand on Borrower shall be deemed to be a waiver of the obligation of Borrower or of the right of Lender to take further action without further notice or demand as provided for in this Note, the Security Instrument or the Other Security Documents. If Borrower is a partnership, corporation or limited liability company, the agreements contained herein shall remain in full force and effect, notwithstanding any changes in the individuals or entities comprising Borrower, and the term "Borrower," as used herein, shall, to the extent permitted by applicable law, include any alternate or successor entity, but any predecessor entity, and its partners or members, as the case may be, shall not thereby be released from any liability incurred prior to the date of such change. (Nothing in the foregoing sentence shall be construed as a consent to, or a waiver of, any prohibition or restriction on transfers of interests in Borrower which may be set forth in the Security Instrument or any Other Security Document.)

Article 10: Waiver of Trial by Jury

BORROWER HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, THE RIGHT TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM, WHETHER IN CONTRACT, TORT OR OTHERWISE, RELATING DIRECTLY OR INDIRECTLY TO THE LOAN EVIDENCED BY THIS NOTE, THE APPLICATION FOR THE LOAN EVIDENCED BY THIS NOTE, THIS NOTE, THE SECURITY INSTRUMENT OR THE OTHER SECURITY DOCUMENTS OR ANY ACTS OR OMISSIONS OF LENDER, ITS OFFICERS, EMPLOYEES, DIRECTORS OR AGENTS IN CONNECTION THEREWITH.

Article 11: Exculpation

(a) Notwithstanding anything to the contrary contained in this Note, the Security Instrument or any Other Security Document (but subject to the provisions of subsections (b), (c) and (d) of this Article 11), Lender shall not enforce the liability and obligation of Borrower to perform and observe the obligations contained in this Note or the Security Instrument by any action or proceeding wherein a money judgment or any deficiency judgment or other judgment establishing any personal liability shall be sought against Borrower or any principal, director, officer, employee, beneficiary, shareholder, partner, member, trustee, agent or affiliate of Borrower or any person owning, directly or indirectly, any legal or beneficial interest in Borrower, or any successors or assigns of any of the foregoing (collectively, the "Exculpated Parties"), except that Lender may bring a foreclosure action, action for specific performance or other appropriate action or proceeding to enable Lender to enforce and realize upon this Note, the Security Instrument, the Other Security Documents, and the interest in the Property, the Rents (as defined in the Security Instrument) and any other collateral given to Lender to secure this Note; provided, however, subject to the provisions of subsections (b), (c) and (d) of this Article 11, that any judgment in any such action or proceeding shall be enforceable against Borrower only to the extent of Borrower's interest in the Property, in the Rents and in any other collateral given to Lender to secure this Note. Lender, by accepting this Note and the Security Instrument, agrees that it shall not, except as otherwise provided in this Article 11, sue for, seek or demand any deficiency judgment against Borrower or any of the Exculpated Parties, in any such action or proceeding, under or by reason of or under or in connection with this Note, the Security Instrument or the Other Security Documents. The provisions of this Article 11 shall not, however, (i) constitute a waiver, release or impairment of any obligation evidenced or secured by this Note, the Security Instrument or the Other Security Documents delivered to Lender; (ii) impair the right of Lender to name Borrower as a party defendant in any action or suit for judicial foreclosure and sale under the Security Instrument; (iii) affect the validity or enforceability of any indemnity, guaranty, master lease or similar instrument made in connection with this Note, the Security Instrument, or the Other Security Documents; (iv) impair the right of Lender to obtain the appointment of a receiver; (v) impair the enforcement of the Assignment of Leases and Rents executed in connection herewith; (vi) impair the right of Lender to enforce the provisions of Section 12.2 of the Security Instrument or of Section 3.12(e) of the Security Instrument; or (vii) impair the right of Lender to obtain a deficiency judgment or other judgment on the Note against Borrower if necessary to fully realize the security granted by the Security Instrument or to commence any other appropriate action or proceeding in order for Lender to exercise its remedies against the Property.

(b) Notwithstanding the provisions of this Article 11 to the contrary, Borrower shall be personally liable to Lender for the Losses (as defined in the Security Instrument) Lender incurs due to: (i) fraud or intentional misrepresentation by Borrower or any of the Exculpated Parties in connection with the Loan; (ii) the gross negligence or willful misconduct of Borrower; (iii) the removal or disposal of any portion of the Property after an Event of Default; (iv) Borrower's misapplication, misappropriation or conversion of Rents received by Borrower after the occurrence of an Event of Default; (v) Borrower's misapplication, misappropriation or conversion of tenant security deposits or Rents collected more than one (1) month in advance which are not delivered to Lender for application to the Loan; (vi) the misapplication, misappropriation or conversion of insurance proceeds or condemnation awards; (vii) Personal Property of Borrower (as defined in the Security Instrument) taken from the Property by or on behalf of Borrower or any of the Exculpated Parties and not replaced with Personal Property of the same utility and of the same or greater value; (viii) any act of arson by Borrower or any of the Exculpated Parties; (ix) any fees or commissions paid by Borrower after the occurrence of an Event of Default to any Exculpated Party in violation of the terms of this Note, the Security Instrument or the Other Security Documents; (x) failure to pay charges for labor or materials or other charges that can create liens on any portion of

the Property; (xi) any security deposits, advance deposits or any other deposits collected under leases with respect to the Property not being delivered to Lender upon a foreclosure of the Property or action in lieu thereof, except to the extent any such security deposits were applied in accordance with the terms and conditions of the applicable Leases (as defined in the Security Instrument) prior to the occurrence of the Event of Default that gave rise to such foreclosure or action in lieu thereof; (xii) any failure by Borrower to permit on-site inspections of the Property as required by the Security Instrument and/or the Other Security Documents; (xiii) any failure of Borrower to appoint a new property manager upon the request of Lender as required by the terms of the Security Instrument and/or the Other Security Documents; (xiv) Borrower's material breach of, or failure to comply with, the representations, warranties and covenants contained in Articles 5.8(b), 5.19 and/or 12 of the Security Instrument; (xv) Borrower's failure to provide financial information to Lender as required by Section 3.12 of the Security Instrument; (xvi) any (z), (aa), (bb), (cc), (dd), (ee), (ff), (gg), (hh), (ii), (jj), (kk), (ll) or (mm) of the Security Instrument. Borrower's failure to deliver to Lender any Letter of Credit (as defined in that certain Springing Letter of Credit and Security Agreement, of even date herewith, given by Borrower in favor of Lender (the "Springing Letter of Credit Agreement")) required to be delivered by Borrower to Lender pursuant to the Springing Letter of Credit and Security Agreement; and/or (xvi) the failure of any tenant at the Property to obtain a permanent certificate of occupancy with respect to the space occupied by each such tenant at the Property; provided, however, that this subsection (xvi) shall be of no further force and effect upon delivery by Borrower to Lender of evidence satisfactory to Lender in its reasonable discretion that each tenant at the Property has obtained a permanent certificate of occupancy with respect to the space occupied by each such tenant at the Property.

(c) Notwithstanding the foregoing, the agreement of Lender not to pursue recourse liability against Borrower as set forth in subsection (a) above, as to Borrower, SHALL BECOME NULL AND VOID and shall be of no further force and effect and the Debt shall be fully recourse to Borrower in the event that; (i) the first full monthly payment of interest under Section I(b) of this Note is not paid when due; (ii) Borrower fails to comply with any provision of Section 4.2(b), (c), (d), (e), (g) or (m) of the Security Instrument; (iii) Borrower defaults under Article 8 of the Security Instrument; (iv) Borrower files a voluntary petition under the U.S. Bankruptcy Code or any other Federal or state bankruptcy or insolvency law; (v) an affiliate, officer, director or representative which controls Borrower, directly or indirectly, files, or joins in the filing of, an involuntary petition against Borrower under the U.S. Bankruptcy Code or any other Federal or state bankruptcy or insolvency law, or solicits or causes to be solicited petitioning creditors for any involuntary petition filed against it, by any other person or entity under the U.S. Bankruptcy Code or any other Federal or state bankruptcy or insolvency law, or solicits or causes to be solicited petitioning creditors for any involuntary petition from any person or entity; (vii) any affiliate, officer, director or representative which controls Borrower consents to or acquiesces in or joins in an application for the appointment of a custodian, receiver, trustee, or examiner for Borrower or any portion of the Property; or (viii) Borrower makes an assignment for the benefit of creditors, or admits, in writing or in any legal proceeding, its insolvency or inability to pay its debts as they become due.

(d) Nothing herein shall be deemed to be a waiver of any right which Lender may have under Section 506(a), 506(b), 11 11(b) or any other provision of the U.S. Bankruptcy Code to file a claim for the full amount of the indebtedness secured by the Security Instrument or to require that all collateral shall continue to secure all of the indebtedness owing to Lender in accordance with this Note, the Security Instrument and the Other Security Documents.

Article 12: Authority

Borrower (and the undersigned representative of Borrower, if any) represents that Borrower has full power, authority and legal right to execute and deliver this Note, the Security Instrument and the Other Security Documents and that this Note, the Security Instrument and the Other Security Documents constitute valid and binding obligations of Borrower.

Article 13: Governing Law

This Note shall be governed, construed, applied and enforced in accordance with the laws of the state in which the Property is located without reference or giving effect to any choice of law doctrine.

Article 14: Notices

All notices required or permitted hereunder shall be given as provided in the Security Instrument.

Article 15: Incorporation by Reference

All of the terms, covenants and conditions contained in the Security Instrument and the Other Security Documents are hereby made part of this Note to the same extent and with the same force as if they were fully set forth herein.

Article 16: Miscellaneous

- (a) Wherever pursuant to this Note it is provided that Borrower pay any costs and expenses, such costs and expenses shall include, but not be limited to, reasonable legal fees and disbursements of Lender, whether with respect to retained firms, the reimbursement for the expenses of in-house staff, or otherwise. Borrower shall pay to Lender on demand any and all expenses, including legal expenses and reasonable attorneys' fees, incurred or paid by Lender in enforcing this Note, whether or not any legal proceeding is commenced hereunder, together with interest thereon at the Default Rate from the date paid or incurred by Lender until such expenses are paid by Borrower.
- (b) This Note may not be modified, amended, waived, extended, changed, discharged or terminated orally or by any act or failure to act on the part of Borrower or Lender, but only by an agreement in writing signed by the party against whom enforcement of any modification, amendment, waiver, extension, change, discharge or termination is sought.
- (c) If Borrower consists of more than one person or party, the obligations and liabilities of each person or party shall be joint and several.
- (d) Whenever used, the singular number shall include the plural, the plural number shall include the singular, and the words "Lender" and "Borrower" shall include their respective successors, assigns, heirs, executors and administrators.

IN WITNESS WHEREOF, Borrower has duly executed this Note as of the day and year first above written.

BORROWER:

AMREIT CASA LINDA, LP, a Texas limited partnership

By: AmREIT Casa Linda GP, Inc.,

a Texas corporation, Its General Partner

By: /s/ Chad C. Braun
Name: Chad C. Braun
Title: Vice President